UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 OR 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): March 20, 2023 (March 14, 2023)

INPIXON

(Exact name of registrant as specified in its charter)

Nevada	001-36404	88-0434915		
(State or other jurisdiction of incorporation)	(Commission File Number)	(I.R.S. Employer Identification No.)		
2479 E. Bayshore Road, Suite 1 Palo Alto, CA	95	94303		
(Address of principal executive off	ices)	(Zip Code)		
Reg	istrant's telephone number, including area code: (408)	702-2167		
	N/A			
	Former name or former address, if changed since last r	eport)		
Check the appropriate box below if the Form 8-K is intend	led to simultaneously satisfy the filing obligation of the	Registrant under any of the following provisions:		
□ Written communications pursuant to Rule 425 under t	he Securities Act (17 CFR 230.425)			
□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)				
□ Pre-commencement communications pursuant to Rule	e 14d-2(b) under the Exchange Act (17 CFR 240.14d-2	(b))		
□ Pre-commencement communications pursuant to Rule	e 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))		
Securities registered pursuant to Section 12(b) of the Act:				
Title of Each Class	Trading Symbol(s)	Name of Each Exchange on Which Registered		
Common Stock	INPX	The Nasdaq Capital Market		
Indicate by check mark whether the registrant is an emerg the Securities Exchange Act of 1934 (§240.12b-2 of this cl		urities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of		
Emerging growth company \square				
If an emerging growth company, indicate by check mark accounting standards provided pursuant to Section 13(a) o		nsition period for complying with any new or revised financial		
-				

Item 1.01 Entry into a Material Definitive Agreement.

As further described in Item 2.01 of this Report, on March 14, 2023, Inpixon ("Inpixon," the "Company," "we," "us," or "our"), announced that Inpixon, its former wholly owned subsidiary, CXApp Holding Corp. ("Legacy CXApp"), and CXApp Inc. (formerly known as KINS Technology Group Inc., "KINS" or "New CXApp") had consummated the previously announced spin-off of Legacy CXApp and combination of Legacy CXApp and New CXApp.

Employee Matters Agreement

On March 14, 2023, in connection with the consummation of the Business Combination and as contemplated by the Separation Agreement, CXApp, Legacy CXApp, Inpixon and Merger Sub entered into the Employee Matters Agreement (the "Employee Matters Agreement"). The material terms of the Employee Matters Agreement are described in the section of the proxy statement/prospectus (the "Proxy Statement/Prospectus"), filed with the Securities and Exchange Commission by KINS on February 13, 2023, beginning on page 139 titled "Proposal No. 1 - The Business Combination Proposal - Related Agreements - Summary of the Ancillary Agreements - Form of Employee Matters Agreement." That description, which is incorporated herein by reference, is qualified in its entirety by the text of the Employee Matters Agreement, which is included as Exhibit 10.1 to this Report and also is incorporated herein by reference.

Tax Matters Agreement

On March 14, 2023, in connection with the consummation of the Business Combination and as contemplated by the Separation Agreement, CXApp, Legacy CXApp and Inpixon entered into the Tax Matters Agreement (the "Tax Matters Agreement"). The material terms of the Tax Matters Agreement are described in the section of the Proxy Statement/Prospectus beginning on page 140 titled "Proposal No. 1 - The Business Combination Proposal - Related Agreements - Summary of the Ancillary Agreements - Form of Tax Matters Agreement." That description, which is incorporated herein by reference, is qualified in its entirety by the text of the Tax Matters Agreement, which is included as Exhibit 10.2 to this Report and also is incorporated herein by reference.

Transition Services Agreement

On March 14, 2023, in connection with the consummation of the Business Combination and as contemplated by the Separation Agreement, Legacy CXApp and Inpixon entered into a Transition Services Agreement (the "Transition Services Agreement"). The material terms of the Transition Services Agreement are described in the section of the Proxy Statement/Prospectus beginning on page 141 titled "Proposal No. 1 - The Business Combination Proposal - Related Agreements - Summary of the Ancillary Agreements - Form of Transition Services Agreement." That description, which is incorporated herein by reference, is qualified in its entirety by the text of the Transition Services Agreement, which is included as Exhibit 10.3 to this Report and also is incorporated herein by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets.

On March 14, 2023 (the "Distribution Date"), Inpixon completed the separation (the "Separation") of its enterprise apps business (including its workplace experience technologies, indoor mapping, events platform, augmented reality and related business solutions) and certain related assets and liabilities (the "Enterprise Apps Business") through a spin-off of Legacy CXApp to Inpixon's shareholders of record as of March 6, 2023 (the "Record Date") on a pro rata basis (the "Distribution") and merger (the "Merger") of Legacy CXApp with a wholly owned subsidiary of New CXApp in a Reverse Morris Trust transaction (collectively, the "Transactions") pursuant to the Agreement and Plan of Merger, dated as of September 25, 2022, by and among Inpixon, Legacy CXApp, New CXApp and its then wholly owned subsidiary, KINS Merger Sub Inc. ("Merger Sub"), and the Separation and Distribution Agreement, dated as of September 25, 2022 (the "Separation Agreement"), by and among Inpixon, Design Reactor Inc., Legacy CXApp and New CXApp (collectively with the other related transaction documents, the "Transaction Agreements"). Pursuant to the Transaction Agreements, Inpixon contributed (the "Contribution") to Legacy CXApp cash and certain assets and liabilities constituting the Enterprise Apps Business, including certain related subsidiaries of Inpixon, to Legacy CXApp. In consideration for the Contribution, Legacy CXApp issued to Inpixon additional shares of Legacy CXApp common stock such that the number of shares of Legacy CXApp common stock then outstanding equaled the number of shares of Legacy CXApp common stock necessary to effect the Distribution. Pursuant to the Distribution, Inpixon shareholders as of the Record Date received one share of Legacy CXApp common stock for each share of Inpixon common stock held as of such date. Pursuant to the Merger Agreement, each share of Legacy CXApp common stock was thereafter exchanged for the right to receive 0.09752221612415190 of a share of New CXApp Class A common stock and 0.3457605844401750 of a share of New CXApp Class C common stock. New CXApp Class A common stock and New CXApp Class C common stock are identical in all respects, except that New CXApp Class C common stock is not listed and will automatically convert into New CXApp Class A common stock on the earlier to occur of (i) the 180th day following the closing of the Merger and (ii) the day that the last reported sale price of New CXApp Class A common stock equals or exceeds \$12.00 per share for any 20 trading days within any 30-trading day period following the closing of the Merger.

Upon the closing of the Transactions, Inpixon's existing securityholders held approximately 50.0% of the shares of New CXApp common stock outstanding.

After the Distribution Date, we do not beneficially own any shares of Legacy CXApp common stock and will no longer consolidate Legacy CXApp into our financial results. Beginning with the quarter ended March 31, 2023, Legacy CXApp's historical financial results for periods prior to the Distribution Date will be reflected in our consolidated financial statements as discontinued operations. The unaudited pro forma consolidated financial statements of the Company giving effect to the Transactions, and the related notes thereto, are attached hereto as Exhibit 99.1.

Item 2.02 Results of Operations and Financial Condition.

To the extent required by Item 2.02 of Form 8-K, the disclosure set forth in Item 8.01 below of this Current Report on Form 8-K is incorporated by reference in this Item 2.02.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description
10.1+	Employee Matters Agreement, dated March 14, 2023, by and among KINS, KINS Merger Sub Inc., Inpixon, and Legacy CXApp.
10.2	Tax Matters Agreement, dated March 14, 2023, by and among KINS, Inpixon, and Legacy CXApp.
10.3+	Transition Services Agreement, dated March 14, 2023, by and between Inpixon and Legacy CXApp.
99.1	Unaudited Pro Forma Condensed Consolidated Financial Statements.
104	Cover Page Interactive Data File (embedded within the Inline XBRL document).

⁺ The annexes, schedules, and certain exhibits to this Exhibit have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Registrant hereby agrees to furnish supplementally a copy of any omitted annex, schedule or exhibit to the SEC upon request.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

INPIXON

Date: March 20, 2023 By: /s/ Nadir Ali

Name: Nadir Ali
Title: Chief Executive Officer

EMPLOYEE MATTERS AGREEMENT

by and among

INPIXON,

CXAPP HOLDING CORP.,

KINS TECHNOLOGY GROUP INC.

and

KINS MERGER SUB INC.

Dated as of March 14, 2023

TABLE OF CONTENTS

ARTICLE 1	I DEFINIT	IONS AND INTERPRETATION	1
	1.1	General	1
	1.2	References; Interpretation	5
ARTICLE		AL PRINCIPLES	5
	2.1	Nature of Liabilities	5
	2.2	Transfers of Employees and Independent Contractors Generally	5
	2.3	Assumption and Retention of Liabilities Generally	7
	2.4	Treatment of Compensation and Benefit Plans; Terms of Employment	9
	2.5	Participation in Company Benefit Plans	9
	2.6	Service Recognition	9
	2.7	Assignment of Restrictive Covenants.	10
	2.8	WARN	10
	2.10	No Termination; No Change in Control	10
ADTICLE	ш серта	IN BENEFIT PLAN PROVISIONS	11
AKTICLE	3.1	Health and Welfare Benefit Plans	11
	3.1	Disability	12
	3.2	401(k) Plans	12
	3.4	Chargeback of Certain Costs	13
	3.4	Chargeback of Certain Costs	13
ARTICLE	IV FOUIT	Y INCENTIVE AWARDS	13
MATTELL	4.1	Company Equity Plans; Company Equity Awards	13
	4.2	Parent Equity Plan	13
	7.2	- man equity - mi	15
ARTICLE	V ADDITI	ONAL MATTERS	13
HATTELL	5.1	Cash Incentive Programs	13
	5.2	Severance	14
	5.3	Time-Off Benefits	14
	5.4	Workers' Compensation Liabilities	14
	5.5	COBRA	14
	5.6	Code Section 409A	14
	5.7	Payroll Taxes and Reporting	15
	5.8	Regulatory Filings	15
	5.9	Certain Requirements	15
ARTICLE	VI OBLIG	ATIONS OF PARENT AND MERGER SUB	15
	6.1	Obligations of Parent	15
ARTICLE '	VII GENE	RAL AND ADMINISTRATIVE	15
	7.1	Employer Rights	15
	7.2	Effect on Employment	15
	7.3	Consent of Third Parties	16
	7.4	Access to Employees	16
	7.5	Beneficiary Designation/Release of Information/Right to Reimbursement	16
	7.6	No Third Party Beneficiaries	16
	7.7	Employee Benefits Administration	16
	7.8	Audit Rights With Respect to Information Provided	17
	7.9	Cooperation	17
ARTICLE '		ELLANEOUS	17
	8.1	Entire Agreement	17
	8.2	Counterparts	17
	8.3	Survival of Agreements	17
	8.4	Notices	18
	8.5	Consents	19
	8.6	Assignment	19
	8.7	Successors and Assigns	19
	8.8	Termination and Amendment	19
	8.9	Subsidiaries	19
	8.10	Title and Headings	19
	8.10 8.11	Governing Law	20
	8.10 8.11 8.12	Governing Law WAIVER OF JURY TRIAL	20 20
	8.10 8.11 8.12 8.13	Governing Law WAIVER OF JURY TRIAL Severability	20 20 21
	8.10 8.11 8.12 8.13 8.14	Governing Law WAIVER OF JURY TRIAL Severability Interpretation	20 20 21 21
	8.10 8.11 8.12 8.13 8.14 8.15	Governing Law WAIVER OF JURY TRIAL Severability Interpretation No Duplication; No Double Recovery	20 20 21 21 21
	8.10 8.11 8.12 8.13 8.14 8.15 8.16	Governing Law WAIVER OF JURY TRIAL Severability Interpretation No Duplication; No Double Recovery No Waiver	20 20 21 21 21 21 21
	8.10 8.11 8.12 8.13 8.14 8.15	Governing Law WAIVER OF JURY TRIAL Severability Interpretation No Duplication; No Double Recovery	20 20 21 21 21

EMPLOYEE MATTERS AGREEMENT

This EMPLOYEE MATTERS AGREEMENT (this "Agreement"), dated as of March 14, 2023, is entered into by and among Inpixon, a Nevada corporation (the "Company"), CXApp Holding Corp., a Delaware corporation and a wholly owned subsidiary of the Company ("SpinCo"), KINS Technology Group Inc., a Delaware corporation ("Parent"), and KINS Merger Sub Inc., a Delaware corporation and wholly owned Subsidiary of Parent ("Merger Sub"). "Party" or "Parties" means the Company, SpinCo, Parent or Merger Sub, individually or collectively, as the case may be. Capitalized terms used in this Agreement, but not otherwise defined in this Agreement, shall have the meaning set forth in the Separation Agreement or the Merger Agreement.

WITNESSETH:

WHEREAS, the Company, acting through its direct and indirect Subsidiaries, currently conducts the Inpixon Retained Business and the Enterprise Apps Business;

WHEREAS, the Board of Directors of the Company (the "Company Board") has determined that it is appropriate, desirable and in the best interests of the Company and its stockholders to separate the Enterprise Apps Business from the Inpixon Retained Business, in the manner contemplated by the Separation and Distribution Agreement by and between the Company and SpinCo, Design Reactor, Inc., a California corporation and a wholly owned subsidiary of the Company ("Design Reactor"), and Parent, dated as of September 25, 2022 (the "Separation Agreement") and the Ancillary Agreements;

WHEREAS, following the Separation and pursuant to the Merger Agreement, Merger Sub shall merge with and into SpinCo and SpinCo will be the surviving corporation and a wholly owned Subsidiary of Parent; and

WHEREAS, in connection with the transactions contemplated by the Separation Agreement and the Merger Agreement, the Parties have agreed to enter into this Agreement for the purpose of allocating assets, Liabilities and responsibilities with respect to certain employee matters and employee compensation and benefit plans and programs among them and to address certain other employment-related matters;

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements, provisions and covenants contained in this Agreement, the Parties hereby agree as follows:

ARTICLE I DEFINITIONS AND INTERPRETATION

- 1.1 General. As used in this Agreement, the following terms shall have the following meanings:
- (a) "401(k) Plan Transition Date" shall mean (i) December 31 of the calendar year in which the Distribution Time occurs, or (ii) such earlier date as mutually agreed by the Parties.
 - (b) "Agreement" shall have the meaning set forth in the Preamble.

- (c) "Auditing Party" shall have the meaning set forth in Section 7.8(a).
- (d) "Benefit Plan" shall mean an "employee benefit plan" (within the meaning of Section 3(3) of ERISA but regardless of whether such plan is subject to ERISA) and each compensation plan, program, agreement or arrangement, including each pension, retirement, profit sharing, 401(k), severance, health and welfare, disability, deferred compensation, employment, termination, change-in-control, retention, fringe benefit, stock purchase, cash bonus or equity-based incentive or other benefit plan, program, agreement, policy or other arrangement, in each case, that is or was maintained for the benefit of current and/or former directors, officers, consultants or employees.
- (e) "Census" shall mean a list of all Business Employees (as defined in the Merger Agreement) (including any Business Employee who is on a leave of absence of any nature, paid or unpaid, authorized or unauthorized) and all Business Independent Contractors (as defined in the Merger Agreement) as of the date hereof, and sets forth for each such individual the following: (i) name or employee identification number; (ii) title or position (including whether full-time or part-time); (iii) place of work (city, state and country); (iv) hire or retention date; (v) current annual or hourly base compensation rate or contract fee and bonus opportunity; (vi) exempt or nonexempt status, or status as an independent contractor or consultant, as applicable; (vii) union representation (if any); (viii) with respect to Business Employees, active or leave status (and if on leave, type of leave and expected return date); (ix) employing or engaging entity; (x) visa or work authorization (if any); and (xi) with respect to Business Independent Contractors, whether engaged through a third-party entity or staffing agency, and the name of such entity or staffing agency, which shall be attached hereto as **Schedule A**, and as may be updated in accordance with Section 2.2(a).
- (f) "Code" means the Internal Revenue Code of 1986, as amended, or any successor federal income tax law. Reference to a specific Code provision also includes any proposed, temporary or final regulation in force under that provision.
 - (g) "Company" shall have the meaning set forth in the Preamble.
 - (h) "Company 401(k) Plan" shall mean the Company's Section 401(k) Savings/Retirement Plan.
- (i) "Company Benefit Plan" shall mean any Benefit Plan sponsored, maintained or contributed to (or required to be contributed to) by any member of the Company Group that (i) is or has been maintained, sponsored, contributed to or entered into by any member of the Company Group for the benefit of any SpinCo Employee or SpinCo Independent Contractor or for which any member of the SpinCo Group could have any Liability and (ii) that is not a SpinCo Benefit Plan.
 - (j) "Company Board" shall have the meaning set forth in the Recitals.
 - (k) "Company Employee" shall mean each employee of the Company or any of its Subsidiaries or Affiliates who does not qualify as a Business Employee.

- (I) "Company Equity Plans" shall mean the Company's 2011 Employee Stock Incentive Plan, and the Company's 2018 Employee Stock Incentive Plan, as amended from time to time.
- (m) "Company Group" shall mean (i) the Company, the Company Retained Business and each Person that is a direct or indirect Subsidiary of the Company as of immediately following the Distribution Time and (ii) each Business Entity that becomes a Subsidiary of the Company after the Distribution Time.
- (n) "Company Independent Contractor" shall mean each individual who is engaged as an independent contractor or consultant by the Company or any of its Subsidiaries or Affiliates who does not qualify as a Business Independent Contractor.
- (o) "Company Individual Agreement" shall mean each Benefit Plan sponsored, maintained entered into or contributed to by the Company under which no more than one service provider is eligible to receive compensation and/or benefits.
 - (p) "Company Option" shall mean an option to purchase shares of Company Common Stock granted pursuant to the Company Equity Plans.
 - (q) "Company Service Provider" shall mean a Company Employee, a Company Independent Contractor or a member of the Company Board.
 - (r) "Design Reactor" shall have the meaning set forth in the Recitals.
 - (s) "Distribution Time" shall mean the effective time of the Distribution pursuant to the Separation Agreement.
 - (t) "Effective Time" shall mean the "Effective Time" as defined in the Merger Agreement.
 - (u) "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
- (v) "Former Company Service Provider" means (i) any individual (other than a Transferred SpinCo Service Provider) who, as of the Distribution Time, is a former employee or independent contractor of the Company or any of its Subsidiaries, or (ii) any individual who is a Company Employee or Company Independent Contractor as of the Distribution Time or thereafter who ceases to be an employee or independent contractor of the Company or any of its Subsidiaries following the Distribution Time.
- (w) "Former SpinCo Service Provider" shall mean any individual who is a Transferred SpinCo Service Provider as of the Distribution Time and thereafter ceases to be an employee or independent contractor of the SpinCo Group following the Distribution Time.
- (x) "Inactive Employees" means any Offer Employee who is on (a) short-term disability or medical leave, (b) long-term disability, (c) leave under the Family Medical Leave Act of 1993 or a similar state or local law, (d) military leave, or (e) any other leave of absence, including temporary leave for purposes of jury or military duty, maternity or paternity leave or approved personal leave.

- (y) "Merger Agreement" shall mean the Agreement and Plan of Merger, dated as of September 25, 2022, by and among the Company, SpinCo, Parent and Merger Sub.
 - (z) "Non-parties" shall have the meaning set forth in Section 7.8(b).
- (aa) "Offer Employee" shall mean each SpinCo Employee identified on the Census as an "Offer Employee," as such Census may be updated as permitted pursuant to Section 6.1(h) of the Merger Agreement.
 - (bb) "Parent" shall have the meaning set forth in the Preamble.
- (cc) "Parent Common Stock" means Class A common stock, par value \$0.0001 per share, of Parent, and Class B common stock, par value \$0.0001 per share, of Parent.
 - (dd) "Parent Equity Plan" shall have the meaning set forth in Section 4.2.
 - (ee) "Party" and "Parties" shall have the meanings set forth in the Preamble.
 - (ff) "Plan Transition Date" shall mean the date that is the earlier to occur of (i) January 1, 2023 or (ii) such earlier or later date as agreed among the Parties.
 - (gg) "SpinCo" shall have the meaning set forth in the Preamble.
 - (hh) "SpinCo 401(k) Plan" shall have the meaning set forth in Section 3.3(b).
 - (ii) "SpinCo Benefit Plan" shall mean any Benefit Plan sponsored, maintained or contributed to exclusively by any member of the SpinCo Group.
 - (jj) "SpinCo Employee" shall mean each Business Employee who is not an Acquired Company Employee.
- (kk) "SpinCo Group" shall mean SpinCo, Design Reactor and each Person that is a direct or indirect Subsidiary of SpinCo as of the Distribution Time (but after giving effect to the Internal Reorganization), including the Transferred Entities (as defined in the Separation Agreement), and, following the Effective Time, Parent and each Person that becomes a Subsidiary of Parent or SpinCo thereafter, provided, however, that for the avoidance of doubt, no member of the Company Group shall be treated as a member of the SpinCo Group.
- (II) "SpinCo Independent Contractor" shall mean each Business Independent Contractor and each independent contractor currently engaged by the SpinCo Group.
- (mm) "SpinCo Service Provider" shall mean a Business Employee, a SpinCo Independent Contractor or a member of the board of directors of SpinCo, in each case, as of immediately prior to the Distribution Time.
 - (nn) " $\pmb{Separation}$ $\pmb{Agreement}$ " shall have the meaning set forth in the Recitals.

- (00) "Transferred Employees" shall have the meaning set forth in Section 2.2(b).
- (pp) "Transferred Independent Contractors" shall have the meaning set forth in Section 2.2(c).
- (qq) "Transferred SpinCo Service Providers" shall mean the Transferred Employees, the Transferred Independent Contractors and any other SpinCo Service Providers employed or engaged by the SpinCo Group as of the Distribution Time.
- 1.2 References; Interpretation. References in this Agreement to any gender include references to all genders, and references to the singular include references to the plural and vice versa. Unless the context otherwise requires, the words "include", "includes" and "including" when used in this Agreement shall be deemed to be followed by the phrase "without limitation". Unless the context otherwise requires, references in this Agreement to Articles, Sections, Annexes, Exhibits and Schedules shall be deemed references to Articles and Sections of, and Annexes, Exhibits and Schedules to, this Agreement. Unless the context otherwise requires, the words "hereof", "hereby" and "herein" and words of similar meaning when used in this Agreement refer to this Agreement in its entirety and not to any particular Article, Section or provision of this Agreement. The words "written request" when used in this Agreement shall include email. Reference in this Agreement to any time shall be to New York City, New York time unless otherwise expressly provided herein. Unless the context requires otherwise, references in this Agreement to the "Company" shall also be deemed to refer to the applicable member of the Company Group, references to "SpinCo" shall also be deemed to refer to the applicable member of the SpinCo Group (including, with respect to periods of time following the Effective Time, Parent), and, in connection therewith, any references to actions or omissions to be taken, or refrained from being taken, as the case may be, by the Company or SpinCo shall be deemed to require the Company, SpinCo or Parent, as the case may be, to cause the applicable members of the Company Group or the SpinCo Group, respectively, to take, or refrain from taking, any such action. In the event of any inconsistency or conflict which may arise in the application or interpretation of any of the definitions set forth in Section 1.1, for the purpose of determining what is and is not included in such definitions, any item explicitly included on a Sc

ARTICLE II GENERAL PRINCIPLES

- 2.1 Nature of Liabilities. All Liabilities assumed or retained by a member of the Company Group under this Agreement shall be "Inpixon Retained Liabilities" for purposes of the Separation Agreement. All Liabilities assumed or retained by a member of the SpinCo Group under this Agreement shall be "Enterprise Apps Liabilities" for purposes of the Separation Agreement.
 - 2.2 Transfers of Employees and Independent Contractors Generally.
- (a) The Company and SpinCo shall mutually update the Census from time to time following the date hereof and, in any event, no later than thirty (30) Business Days (as defined in the Merger Agreement) prior to the Distribution Time, to reflect new hire/engagements, terminations or other personnel changes occurring between the date hereof and the date of such update, as permitted pursuant to Section 6.1(h) of the Merger Agreement, and any such updates shall be provided to the Parent. Seven (7) Business Days prior to the Distribution Time, the Company and SpinCo shall provide Parent with a final updated Census reflecting new hires/engagements, terminations or other personnel changes occurring between the date the last update was provided and the Distribution Time, as permitted pursuant to Section 6.1(h) of the Merger Agreement.

(b) The Company and SpinCo will cooperate to cause each of the SpinCo Employees (other than each SpinCo Employee who is an Offer Employee) to be employed by a member of the SpinCo Group prior to the Distribution Time. The Company shall cooperate in good faith with Parent and its Affiliates to (i) make each Offer Employee reasonably accessible to Parent and its Affiliates to assist in efforts to secure offers of employment with each such Offer Employee and (ii) encourage (without the payment of additional compensation, benefits or other monetary or non-monetary incentives) each Offer Employee to accept an offer of employment with Parent or one of its Affiliates (including, following the Effective Time, SpinCo and the other members of the SpinCo Group). Parent or one of its Affiliates (including, following the Effective Time, SpinCo and the other members of the SpinCo Group) shall offer employment to the Offer Employees upon such terms and conditions of employment as set forth in Section 2.4, commencing at midnight local time on the Closing Date; provided that any Offer Employee who is an Inactive Employee immediately prior to the Closing Shall receive an offer of employment in accordance with, and subject to the terms of, this Section 2.2(b) below. The Company shall terminate the employment of all Offer Employees (other than the Inactive Employees) effective as of or immediately prior to the Closing and shall comply with, and hold Parent and its Affiliates harmless from, all legal or contractual requirements arising in connection with or as a result of such terminations of employment. The applicable date on which the each applicable Offer Employee commences employment with Parent or one of its Affiliates, either on or following the Closing Date as set forth in this Section 2.2(b), shall be the "Transfer Date" of such Offer Employee. As of the Distribution Time or, with respect to the Offer Employees, as of the applicable Transfer Date, the Company shall ensure that each SpinCo Employee is released from any post-termination or employment restrictions that would prohibit or restrict such SpinCo Employee from performing their duties for the SpinCo Group following the Distribution Time. All SpinCo Employees (other than the Offer Employees) and Acquired Company Employees who are employed by the SpinCo Group as of the Distribution Time shall continue to be employees of the SpinCo Group immediately after the Distribution Time. All SpinCo Employees (other than the Offer Employees) and Acquired Company Employees who are employed by the SpinCo Group as of the Distribution Time and all Offer Employees who commence employment with Parent or one of its Affiliate's pursuant to Parent's or its Affiliate's offer of employment on the Closing Date (or, with respect to Inactive Employees, such later date contemplated by this Section 2.2(b)) are referred to herein, collectively, as the "Transferred Employees". Notwithstanding the foregoing, with respect to any Inactive Employee, Parent's or its Affiliate's offer of employment shall be contingent on such Inactive Employee's return to active status within six (6) months following the Closing Date (or such longer period as required by applicable Law). The Company and its Affiliates shall continue to employ and shall remain responsible for any liabilities and obligations related to any Inactive Employee unless and until such individual becomes a Transferred Employee.

(c) The Company and SpinCo will cooperate to cause the engagement of each Business Independent Contractor set forth on **Schedule B** hereto to be transferred to a member of the SpinCo Group prior to the Distribution Time. As of the Closing, the Company shall ensure that each Business Independent Contractor is released from any engagement with the Company and its Subsidiaries (other than the SpinCo Group), including any restrictions or obligations that would prohibit or restrict such Person from performing such Person's work for the SpinCo Group following the Distribution Time. All SpinCo Independent Contractors who are engaged by the SpinCo Group as of the Distribution Time shall continue to be engaged by the SpinCo Group immediately after the Distribution Time and are referred to herein as the "**Transferred Independent Contractors**".

(d) The Company Group and SpinCo Group agree to execute, and to seek to have the applicable SpinCo Service Providers execute, such documentation, if any, as may be necessary to reflect the transfers or assignments, as applicable, described in this <u>Section 2.2</u>.

2.3 Assumption and Retention of Liabilities Generally.

- (a) Except as otherwise provided by this Agreement, on or prior to the Distribution Time, but in any case prior to the Distribution, or, with respect to each Offer Employee, on the applicable Transfer Date, SpinCo and the applicable members of the SpinCo Group shall accept, assume and agree faithfully to perform, discharge and fulfill all of the following Liabilities in accordance with their respective terms (each of which shall be considered an Enterprise Apps Liability), regardless of when or where such Liabilities arose or arise, whether the facts on which they are based occurred prior to or subsequent to the Distribution Time or, with respect to each Offer Employee, on the applicable Transfer Date, regardless of where or against whom such Liabilities are asserted or determined (including any Liabilities arising out of claims made by the Company's or SpinCo's respective directors, officers, employees, former employees, agents, Subsidiaries or Affiliates against any member of the Company Group or the SpinCo Group) or whether asserted or determined prior to the date hereof, and regardless of whether arising from or alleged to arise from negligence, recklessness, violation of applicable Law, fraud or misrepresentation by any member of the Company Group or the SpinCo Group, or any of their respective directors, officers, employees, former employees, agents, Subsidiaries or Affiliates:
 - (i) any and all wages, salaries, incentive compensation, commissions, bonuses and any other employee compensation or benefits payable to or on behalf of any Transferred SpinCo Service Providers after the Distribution Time, without regard to when such wages, salaries or other employee compensation or benefits are or may have been awarded or earned, provided that in no event shall SpinCo or SpinCo Group be liable for any equity or equity-related award that was granted to, accrued for, earned by or payable to a Transferred SpinCo Service Provider with respect to any period on or prior to the Distribution Time or any transaction bonus or other incentive compensation amounts that may be granted to accrued for, earned by or payable to a Transferred SpinCo Service Provider as a result of the consummation of the transactions contemplated by the Merger Agreement;
 - (ii) any and all Liabilities whatsoever with respect to claims under a SpinCo Benefit Plan, taking into account the SpinCo Benefit Plan's assumption of Liabilities with respect to Transferred SpinCo Service Providers that were originally the Liabilities of the corresponding Company Benefit Plan with respect to periods prior to the Distribution Time; and
 - (iii) any and all Liabilities expressly assumed or retained by any member of the SpinCo Group pursuant to this Agreement.

- (b) Except as otherwise provided by this Agreement, on or prior to the Distribution Time, but in any case prior to the Distribution, the Company and certain members of the Company Group designated by the Company shall accept, assume and agree faithfully to perform, discharge and fulfill all of the following Liabilities in accordance with their respective terms (each of which shall be considered a Inpixon Retained Liability), regardless of when or where such Liabilities arose or arise, or whether the facts on which they are based occurred prior to or subsequent to the Distribution Time, regardless of where or against whom such Liabilities are asserted or determined (including any Liabilities arising out of claims made by the Company's or SpinCo's respective directors, officers, employees, former employees, agents, Subsidiaries or Affiliates against any member of the Company Group or the SpinCo Group) or whether asserted or determined prior to the date hereof, and regardless of whether arising from or alleged to arise from negligence, recklessness, violation of applicable Law, fraud or misrepresentation by any member of the Company Group or the SpinCo Group, or any of their respective directors, officers, employees, former employees, agents, Subsidiaries or Affiliates:
 - (i) any and all wages, salaries, incentive compensation, equity compensation, commissions, bonuses and any other employee compensation or benefits payable to or on behalf of any Company Employees, Company Independent Contractors, Former Company Service Providers, and any SpinCo Service Providers who do not become Transferred SpinCo Service Providers, without regard to when such wages, salaries, incentive compensation, equity compensation, commissions, bonuses or other employee compensation or benefits are or may have been awarded or earned;
 - (ii) any and all Liabilities whatsoever with respect to claims under a Company Benefit Plan, taking into account a corresponding SpinCo Benefit Plan's assumption of Liabilities with respect to Transferred Employees that were originally the Liabilities of such Company Benefit Plan with respect to periods prior to the Distribution Time; provided that, the Company and the Company Group shall be liable for all equity or equity-related awards, that were granted to, accrued for, earned by or payable to a Transferred SpinCo Service Provider with respect to any period on or prior to the Distribution Time and any transaction bonus or other incentive compensation amounts that may be granted to accrued for, earned by or payable to a Transferred SpinCo Service Provider as a result of the consummation of the transactions contemplated by the Merger Agreement ("Transaction Bonuses");
 - (iii) any and all Liabilities with respect to any Company Employees, Company Independent Contractors, Former Company Service Providers, and any SpinCo Service Providers who do not become Transferred SpinCo Service Providers; and
 - (iv) any and all Liabilities expressly assumed or retained by any member of the Company Group pursuant to this Agreement.
- (c) To the extent that this Agreement does not address particular Liabilities under any Benefit Plan and the Parties later determine that they should be allocated in connection with the Distribution, the Parties shall agree in good faith on the allocation, taking into account the handling of comparable Liabilities under this Agreement.

- (d) The Parties shall promptly reimburse one another, upon reasonable request of the Party requesting reimbursement and the presentation by such Party of such substantiating documentation as the other Party shall reasonably request, for the cost of any obligations or Liabilities satisfied or assumed by the Party requesting reimbursement or its Affiliates that are, or that have been made pursuant to this Agreement, the responsibility of the other Parties or any of its Affiliates.
- (e) Notwithstanding any provision of this Agreement or the Separation Agreement to the contrary, SpinCo shall, or shall cause one or more members of the SpinCo Group to, accept, assume (or, as applicable, retain) and perform, discharge and fulfill all Liabilities that have been accepted, assumed or retained under this Agreement.
- 2.4 Treatment of Compensation and Benefit Plans; Terms of Employment. Except as otherwise (i) required by applicable Law, or (ii) expressly provided for in this Agreement, for a period of twelve (12) months following the Distribution Time (or if shorter, during the period of employment), SpinCo shall, or shall cause a member of the SpinCo Group to provide or cause to be provided to each SpinCo Employee (A) a base salary or hourly wage rate, as applicable, that is at least equal to the base salary or hourly wage rate provided to such SpinCo Employee immediately prior to the Distribution Time, (B) subject to Section 5.1, a cash incentive or sales commission opportunity no less favorable than the cash incentive or sales commission opportunity in effect for such SpinCo Employee, if any, immediately prior to the Distribution Time, (C) health, welfare and retirement benefits that are substantially similar in the aggregate to those provided to such SpinCo Employee immediately prior to the Distribution Time, and (D) severance benefits (including severance payments, transition payments and continued health coverage but excluding any equity or equity-related payments or benefits) that are substantially similar to those provided to such SpinCo Employee immediately prior to the Distribution Time.
- 2.5 <u>Participation in Company Benefit Plans</u>. Except as otherwise provided pursuant to this Agreement or as required by Applicable Law, effective no later than the Plan Transition Date, (i) SpinCo and each member of the SpinCo Group, to the extent applicable, shall cease to be a participating company in any Company Benefit Plan and (ii) each then active SpinCo Employee shall cease to participate in, be covered by, accrue benefits under, be eligible to contribute to or have any rights under any Company Benefit Plan (except to the extent of previously accrued obligations that remain a Liability of any member of the Company Group pursuant to this Agreement).

2.6 Service Recognition.

(a) From and after the Distribution Time, and in addition to any applicable obligations under applicable Law, SpinCo shall, and shall cause each member of the SpinCo Group to, give each SpinCo Employee full credit for purposes of eligibility, vesting, and determination of level of benefits (other than with respect to any equity or equity-related compensation, defined benefit pension benefits or post-termination health or welfare benefits) under any SpinCo Benefit Plan for such SpinCo Employee's prior service with any member of the Company Group or SpinCo Group or any predecessor thereto, to the same extent such service was recognized by the applicable Company Benefit Plan; provided, that, such service shall not be recognized to the extent it would result in the duplication of benefits.

- (b) Except to the extent prohibited by applicable Law, effective as of the Plan Transition Date with respect to any applicable SpinCo Benefit Plan that is a health or welfare benefit plan: (i) SpinCo shall use commercially reasonable efforts to waive or cause to be waived all limitations as to preexisting conditions or waiting periods with respect to participation and coverage requirements applicable to each SpinCo Employee under any SpinCo Benefit Plan in which SpinCo Employees participate (or are eligible to participate) to the same extent that such conditions and waiting periods were satisfied or waived under an analogous Company Benefit Plan, and (ii) SpinCo shall use commercially reasonable efforts to provide or cause each SpinCo Employee to be provided with credit for any co-payments, deductibles or other out-of-pocket amounts paid during the plan year in which the SpinCo Employees become eligible to participate in the SpinCo Benefit Plans in satisfying any applicable co-payments, deductibles or other out-of-pocket requirements under any such plans for such plan year.
- 2.7 <u>Assignment of Restrictive Covenants</u>. The Company hereby assigns to the SpinCo Group, as of the Distribution Time, the rights of the Company Group under any nondisclosure, noncompetition, non-solicitation, non-hire, non-disparagement, intellectual property assignment or similar agreement between the Company or its Subsidiaries (other than any member of the SpinCo Group), on the one hand, and any Transferred Service Provider, on the other hand, to the extent that such agreement relates to the Enterprise Apps Business.
- 2.8 WARN. Notwithstanding anything set forth in this Agreement to the contrary, none of the transactions contemplated by or undertaken by this Agreement is intended to and shall not constitute or give rise to an "employment loss" or employment separation within the meaning of the federal Worker Adjustment and Retraining Notification (WARN) Act, or any other federal, state, or local law or legal requirement addressing mass employment separations.
- 2.9 <u>Communication</u>. Any written or oral communications proposed to be delivered to Business Employees regarding such Business Employees' level of (or rights with respect to) continued employment or benefits or compensation at or after the Distribution Time, in connection with such Business Employees' rights and obligations contained in this Agreement (if any), or otherwise respecting any changes or potential changes in employee benefit plans, practices or procedures that may or will occur in connection with or following the transactions contemplated by the Merger Agreement shall be subject to the review and prior consent of Parent (such consent not to be unreasonably withheld, conditioned or delayed).
- 2.10 No Termination; No Change in Control. No Company Employee or SpinCo Employee shall be deemed to (a) terminate employment or service solely by virtue of the consummation of the Distribution, any transfer of employment or other service relationship contemplated hereby, or any related transactions or events contemplated by the Separation Agreement, this Agreement, the Merger Agreement, or any Ancillary Agreement, or (b) become entitled to any severance, termination, separation or similar rights, payments or benefits, whether under any Benefit Plan, the Company Equity Plans, any Company Individual Agreement or any other compensatory agreement or arrangement maintained by the Company or SpinCo or otherwise, in connection with any of the foregoing. The Parties hereto agree that none of the transactions contemplated by the Separation Agreement, the Merger Agreement, or this Agreement, constitutes a "change in control," "change of control" or similar term, as applicable, within the meaning of any Benefit Plan, the Company Equity Plans, any Company Individual Agreement or any other compensatory agreement or arrangement maintained by the Company or SpinCo.

ARTICLE III CERTAIN BENEFIT PLAN PROVISIONS

3.1 Health and Welfare Benefit Plans.

- (a)(i) Effective as of the Plan Transition Date, the Company shall or shall cause a member of the Company Group to cause the participation of each thenactive SpinCo Employee who is a participant in a Company Benefit Plan to cease; provided, however, that in no event shall the foregoing require the termination or cessation of any Benefit Plan of any Transferred Entities to the extent that such Transferred Entity may retain such Benefit Plans in effect for the benefit of SpinCo Employees and (ii) SpinCo shall or shall cause a member of the SpinCo Group to (A) have in effect, no later than the Business Day immediately prior to the Plan Transition Date, SpinCo Benefit Plans providing health and welfare benefits for the benefit of each such SpinCo Employee with terms that are substantially similar in the aggregate to those provided to the applicable SpinCo Employee under the Company Benefit Plans immediately prior to the date on which such SpinCo Benefit Plans become effective; and (B) effective on and after the Plan Transition Date, fully perform, pay and discharge all claims of SpinCo Employees or Former SpinCo Service Providers, for claims incurred under such SpinCo Benefit Plans and pay or reimburse the Company for any claims incurred under any Company Benefit Plan that is a health or welfare plan (to the extent not fully covered by insurance) on or prior to the date on which such SpinCo Benefit Plans become effective, regardless of whether any such claim was presented for payment prior to, on or after such date.
- (a) Without duplication of amounts otherwise already covered in this Agreement or the Transition Services Agreement, the applicable member of the SpinCo Group shall reimburse the Company or the applicable Company Benefit Plan in the ordinary course of business consistent with past practice for any premiums and its proportionate share of any administrative or services costs related to SpinCo Employees or Former SpinCo Service Providers solely with respect to any period prior to the Plan Transition Date paid by a Company Benefit Plan (whether prior to or after the Distribution Time) and not charged back to the appropriate and applicable member of the SpinCo Group prior to the Plan Transition Date.
- (b) Notwithstanding anything to the contrary in this Section 3.1, SpinCo Employees will continue to be considered to be "participants" in any Company Benefit Plan that is either a health care flexible spending account program or a dependent-care flexible spending account program for the duration of any grace period and/or claims run-out period following the calendar year in which the Plan Transition Date occurs (in either case, solely as provided under the terms of such Company Benefit Plans), provided that such SpinCo Employees will be considered to be participants solely for purposes of utilizing such grace period and/or claims run-out period; will not be allowed to make any deferral or contribution elections under such Company Benefit Plans beyond the Plan Transition Date; and will cease to be participants in such Company Benefit Plans upon the expiration of any grace period and/or claims run-out period.

3.2 Disability.

- (a) To the extent any Transferred Employee is, as of the Plan Transition Date, receiving payments as part of any short-term disability program that is part of a Company Benefit Plan, such Transferred Employee's rights to continued short-term disability benefits (a) will end under any Company Benefit Plan as of the Plan Transition Date; and (b) all remaining rights will be recognized under a SpinCo Benefit Plan as of the Plan Transition Date, and the remainder (if any) of such Transferred Employee's short-term disability benefits will be paid by a SpinCo Benefit Plan. In the event that any Transferred Employee described above shall have any dispute with the short-term disability benefits they are receiving under a SpinCo Benefit Plan, any and all appeal rights of such employees shall be realized through the SpinCo Benefit Plan (and any appeal rights such Transferred Employee may have under any Company Benefit Plan will be limited to benefits received and time periods occurring prior to the Plan Transition Date). Any Transferred Employee or Former SpinCo Service Provider who is receiving short-term disability benefits under a Company Benefit Plan as of the Plan Transition Date and thereafter becomes entitled to long-term disability benefits upon the expiration of such short-term disability period (whether under a Company Benefit Plan or SpinCo Benefit Plan), shall be provided long-term disability benefits under the long-term disability plan which is a Company Benefit Plan.
- (b) For any Business Employee who is, as of the Distribution Time, receiving payments as part of any long-term disability program that is part of a Company Benefit Plan, and has been receiving payments from such plan for twelve (12) months or fewer before the Distribution Time, to the extent such Business Employee may have any "return to work" rights under the terms of such Company Benefit Plan, such Business Employee's eligibility for re-employment shall be with SpinCo or a member of the SpinCo Group, subject to availability of a suitable position (with such availability to be determined in the sole discretion by SpinCo or the applicable member of the SpinCo Group), provided however that, notwithstanding the foregoing, no Business Employee described in this subsection will be eligible for re-employment as described in this subsection after the six (6) month anniversary of the Distribution Time (or such longer period as required by applicable Law).

3.3 401(k) Plans.

- (a) From the Distribution Time and continuing until the 401(k) Plan Transition Date, SpinCo shall be an "adopting employer" (as defined in the Company 401(k) Plan) and the Company 401(k) Plan shall provide for the SpinCo Group to continue to participate in the Company 401(k) Plan for the benefit of SpinCo Employees, and the Company consents to such adoption and maintenance, in accordance with the terms of the Company 401(k) Plan.
- (b) Effective no later than the 401(k) Plan Transition Date, (i) SpinCo shall establish a defined contribution savings plan and related trust that satisfies the requirements of Sections 401(a) and 401(k) of the Code in which each SpinCo Employee who participated in the Company 401(k) Plan immediately prior thereto shall be immediately eligible to participate (the "SpinCo 401(k) Plan"), with terms that are substantially similar to those provided to such SpinCo Employees by the Company 401(k) Plan immediately prior to the date on which such SpinCo 401(k) Plan become effective, (ii) the Company shall or shall cause a member of the Company Group to cause the active participation of each SpinCo Employee who is a participant in the Company 401(k) Plan to cease as of the date on which the SpinCo 401(k) Plan becomes effective, and (iii) as soon as practicable after the SpinCo 401(k) Plan becomes effective, subject to the consent of the SpinCo 401(k) Plan administrator and reasonable proof of qualification of the Company 401(k) Plan, the Company shall cause the accounts (including any outstanding participant loan balances) in the Company 401(k) Plan attributable to SpinCo Employees and all of the assets in the Company 401(k) Plan related thereto to be transferred to the SpinCo 401(k) Plan pursuant to a trustee-to-trustee transfer that meets the requirements of Section 414(l) of the Code.

- (c)The Company shall retain all accounts and all assets and Liabilities relating to the Company 401(k) Plan in respect of each Former SpinCo Service Provider whose employment terminated prior to the 401(k) Plan Transition Date.
- 3.4 Chargeback of Certain Costs. Without duplication of amounts otherwise already covered in this Agreement or the Transition Services Agreement, nothing contained in this Agreement shall limit the Company's ability to charge back any Liabilities that it incurs in respect of any SpinCo Service Provider under a Company Benefit Plan which is a retirement plan or health or welfare benefit plan to any of its operating companies in the ordinary course of business consistent with its past practices. Subject, and in addition, to the foregoing, the Company shall allocate and charge back to SpinCo or a member of the SpinCo Group (without duplication) its proportionate share of Liabilities (other than those arising from the Company's or its agent's gross misconduct or negligence) that the Company incurs by reason of the continued participation of SpinCo Employees, SpinCo Independent Contractors and Former SpinCo Service Providers in such Company Benefit Plans following the Distribution Time (which Liabilities shall, for the avoidance of doubt, be subject to reimbursement under Section 2.3(d) of this Agreement).

ARTICLE IV EQUITY INCENTIVE AWARDS

- 4.1 Company Equity Plans; Company Equity Awards. Notwithstanding anything to the contrary herein, the Company shall retain all liabilities related to stock options and other equity or equity-related awards of the Company granted to any SpinCo Service Provider or Company Service Provider under the Company Equity Plans or otherwise, which shall at all times constitute an Inpixon Retained Liability.
- 4.2 Parent Equity Plan. Prior to the Effective Time, Parent shall approve and adopt, subject to receipt of Acquiror Stockholder Approval, an incentive equity plan (the "Parent Equity Plan"); in form and substance reasonably acceptable to the Company and SpinCo in consultation with Parent, and effective as of the Effective Time. The Parent Equity Plan will provide for the grant of awards of Parent Common Stock with a reserve of shares equal to 35% of the total pool of shares available for issuance under the Parent Equity Plan within six (6) months after the Effective Time for the purpose of granting such awards to Transferred SpinCo Service Providers (in such individual amounts and with such vesting schedules as determined by the Compensation Committee of the Board of Directors of Parent).

ARTICLE V ADDITIONAL MATTERS

5.1 <u>Cash Incentive Programs</u>. SpinCo shall assume all Liabilities with respect to all cash incentive compensation, commissions or similar cash payments earned by or payable to SpinCo Employees in the year in which the Distribution Time occurs and thereafter. The Company shall retain all Liabilities with respect to any cash incentive compensation, commissions or similar cash payments earned by or payable to Company Employees for the year in which the Distribution Time occurs and thereafter and any Transaction Bonuses.

- 5.2 Severance. SpinCo shall be solely responsible for all Liabilities in respect of all costs arising out of payments and benefits relating to the termination or alleged termination of any Transferred Employee's employment that occurs after the Distribution Time, including as a result of, in connection with or following the consummation of the transactions contemplated by the Separation Agreement or Merger Agreement, including any amounts required to be paid (including any payroll or other taxes), and the costs of providing benefits, under any applicable severance, separation, redundancy, termination or similar plan, program, practice, contract, agreement, law or regulation (such benefits to include any medical or other welfare benefits, outplacement benefits, accrued vacation, and taxes).
- 5.3 <u>Time-Off Benefits</u>. Unless otherwise required under applicable Law (or as would result in duplication of benefits), SpinCo shall (i) credit each SpinCo Employee who becomes a Transferred Employee with the amount of accrued but unused vacation time, paid time-off and other time-off benefits as such SpinCo Employee had with the Company Group as of immediately before the date on which the employment of the SpinCo Employee transfers to SpinCo and (ii) permit each such SpinCo Employee to use such accrued but unused vacation time, paid time off and other time-off benefits in accordance with the terms and conditions of Parent's applicable policies. All such Liabilities shall be included in the definition of "Enterprise Apps Liabilities".
- 5.4 Workers' Compensation Liabilities. Effective no later than the Distribution Time, SpinCo shall assume all Liabilities for Transferred SpinCo Service Providers related to any and all workers' compensation injuries, incidents, conditions, claims or coverage, whenever incurred (including claims incurred prior to the Distribution Time but not reported until after the Distribution Time), and SpinCo shall be fully responsible for the administration, management and payment of all such claims and satisfaction of all such Liabilities. Notwithstanding the foregoing, if SpinCo is unable to assume any such Liability or the administration, management or payment of any such claim solely because of the operation of applicable Law, the Company shall retain such Liabilities and SpinCo shall reimburse and otherwise fully indemnify the Company for all such Liabilities, including the costs of administering the plans, programs or arrangements under which any such Liabilities have accrued or otherwise arisen.
- 5.5 COBRA Compliance. The Company shall retain responsibility for compliance with the health care continuation requirements of COBRA with respect to SpinCo Employees or Former SpinCo Service Providers who, on or prior to the Plan Transition Date, were covered under a Company Benefit Plan and who had incurred a COBRA qualifying event and were eligible to elect COBRA under a Company Benefit Plan on or prior to the Plan Transition Date. SpinCo shall be responsible for administering compliance with the health care continuation requirements of COBRA, and the corresponding provisions of the SpinCo Benefit Plans with respect to SpinCo Employees and their covered dependents who incur a COBRA qualifying event or loss of coverage at any time after the Plan Transition Date.
- 5.6 <u>Code Section 409A</u>. Notwithstanding anything in this Agreement to the contrary, the Parties shall negotiate in good faith regarding the need for any treatment different from that otherwise provided herein with respect to the payment of compensation to ensure that the treatment of such compensation does not cause the imposition of a Tax under Section 409A of the Code. In no event, however, shall any Party be liable to another in respect of any Taxes imposed under, or any other costs or Liabilities relating to, Section 409A of the Code.

- 5.7 Payroll Taxes and Reporting. The Parties shall, to the extent practicable, (i) treat SpinCo or a member of the SpinCo Group as a "successor employer" and the Company (or the appropriate member of the Company Group) as a "predecessor," within the meaning of Sections 3121(a)(1) and 3306(b)(1) of the Code, with respect to SpinCo Employees for purposes of Taxes imposed under the United States Federal Unemployment Tax Act or the United States Federal Insurance Contributions Act, and (ii) cooperate with each other to avoid, to the extent reasonably practicable, the filing of more than one IRS Form W-2 with respect to each SpinCo Employee for the calendar year in which the Distribution Time occurs.
- 5.8 <u>Regulatory Filings</u>. Subject to applicable Law and the Tax Matters Agreement, the Company shall retain responsibility for all employee-related regulatory filings for reporting periods ending at or prior to the Distribution Time, except for Equal Employment Opportunity Commission EEO-1 reports and affirmative action program (AAP) reports and responses to Office of Federal Contract Compliance Programs (OFCCP) submissions, for which the Company shall provide data and information (to the extent permitted by applicable Laws) to SpinCo, which shall be responsible for making such filings in respect of SpinCo Employees.
- 5.9 <u>Certain Requirements</u>. Notwithstanding anything in this Agreement to the contrary, if applicable Law requires that any assets or Liabilities be retained by the Company Group or transferred to or assumed by the SpinCo Group in a manner that is different from that set forth in this Agreement, such retention, transfer or assumption shall be made in accordance with the terms of such applicable Law and shall not be made as otherwise set forth in this Agreement and the Parties shall reasonably cooperate to adjust for any related economic consequences.

ARTICLE VI OBLIGATIONS OF PARENT AND MERGER SUB

6.1 Obligations of Parent. Following the Effective Time, Parent agrees to cause, and to take all actions to enable, SpinCo and the members of the SpinCo Group to adhere to each provision of this Agreement which requires an act on the part of SpinCo or any member of the SpinCo Group or any of its or their Affiliates, and to cause or enable SpinCo and the SpinCo Group to comply with their obligations to provide or establish compensation or benefits to SpinCo Service Providers in accordance with this Agreement pursuant to a Benefit Plan sponsored or maintained by Parent or any of its Subsidiaries.

ARTICLE VII GENERAL AND ADMINISTRATIVE

- 7.1 Employer Rights. Nothing in this Agreement shall be deemed to be an amendment to any Company Benefit Plan or SpinCo Benefit Plan or to prohibit any member of the Company Group or SpinCo Group, as the case may be, from amending, modifying or terminating any Company Benefit Plan or SpinCo Benefit Plan at any time within its sole discretion.
- 7.2 Effect on Employment. Nothing in this Agreement is intended to or shall confer upon any employee or former employee of the Company, SpinCo or any of their respective Affiliates any right to continued employment, or any recall or similar rights to any such individual on layoff or any type of approved leave.

- 7.3 <u>Consent of Third Parties</u>. If any provision of this Agreement is dependent on the consent of any third party and such consent is withheld, the Parties shall use their reasonable efforts to implement the applicable provisions of this Agreement to the fullest extent practicable. If any provision of this Agreement cannot be implemented due to the failure of such third party to consent, the Parties hereto shall negotiate in good faith to implement the provision (as applicable) in a mutually satisfactory manner.
- 7.4 Access to Employees. On and after the Distribution Time, the Parties shall, or shall cause each of their respective Affiliates to, make available to each other those of their employees who may reasonably be needed in order to defend or prosecute any legal or administrative action (other than a legal action among the Parties) to which any employee or director of the Company Group or the SpinCo Group or any Company Benefit Plan or SpinCo Benefit Plan is a party and which relates to a Company Benefit Plan or SpinCo Benefit Plan. The Party to whom an employee is made available in accordance with this Section 7.4 shall pay or reimburse the other Parties for all reasonable expenses which are incurred by such other Party in connection therewith, including all reasonable travel, lodging, and meal expenses, but excluding any amount for such employee's time spent in connection herewith.
- 7.5 Beneficiary Designation/Release of Information/Right to Reimbursement. To the extent permitted by applicable Law and except as otherwise provided for in this Agreement or any agreement between SpinCo and the provider of its applicable Benefit Plan, all beneficiary designations, authorizations for the release of information and rights to reimbursement made by or relating to SpinCo Employees under Company Benefit Plan shall be transferred to and be in full force and effect under the corresponding SpinCo Benefit Plan until such beneficiary designations, authorizations or rights are replaced or revoked by, or no longer apply, to the relevant SpinCo Employee.
- 7.6 No Third Party Beneficiaries. This Agreement is solely for the benefit of the Parties and, except to the extent otherwise expressly provided herein, nothing in this Agreement, express or implied, is intended to confer any rights, benefits, remedies, obligations or Liabilities under this Agreement upon any Person, including any SpinCo Employee or other current or former employee, officer, director or contractor of the Company Group or SpinCo Group, other than the Parties and their respective successors and assigns.
- 7.7 Employee Benefits Administration. At all times following the date hereof, the Parties will cooperate in good faith as necessary to facilitate the administration of employee benefits and the resolution of related employee benefit claims with respect to SpinCo Employees, Former SpinCo Service Providers and employees and other service providers of the Company, as applicable, including with respect to the provision of employee level information necessary for the other Parties to manage, administer, finance and file required reports with respect to such administration.

7.8 Audit Rights With Respect to Information Provided.

(a) Each Party, and their duly authorized representatives, shall have the right to conduct reasonable audits with respect to all information required to be provided to it by the other Parties under this Agreement. The Party conducting the audit (the "Auditing Party") may adopt reasonable procedures and guidelines for conducting audits and the selection of audit representatives under this Section 7.8. The Auditing Party shall have the right to make copies of any records at its expense, subject to any restrictions imposed by applicable laws and to any confidentiality provisions set forth in the Separation Agreement, which are incorporated by reference herein.

The Party being audited shall provide the Auditing Party's representatives with reasonable access during normal business hours to its operations, computer systems and paper and electronic files, and provide workspace to its representatives. After any audit is completed, the Party being audited shall have the right to review a draft of the audit findings and to comment on those findings in writing within thirty (30) Business Days after receiving such draft.

- (b) The Auditing Party's audit rights under this Section 7.8 shall include the right to audit, or participate in an audit facilitated by the Party being audited, of any Subsidiaries and Affiliates of the Party being audited and to require the other Parties to request any benefit providers and third parties with whom the Party being audited has a relationship, or agents of such Party, to agree to such an audit to the extent any such Persons are affected by or addressed in this Agreement (collectively, the "Non-parties"). The Party being audited shall, upon written request from the Auditing Party, provide an individual (at the Auditing Party's expense) to supervise any audit of a Non-party. The Auditing Party shall be responsible for supplying, at the Auditing Party's expense, additional personnel sufficient to complete the audit in a reasonably timely manner. The responsibility of the Party being audited shall be limited to providing, at the Auditing Party's expense, a single individual at each audited site for purposes of facilitating the audit.
- 7.9 <u>Cooperation</u>. Each of the Parties hereto will use its commercially reasonable efforts to share information and promptly take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Laws to consummate the transactions contemplated by this Agreement.

ARTICLE VIII MISCELLANEOUS

- 8.1 Entire Agreement. This Agreement, the Separation Agreement, the Merger Agreement, and the Ancillary Agreements, including the Exhibits and Schedules thereto, shall constitute the entire agreement among the Parties with respect to the subject matter hereof and shall supersede all previous negotiations, commitments, course of dealings and writings with respect to such subject matter.
- 8.2 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts (including by electronic or .pdf transmission), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Delivery of any signature page by facsimile, electronic or .pdf transmission shall be binding to the same extent as an original signature page.
- 8.3 <u>Survival of Agreements</u>. Except as otherwise contemplated by this Agreement, all covenants and agreements of the Parties contained in this Agreement shall survive the Distribution Time and Effective Time and remain in full force and effect in accordance with their applicable terms.

8.4 Notices. All notices and other communications among the Parties shall be in writing and shall be deemed to have been duly given (a) when delivered in person, (b) when delivered after posting in the national mail having been sent registered or certified mail return receipt requested, postage prepaid, (c) when delivered by FedEx or other internationally recognized overnight delivery service or (d) when delivered by facsimile (solely if receipt is confirmed) or email (so long as the sender of such email does not receive an automatic reply from the recipient's email server indicating that the recipient did not receive such email), addressed as follows:

To the Company:

Inpixon 2479 E. Bayshore Road, Suite 195 Palo Alto, California 94303

Attention: Nadir Ali, Chief Executive Officer

Email: nadir.ali@inpixon.com

With a copy (which shall not constitute notice) to:

Mitchell Silberberg & Knupp LLP 437 Madison Ave., 25th Floor New York, New York 10022 Attention: Blake J. Baron Email: bjb@msk.com

To SpinCo:

CXApp Holding Corp. 2479 E. Bayshore Road, Suite 195 Palo Alto, California 94303

Attention: Nadir Ali, Chief Executive Officer

Email: nadir.ali@inpixon.com

With a copy (which shall not constitute notice) to:

Mitchell Silberberg & Knupp LLP 437 Madison Ave., 25th Floor New York, New York 10022 Attention: Blake J. Baron Email: bjb@msk.com

To Parent or Merger Sub:

KINS Technology Group Inc. Four Palo Alto Square, Suite 200 3000 El Camino Real Palo Alto, California 94306

Attention: Khurram Sheikh, Chief Executive Officer

Email: khurram@kins-tech.com

With a copy (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP 525 University Avenue, Suite 1400 Palo Alto, California 94301

Attention: Michael Mies

Email: michael.mies@skadden.com

or to such other address or addresses as the Parties may from time to time designate in writing by like notice.

- 8.5 Consents. Any consent required or permitted to be given by any Party to the other Parties under this Agreement shall be in writing and signed by the Party giving such consent and shall be effective only against such Party (and its Group).
- 8.6 <u>Assignment</u>. This Agreement shall not be assignable, in whole or in part, directly or indirectly, by any Party hereto without the prior written consent of the other Parties, and any attempt to assign any rights or obligations arising under this Agreement without such consent shall be void. Notwithstanding the foregoing, and subject to any restrictions on assignment by SpinCo pursuant to Article IV of the Tax Matters Agreement, this Agreement shall be assignable to (i) with respect to the Company, an Affiliate of the Company, and with respect to SpinCo, and Affiliate of SpinCo, or (ii) a bona fide third party in connection with a merger, reorganization, consolidation or the sale of all or substantially all the assets of a party hereto so long as the resulting, surviving or transferee entity assumes all the obligations of the relevant party hereto by operation of Law or pursuant to an agreement in form and substance reasonably satisfactory to the other Parties to this Agreement; provided however that in the case of each of the preceding clauses (i) and (ii), no assignment permitted by this <u>Section 8.6</u> shall release the assigning Party from liability for the full performance of its obligations under this Agreement.
- 8.7 <u>Successors and Assigns</u>. The provisions of this Agreement and the obligations and rights hereunder shall be binding upon, inure to the benefit of and be enforceable by (and against) the Parties and their respective successors and permitted assigns.
- 8.8 <u>Termination and Amendment</u>. This Agreement may be amended or modified, in whole or in part, only by a duly authorized agreement in writing executed by the Parties in the same manner (but not necessarily by the same Persons) as this Agreement, and which makes reference to this Agreement. This Agreement shall terminate automatically without any further action of the Parties upon a termination of the Merger Agreement, and no Party will have any further obligations to the other Parties hereunder.
- 8.9 <u>Subsidiaries</u>. Each of the Parties shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth herein to be performed by any Subsidiary of such Party or by any entity that becomes a Subsidiary of such Party at and after the Distribution Time, to the extent such Subsidiary remains a Subsidiary of the applicable Party.
- 8.10 <u>Title and Headings</u>. Titles and headings to sections herein are inserted for the convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.

8.11 Governing Law. Except to the extent that federal law applies, this Agreement, and all claims, disputes, controversies or causes of action (whether in contract, tort, equity or otherwise) that may be based upon, arise out of or relate to this Agreement (including any schedule or exhibit hereto) or the negotiation, execution or performance of this Agreement (including any claim, dispute, controversy or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed by and construed in accordance with the Laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware. Each of the Parties agrees that any Action related to this agreement shall be brought exclusively in the Chosen Courts. By executing and delivering this Agreement, each of the Parties irrevocably: (a) accepts generally and unconditionally submits to the exclusive jurisdiction of the Chosen Courts for any Action relating to this Agreement; (b) waives any objections which such party may now or hereafter have to the laying of venue of any such Action contemplated by this Section 8.11 and hereby further irrevocably waives and agrees not to plead or claim that any such Action has been brought in an inconvenient forum; (c) agrees that it will not attempt to deny or defeat the personal jurisdiction of the Chosen Courts by motion or other request for leave from any such court; (d) agrees that it will not bring any Action contemplated by this Section 8.11 in any court other than the Chosen Courts; (e) agrees that service of all process, including the summons and complaint, in any Action may be made by registered or certified mail, return receipt requested, to such party at their respective addresses provided in accordance with Section 8.4 or in any other manner permitted by Law; and (f) agrees that service as provided in the preceding clause (e) is sufficient to confer personal jurisdiction over such party in the Action, and otherwise constitutes effective and binding service in every respect. Each of the parties hereto agrees that a final judgment in any Action in a Chosen Court as provided above may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law, and each party further agrees to the non-exclusive jurisdiction of the Chosen Courts for the enforcement or execution of any such judgment.

8.12 WAIVER OF JURY TRIAL. THE PARTIES HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVE THEIR RIGHT TO TRIAL BY JURY IN ANY JUDICIAL PROCEEDING IN ANY COURT RELATING TO ANY DISPUTE, CONTROVERSY OR CLAIM ARISING OUT OF, RELATING TO OR IN CONNECTION WITH THIS AGREEMENT (INCLUDING ANY SCHEDULE OR EXHIBIT HERETO) OR THE BREACH, TERMINATION OR VALIDITY OF THIS AGREEMENT OR THE NEGOTIATION, EXECUTION OR PERFORMANCE OF THIS AGREEMENT. NO PARTY TO THIS AGREEMENT SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER LITIGATION PROCEDURE BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY RELATED INSTRUMENTS. NO PARTY WILL SEEK TO CONSOLIDATE ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 8.12. NO PARTY HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PARTY THAT THE PROVISIONS OF THIS SECTION 8.12 WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.

- 8.13 Severability. If any provision of this Agreement, or the application of any such provision to any Person or circumstance, shall be held invalid, illegal or unenforceable in any respect by a court of competent jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision hereof. The Parties further agree that if any provision contained herein is, to any extent, held invalid or unenforceable in any respect under the Laws governing this Agreement, they shall take any actions necessary to render the remaining provisions of this Agreement valid and enforceable to the fullest extent permitted by Law and, to the extent necessary, shall amend or otherwise modify this Agreement to replace any provision contained herein that is held invalid or unenforceable with a valid and enforceable provision giving effect to the intent of the Parties.
- 8.14 <u>Interpretation</u>. The Parties have participated jointly in the negotiation and drafting of this Agreement. This Agreement shall be construed without regard to any presumption or rule requiring construction or interpretation against the Party drafting or causing any instrument to be drafted.
- 8.15 No Duplication; No Double Recovery. Nothing in this Agreement is intended to confer to or impose upon any Party a duplicative right, entitlement, obligation or recovery with respect to any matter arising out of the same facts and circumstances.
- 8.16 No Waiver. No failure to exercise and no delay in exercising, on the part of any Party, any right, remedy, power or privilege hereunder shall operate as a waiver hereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.
- 8.17 No Admission of Liability. The allocation of Assets and Liabilities herein is solely for the purpose of allocating such Assets and Liabilities among the Parties and is not intended as an admission of liability or responsibility for any alleged Liabilities vis-à-vis any third party, including with respect to the Liabilities of any non-wholly owned subsidiary of any Party.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed as of the day and year first above written.

INPIXON

By: /s/ Nadir Ali

Name: Nadir Ali Title: CEO

CXAPP HOLDING CORP.

By: /s/ Nadir Ali

Name: Nadir Ali

Title: Chief Executive Officer

KINS TECHNOLOGY GROUP INC.

By: /s/ Khurram Sheikh

Name: Khurram P. Sheikh Title: Chief Executive Officer

KINS MERGER SUB INC.

By: /s/ Khurram Sheikh

Name: Khurram P. Sheikh

Title: Director

TAX MATTERS AGREEMENT

by and among

KINS TECHNOLOGY GROUP INC.

INPIXON

and

CXAPP HOLDING CORP.

Dated as of March 14, 2023

TABLE OF CONTENTS

ARTICLE I DEFINITIONS Section 1.1 General ARTICLE II PAYMENTS AND TAX REFUNDS Section 2.1 Allocation of Federal Taxes Section 2.2 Allocation of State Taxes Section 2.3 Allocation of Foreign Taxes	2 2 9 9
Section 1.1 General ARTICLE II PAYMENTS AND TAX REFUNDS Section 2.1 Allocation of Federal Taxes Section 2.2 Allocation of State Taxes Section 2.3 Allocation of Foreign Taxes	9
Section 2.1 Allocation of Federal Taxes Section 2.2 Allocation of State Taxes Section 2.3 Allocation of Foreign Taxes	9
Section 2.1 Allocation of Federal Taxes Section 2.2 Allocation of State Taxes Section 2.3 Allocation of Foreign Taxes	9
Section 2.2 Allocation of State Taxes Section 2.3 Allocation of Foreign Taxes	
Section 2.3 Allocation of Foreign Taxes	10
	10
	11
Section 2.4 Certain Transaction Taxes	12
Section 2.5 Determinations Regarding the Allocation and Attribution of Taxes	12
Section 2.6 Allocation of Employment Taxes	13
Section 2.7 Tax Refunds	13
Section 2.8 Tax Benefits	13
Section 2.9 Prior Agreements	13
ARTICLE III PREPARATION AND FILING OF TAX RETURNS	14
Section 3.1 Remainco's Responsibility	14
Section 3.1 Remainted's Responsibility Section 3.2 Spinco's Responsibility	14
Section 3.2 Spinico's Responsionity Section 3.3 Right To Review Tax Returns	14
Section 3.4 Cooperation	14
Section 3.5 Tax Reporting Practices	14
Section 3.6 Reporting of Reorganization	15
Section 3.7 Payment of Taxes	16
Section 3.8 Amended Returns and Carrybacks	17
Section 3.9 Tax Attributes	17
Section 2.5 Markanounce	1,
ARTICLE IV TAX-FREE STATUS OF THE DISTRIBUTION	18
Section 4.1 Representations and Warranties	18
Section 4.2 Restrictions Relating to the Distribution	19
ARTICLE V INDEMNITY OBLIGATIONS	21
Section 5.1 Indemnity Obligations	21
Section 5.2 Indemnification Payments	21
Section 5.3 Payment Mechanics	22
Section 5.4 Treatment of Payments	22
ARTICLE VI TAX CONTESTS	22
Section 6.1 Notice	22
Section 6.2 Separate Returns	22
Section 6.3 Joint Return	23
Section 6.4 Obligation of Continued Notice	23
Section 6.5 Settlement Rights	23
Section 6.6 Tax Contest Participation	23
ARTICLE VII COOPERATION	24
Section 7.1 General	24
Section 7.2 Consistent Treatment	24
ADTICLE VIII DETENTION OF DECODDS: ACCESS	24
ARTICLE VIII RETENTION OF RECORDS; ACCESS Section 8.1 Retention of Records	24
Section 8.2 Access to Tax Records	25
Section 6.2 Access to 14x records	23
ARTICLE IX DISPUTE RESOLUTION	25
Section 9.1 Dispute Resolution	25

ARTICLE X MISCELLANEOUS PROVISIONS		
Section 10.1 Entire Agreement; Construction	25	
Section 10.2 Interest on Late Payments	26	
Section 10.3 Successors and Assigns	26	
Section 10.4 Subsidiaries	26	
Section 10.5 Assignability	26	
Section 10.6 No Fiduciary Relationship	26	
Section 10.7 Further Assurances	26	
Section 10.8 Survival	26	
Section 10.9 Notices	27	
Section 10.10 Counterparts	27	
Section 10.11 Consents	27	
Section 10.12 Expenses	27	
Section 10.13 Termination and Amendment	27	
Section 10.14 Titles and Headings	28	
Section 10.15 Severability	28	
Section 10.16 Interpretation	28	
Section 10.17 No Duplication; No Double Recovery	28	
Section 10.18 No Waiver	28	
Section 10.19 Governing Law	28	
Section 10.20 Distribution Time	28	

TAX MATTERS AGREEMENT

This TAX MATTERS AGREEMENT (this "Agreement"), is entered into as of March 14, 2023 by and among KINS Technology Group Inc., a Delaware corporation ("Parent"), Inpixon, a Nevada corporation ("Remainco"), and CXApp Holding Corp., a Delaware corporation ("Spinco" and, together with Parent and Remainco, the "Parties"). Capitalized terms used in this Agreement and not otherwise defined herein shall have the meanings ascribed to such terms in the Separation and Distribution Agreement, dated as of the date hereof, by and among the Parties (the "Separation Agreement").

RECITALS

WHEREAS, the board of directors of Remainco has determined that it is in the best interests of Remainco to separate Remainco's business from Spinco's business pursuant to the Separation Agreement (the "Separation") and, following the Separation, to undertake the Distribution;

WHEREAS, Remainco will effect certain restructuring transactions for the purpose of aggregating Spinco's business in the Spinco Group (as defined below) prior to the Distribution (the "*Reorganization*") and in connection therewith, undertake the Contribution to Spinco in exchange for which Spinco shall issue to Remainco shares of Spinco Common Stock;

WHEREAS, Remainco intends to effect the Distribution in a transaction that, together with the Contribution, is intended to qualify as tax-free pursuant to Sections 355 and 368(a)(1)(D) of the Code;

WHEREAS, pursuant to that Merger Agreement entered into as of September 25, 2022, by and among Remainco, Spinco, Parent, and Merger Sub (the "Merger Agreement"), following the completion of the Distribution, Merger Sub will be merged with and into Spinco, with Spinco continuing as the surviving corporation;

WHEREAS, the Parties intend that the Merger (as defined below) will qualify as a "reorganization" within the meaning of Section 368(a) of the Code;

WHEREAS, certain members of the Remainco Group (as defined below), on the one hand, and certain members of the Spinco Group, on the other hand, file certain Tax Returns on a consolidated, combined or unitary basis for certain federal, state, local and foreign Tax purposes; and

WHEREAS, the Parties desire to (a) provide for the payment of Tax liabilities and entitlement to refunds thereof, allocate responsibility for, and cooperation in, the filing of Tax Returns, and provide for certain other matters relating to Taxes and (b) set forth certain covenants and indemnities relating to the preservation of the tax-free status of the Distribution combined with certain steps in the Reorganization.

NOW, THEREFORE, in consideration of the mutual agreements, provisions and covenants contained in this Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

Section 1.1 General. As used in this Agreement, the following terms shall have the following meanings:

- (1) "Adjustment" shall mean an adjustment of any item of income, gain, loss, deduction, credit or any other item affecting Taxes of a taxpayer pursuant to a Final Determination.
- (2) "Affiliate" shall mean, with respect to a Person, any other Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, the specified Person. For this purpose, "control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise. The term "Affiliate" shall refer to Affiliates of a Person as determined immediately after the Merger.
 - (3) "Agreement" shall have the meaning set forth in the preamble hereto.
- (4) "Ancillary Agreements" shall have the meaning set forth in the Separation Agreement; provided, however, this Agreement shall not be considered an "Ancillary Agreement."
 - (5) "Business Day" shall have the meaning set forth in the Separation Agreement.
- (6) "Controlling Party" shall mean, with respect to a Tax Contest, the Party entitled to control such Tax Contest pursuant to Sections 6.2 and 6.3 of this Agreement.
 - (7) "Code" shall mean the Internal Revenue Code of 1986, as amended.
 - (8) "Contribution" shall have the meaning set forth in the Separation Agreement.
 - (9) "Distribution" shall have the meaning set forth in the Separation Agreement.
 - (10) "Distribution Date" shall mean the date on which the Distribution is completed.
 - (11) "Distribution Taxes" means any Taxes incurred solely as a result of the failure of the Tax-Free Status of the Internal Transactions.
 - (12) "Distribution Time" shall have the meaning set forth in the Separation Agreement.
 - (13) "Employee Matters Agreement" shall have the meaning set forth in the Separation Agreement.

- (14) "Employment Tax" shall mean those Liabilities (as defined in the Separation Agreement) for Taxes which are allocable pursuant to the provisions of the Employee Matters Agreement.
- (15) "Equity Awards" means options, share appreciation rights, restricted shares, share units or other compensatory rights with respect to Spinco Common Stock or Parent stock.
- (16) "Federal Income Tax" shall mean any Tax imposed by Subtitle A of the Code other than an Employment Tax, and any interest, penalties, additions to tax, or additional amounts in respect of the foregoing.
- (17) "Federal Other Tax" any Tax imposed by the federal government of the United States other than any Federal Income Tax and any interest, penalties, additions to Tax, or additional amounts in respect of the foregoing.
 - (18) "Federal Tax" means any Federal Income Tax or Federal Other Tax.
- (19) "Final Determination" shall mean the final resolution of liability for any Tax for any Tax Period, by or as a result of (a) a final decision, judgment, decree or other order by any court of competent jurisdiction that can no longer be appealed, (b) a final settlement with the IRS, a closing agreement or accepted offer in compromise under Sections 7121 or 7122 of the Code, or a comparable agreement under the Laws of other jurisdictions, which resolves the entire Tax liability for any Tax Period, (c) any allowance of a refund or credit in respect of an overpayment of Tax, but only after the expiration of all periods during which such refund or credit may be recovered by the jurisdiction imposing the Tax, or (d) any other final resolution, including by reason of the expiration of the applicable statute of limitations or the execution of a pre-filing agreement with the IRS or other Taxing Authority.
- (20) "Foreign Income Tax" shall mean any Tax imposed by any foreign country or any possession of the United States, or by any political subdivision of any foreign country or United States possession, which is an income Tax as defined in Treasury Regulations § 1.901-2, and any interest, penalties, additions to tax, or additional amounts in respect of the foregoing.
- (21) "Foreign Other Tax" shall mean any Tax imposed by any foreign country or any possession of the United States, or by any political subdivision of any foreign country or United States possession, other than any Foreign Income Taxes, and any interest, penalties, additions to tax or additional amounts in respect of the foregoing.
 - (22) "Foreign Tax" shall mean any Foreign Income Taxes or Foreign Other Taxes.
 - (23) "Governmental Entity" shall have the meaning set forth in the Separation Agreement.
 - (24) "Group" shall mean the Remainco Group, the Spinco Group or the Parent Group, as the context requires.

- (25) "Indemnifying Party" shall have the meaning set forth in Section 5.2.
- (26) "Indemnitee" shall have the meaning set forth in Section 5.2.
- (27) "IRS" shall mean the United States Internal Revenue Service or any successor thereto, including, but not limited to its agents, representatives, and attorneys.
- (28) "Joint Return" shall mean any Tax Return that actually includes, by election or otherwise, or is required to include under applicable Law, one or more members of the Remainco Group together with one or more members of the Spinco Group.
 - (29) "Law" shall have the meaning set forth in the Separation Agreement.
 - (30) "Merger" shall have the meaning set forth in the Merger Agreement.
 - (31) "Merger Sub" shall have the meaning set forth in the Merger Agreement.
- (32) "Non-Controlling Party" shall mean, with respect to a Tax Contest, the Party that is not entitled to control such Tax Contest pursuant to Sections 6.2 and 6.3 of this Agreement.
 - (33) "Parent" shall have the meaning set forth in the preamble hereto.
 - (34) "Parent Group" shall mean Parent and each of its direct and indirect Subsidiaries after the Merger.
 - (35) "Parties" shall mean the parties to this Agreement.
 - (36) "Past Practices" shall have the meaning set forth in Section 3.5.
 - (37) "Person" shall have the meaning set forth in the Separation Agreement.
- (38) "Post-Distribution Period" shall mean any Tax Period (or portion thereof) beginning after the Distribution Date, including for the avoidance of doubt, the portion of any Straddle Period beginning after the Distribution Date.
 - (39) "Post-Distribution Ruling" shall have the meaning set forth in Section 4.2(c).
- (40) "Pre-Distribution Period" shall mean any Tax Period (or portion thereof) ending on or before the Distribution Date, including for the avoidance of doubt, the portion of any Straddle Period ending at the end of the day on the Distribution Date.
 - (41) "Prohibited Acts" shall have the meaning set forth in Section 4.2.

- (42) "Proposed Acquisition Transaction" shall mean a transaction or series of transactions (or any agreement, understanding or arrangement, within the meaning of Section 355(e) of the Code and Treasury Regulations § 1.355-7, or any other regulations promulgated thereunder, to enter into a transaction or series of transactions), whether such transaction is supported by Spinco or Parent management or shareholders, is a hostile acquisition, or otherwise, as a result of which Spinco (or any successor thereto) or Parent would merge or consolidate with any other Person or as a result of which one or more Persons would (directly or indirectly) acquire, or have the right to acquire, an amount of stock of Spinco or Parent that would, when combined with any other changes in ownership of Spinco stock or Parent stock pertinent for purposes of Section 355(e) of the Code (including the Merger), comprise 45% or more of (a) the value of all outstanding shares of stock of Spinco or Parent, as applicable, as of the date of such transaction, or in the case of a series of transactions, the date of the last transaction of such series, or (b) the total combined voting power of all outstanding shares of voting stock of Spinco or Parent, as applicable, as of the date of such transaction, or in the case of a series of transactions, the date of the last transaction of such series. Notwithstanding the foregoing, a Proposed Acquisition Transaction shall not include (i) the adoption by Spinco or Parent of a shareholder rights plan, (ii) issuances by Spinco or Parent that satisfy Safe Harbor VIII (relating to acquisitions in connection with a person's performance of services) or Safe Harbor IX (relating to acquisitions by a retirement plan of an employer) of Treasury Regulations § 1.355-7(d), including such issuances net of exercise price and/or tax withholding (provided, however, that any sale of such stock in connection with a net exercise or tax withholding is not exempt under this clause (ii) unless it satisfies the requirements of Safe Harbor VII of Treasury Regulations § 1.355-7(d)) or (iii) acquisitions that satisfy Safe Harbor VII of Treasury Regulations § 1.355-7(d). For purposes of determining whether a transaction constitutes an indirect acquisition, any recapitalization resulting in a shift of voting power or any redemption of shares of stock shall be treated as an indirect acquisition of shares of stock by the non-exchanging shareholders. For purposes of this definition, each reference to Spinco shall include a reference to any entity treated as a successor thereto. This definition and the application thereof is intended to monitor compliance with Section 355(e) of the Code and the Treasury Regulations promulgated thereunder and shall be interpreted accordingly. Any clarification of, or change in, the statute or regulations promulgated under Section 355(e) of the Code shall be incorporated in this definition and its interpretation. For the avoidance of doubt, the Merger shall not constitute a proposed Acquisition Transaction.
 - (43) "Protective Section 336(e) Elections" shall have the meaning set forth in Section 3.6(b).
- (44) "Reasonable Basis" shall mean reasonable basis within the meaning of Section 6662(d)(2)(B)(ii)(II) of the Code and the Treasury Regulations promulgated thereunder (or such other level of confidence required by the Code at that time to avoid the imposition of penalties).
- (45) "Refund" shall mean any refund, reimbursement, offset, credit, or other similar benefit in respect of Taxes (including any overpayment of Taxes that can be refunded or, alternatively, applied against other Taxes payable), including any interest paid on or with respect to such refund of Taxes; provided, however, that the amount of any refund of Taxes shall be net of any Taxes imposed on, related to, or attributable to, the receipt of or accrual of such refund, including any Taxes imposed by way of withholding or offset.
 - (46) "Remainco" shall have the meaning set forth in the preamble hereto.
- (47) "Remainco Affiliated Group" shall mean an affiliated group (as that term is defined in Section 1504 of the Code and the regulations thereunder) of which a member of the Remainco Group is a member.

- (48) "Remainco Common Stock" shall mean the common stock of Remainco, par value \$0.001 per share.
- (49) "Remainco Federal Consolidated Income Tax Return" shall mean any U.S. federal income Tax Return for a Remainco Affiliated Group.
- (50) "Remainco Group" shall mean Remainco and each Person that is a Subsidiary of Remainco; provided, however, that no member of the Spinco Group shall be a member of the Remainco Group.
 - (51) "Remainco Retained Business" shall have the meaning given to the term "Inpixon Retained Business" in the Separation Agreement.
- (52) "Remainco Separate Return" shall mean any Tax Return of or including any member of the Remainco Group (including any consolidated, combined, or unitary return) that does not include any member of the Spinco Group.
 - (53) "Reorganization" shall have the meaning set forth in the recitals.
- (54) "Responsible Party" shall mean, with respect to any Tax Return, the Party having responsibility for preparing and filing such Tax Return pursuant to this Agreement.
 - (55) "Restricted Period" shall mean the period which begins with the Distribution Date and ends two (2) years thereafter.
 - (56) "Section 336(e) Allocation Statement" shall have the meaning set forth in Section 3.6(c).
- (57) "Section 336(e) Tax Benefit Percentage" means, with respect to any Distribution Taxes and Tax-Related Losses attributable thereto, the percentage equal to one hundred percent (100%) minus the percentage of such Distribution Taxes and Tax-Related Losses for which Remainco is entitled to indemnification under this Agreement.
 - (58).
 - (59) "Separate Return" shall mean a Remainco Separate Return or a Spinco Separate Return, as the case may be.
 - (60) "Separation" shall have the meaning set forth in the recitals.
 - (61) "Separation Agreement" shall have the meaning set forth in the preamble hereto.
 - (62) "Spinco" shall have the meaning set forth in the preamble hereto.

- (63) "Spinco Business" shall have the meaning given to the term "Enterprise Apps Business" in the Separation Agreement.
- (64) "Spinco Common Stock" shall mean the Common Stock, par value \$0.00001 per share, of Spinco.
- (65) "Spinco Group" shall mean Spinco and each Person that will be a Subsidiary of Spinco as of immediately after the Distribution Time; provided, that, for the avoidance of doubt, no member of the Remainco Group shall be a member of the Spinco Group.
- (66) "Spinco Separate Return" shall mean any Tax Return of or including any member of the Spinco Group (including any consolidated, combined, or unitary return) that does not include any member of the Remainco Group.
 - (67) "Straddle Period" shall mean any taxable year or other Tax Period that begins on or before the Distribution Date and ends after the Distribution Date.
- (68) "State Income Tax" means any Tax imposed by any State of the United States or by any political subdivision of any such State that is imposed on or measured by income, including state or local franchise or similar Taxes measured by income, as well as any state or local franchise or similar Taxes imposed in lieu of or in addition to a tax imposed on or measured by income and any interest, penalties, additions to tax, or additional amounts in respect of the foregoing.
- (69) "State Other Tax" means any Tax imposed by any state of the United States or by any political subdivision of any such state or the District of Columbia, other than any State Income Tax, and any interest, penalties, additions to tax, or additional amounts in respect of the foregoing
 - (70) "State Taxes" means any State Income Tax or any State Other Tax.
 - (71) "Subsidiary" shall have the meaning set forth in the Separation Agreement.
- (72) "Tax" or "Taxes" shall mean (i) all taxes, charges, fees, duties, levies, imposts, rates or other assessments or governmental charges of any kind imposed by any federal, state, local or non-U.S. Governmental Entity or political subdivision thereof, including, without limitation, income, gross receipts, employment, estimated, excise, severance, stamp, occupation, premium, windfall profits, environmental, custom duties, property, sales, use, license, capital stock, transfer, franchise, registration, payroll, withholding, social security, unemployment, disability, value added, alternative or add-on minimum or other taxes, whether disputed or not, and including any interest, penalties, charges or additions attributable thereto, (ii) liability for the payment of any amount of the type described in clause (i) above arising as a result of being (or having been) a member of any group or being (or having been) included or required to be included in any Tax Return related thereto, and (iii) liability for the payment of any amount of the type described in clauses (i) or (ii) above as a result of any express or implied obligation to indemnify or otherwise assume or succeed to the liability of any other Person.

- (73) "Tax Attribute" shall mean net operating losses, capital losses, research and experimentation credit carryovers, investment tax credit carryovers, earnings and profits, foreign tax credit carryovers, overall foreign losses, overall domestic losses, previously taxed income, separate limitation losses and any other losses, deductions, credits or other comparable items that could affect a Tax liability for a past or future Tax Period.
 - (74) "Tax Benefit" shall have the meaning set forth in Section 2.8.
- (75) "Tax Certificates" shall mean any certificates of officers of Parent, Remainco and Spinco, provided to Skadden, Arps, Slate, Meagher & Flom LLP, Mitchell Silberberg & Knupp LLP or any other law or accounting firm in connection with any Tax Opinion issued in connection with the Reorganization, Distribution, or Merger.
 - (76) "Tax Contest" shall have the meaning set forth in Section 6.1.
- (77) "Tax-Free Status of the Internal Transactions" shall mean the qualification of the Contribution and the Distribution, taken together, (A) as tax free under Sections 355 and 368(a)(1)(D) of the Code, (B) as a transaction in which the stock distributed thereby is "qualified property" for purposes of Sections 355(c) and 361(c) of the Code and (C) as a transaction in which Remainco, Spinco and the holders of Remainco Common Stock recognize no income or gain for U.S. federal income tax purposes pursuant to Sections 355, 361 and 1032 of the Code, and in the case of Remainco Common Stock, for cash in lieu of fractional shares of Spinco Common Stock and in the case of Remainco and Spinco, amounts subject to Section 356 of the Code and intercompany items or excess loss accounts taken into account pursuant to the Treasury Regulations promulgated pursuant to Section 1502 of the Code.
- (78) "Tax-Free Status of the Merger" shall mean the qualification of the Merger as a reorganization under 368(a) of the Code and as a transaction in which the shareholders of Spinco recognize no income or gain pursuant to Section 354(a) of the Code (except to the extent of any cash received in lieu of fractional shares of Parent stock).
 - (79) "Tax-Free Status of the Transactions" shall mean both the Tax-Free Status of the Internal Transactions and the Tax-Free Status of the Merger.
 - (80) "Tax Item" shall mean any item of income, gain, loss, deduction, or credit.
 - (81) "Tax Law" shall mean the law of any Taxing Authority or political subdivision thereof relating to any Tax.
 - (82) "Tax Materials" shall have the meaning set forth in Section 4.1(a).
- (83) "*Tax Opinion*" shall mean any written opinion of Skadden, Arps, Slate, Meagher & Flom LLP, Mitchell Silberberg & Knupp LLP or any other Law or accounting firm, regarding certain tax consequences of certain transactions executed as part of the Separation, the Reorganization, the Contribution, the Distribution or the Merger, as applicable.
 - (84) "Tax Period" means, with respect to any Tax, the period for which such Tax is reported as provided under the Code or other applicable Tax Law.

- (85) "Tax Records" shall have the meaning set forth in Section 8.1.
- (86) "Tax-Related Losses" shall mean with respect to any Taxes, (i) all accounting, legal and other professional fees, and court costs incurred in connection with such Taxes, as well as any other out-of-pocket costs incurred in connection with such Taxes; and (ii) all costs, expenses and damages associated with stockholder litigation or controversies and any amount paid by Remainco (or any of its Affiliates) or Spinco (or any of its Affiliates) in respect of the liability of shareholders, whether paid to shareholders or to the IRS or any other Taxing Authority, in each case, resulting from the failure of the Tax-Free Status of the Transactions.
- (87) "Tax Return" shall mean any return, report, certificate, form or similar statement or document (including any related supporting information or schedule attached thereto and any information return, amended tax return, claim for refund or declaration of estimated tax) supplied to or filed with, or required to be supplied to or filed with, a Taxing Authority, in each case, in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.
- (88) "Taxing Authority" shall mean any Governmental Entity or any subdivision, agency, commission or entity thereof or any quasi-governmental or private body having jurisdiction over the assessment, determination, collection or imposition of any Tax (including the IRS).
 - (89) "Transactions" shall mean the Contribution, Distribution and Merger.
- (90) "Transaction Taxes" shall mean all Taxes imposed on the Remainco Group or the Spinco Group in connection with the Separation, the Reorganization, the Contribution or the Distribution other than Distribution Taxes.
 - (91) "Treasury Regulations" shall mean the regulations promulgated from time to time under the Code as in effect for the relevant Tax Period.
- (92) "Unqualified Tax Opinion" shall mean a "will" opinion, without substantive qualifications, of a nationally recognized law firm or accounting firm, to the effect that a transaction will not affect the Tax-Free Status of the Transactions. Any such opinion may assume that the Tax-Free Status of the Transactions would apply if not for the occurrence of the transaction in question.

ARTICLE II PAYMENTS AND TAX REFUNDS

Section 2.1 Allocation of Federal Taxes. Except as otherwise provided in Section 2.4, Federal Taxes shall be allocated as follows:

- (a) Federal Income Taxes.
- (i) Remainco shall be responsible for any and all Federal Income Taxes (including any increase in such Taxes as a result of a Final Determination) due with respect to or required to be reported on (A) any Joint Return; provided, however, that Parent and Spinco shall be responsible for any and all such Taxes that are attributable to the Spinco Business with respect to any Post-Distribution Period, (B) any Remainco Separate Return, or (C) any Spinco Separate Return with respect to any Pre-Distribution Period.

- (ii) Parent and Spinco shall be responsible for any and all Federal Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Spinco Separate Return with respect to any Post-Distribution Period.
- (b) Federal Other Taxes Relating to Joint Returns. Remainco shall be responsible for any and all Federal Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Joint Return; provided, however, that Parent and Spinco shall be responsible for any and all such Taxes that are attributable to the Spinco Business with respect to any Post-Distribution Period.
 - (c) Federal Other Taxes Relating to Separate Returns.
 - (i) Remainco shall be responsible for any and all Federal Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on (A) any Remainco Separate Return or (B) any Spinco Separate Return with respect to any Pre-Distribution Period.
 - (ii) Parent and Spinco shall be responsible for any and all Federal Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Spinco Separate Return with respect to any Post-Distribution Period.
 - Section 2.2 Allocation of State Taxes. Except as otherwise provided in Section 2.4, State Taxes shall be allocated as follows.
- (a) State Income Taxes Relating to Joint Returns. Remainco shall be responsible for any and all State Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Joint Return; provided, however, that Parent and Spinco shall be responsible for any and all such Taxes that are attributable to the Spinco Business with respect to any Post-Distribution Period.
 - (b) State Income Taxes Relating to Separate Returns.
 - (i) Remainco shall be responsible for any and all State Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on (A) any Remainco Separate Return or (B) any Spinco Separate Return with respect to any Pre-Distribution Period.
 - (ii) Parent and Spinco shall be responsible for any and all State Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Spinco Separate Return with respect to any Post-Distribution Period.

- (c) State Other Taxes Relating to Joint Returns. Remainco shall be responsible for any and all State Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Joint Return; provided, however, that Parent and Spinco shall be responsible for any and all such Taxes that are attributable to the Spinco Business with respect to any Post-Distribution Period.
 - (d) State Other Taxes Relating to Separate Returns.
 - (i) Remainco shall be responsible for any and all State Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on (A) any Remainco Separate Return or (B) any Spinco Separate Return with respect to any Pre-Distribution Period.
 - (ii) Parent and Spinco shall be responsible for any and all State Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Spinco Separate Return for any Post-Distribution Period.
 - Section 2.3 Allocation of Foreign Taxes. Except as otherwise provided in Section 2.4, Foreign Taxes shall be allocated as follows:
- (a) Foreign Income Taxes Relating to Joint Returns. Remainco shall be responsible for any and all Foreign Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Joint Return; provided, however, that Parent and Spinco shall be responsible for any and all such Taxes that are attributable to the Spinco Business with respect to any Post-Distribution Period.
 - (b) Foreign Income Taxes Relating to Separate Returns.
 - (i) Remainco shall be responsible for any and all Foreign Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on (A) any Remainco Separate Return or (B) any Spinco Separate Return with respect to any Pre-Distribution Period.
 - (ii) Parent and Spinco shall be responsible for any and all Foreign Income Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Spinco Separate Return with respect to any Post-Distribution Period.
- (c) Foreign Other Taxes Relating to Joint Returns. Remainco shall be responsible for any and all Foreign Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Joint Return; provided, however, that Parent and Spinco shall be responsible for any and all such Taxes that are attributable to the Spinco Business with respect to any Post-Distribution Period.

- (d) Foreign Other Taxes Relating to Separate Returns.
- (i) Remainco shall be responsible for any and all Foreign Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on (A) any Remainco Separate Return or (B) any Spinco Separate Return with respect to any Pre-Distribution Period.
- (ii) Parent and Spinco shall be responsible for any and all Foreign Other Taxes (including any increase in such Taxes as a result of a Final Determination) required to be reported on any Spinco Separate Return with respect to any Post-Distribution Period.
- Section 2.4 Transaction Taxes and Distribution Taxes. Notwithstanding the provisions set forth in Sections 2.1, 2.2, and 2.3:
- (a) Parent and Spinco shall pay and be responsible for any Transaction Taxes in excess of the Transaction Taxes that would have been imposed on the Separation, the Reorganization, the Contribution, or the Distribution had such transactions been consummated but the Merger was not consummated; and
- (b) Remainco shall pay and be responsible for any and all Transaction Taxes other than those Transaction Taxes described in Section 2.4(a), and any and all Distribution Taxes (including any Taxes imposed on Remainco pursuant to Section 355(e) of the Code).
- Section 2.5 Determinations Regarding the Allocation and Attribution of Taxes. For purposes of Sections 2.1, 2.2, and 2.3, Taxes shall be allocated, to the extent relevant, in accordance with the following:
 - (a) With respect to the Remainco Federal Consolidated Income Tax Return for the taxable year that includes the Distribution Date, Remainco shall use the closing of the books method under Treasury Regulations § 1.1502-76, unless otherwise agreed by Remainco and Parent.
 - (b) Remainco, Parent, and Spinco shall take all actions necessary or appropriate to close the taxable year of each member of the Spinco Group for all Tax purposes as of the close of the Distribution Date to the extent permitted by applicable Law. With respect to Taxes for any Straddle Period, (a) if applicable Law does not permit a member of the Spinco Group to close its taxable year on the Distribution Date, then the allocation of income or deductions required to determine any Taxes or other amounts attributable to the portion of the Straddle Period ending on, or beginning after, the Distribution Date shall be made by means of a closing of the books and records of such member as of the close of the Distribution Date; provided that exemptions, allowances, or deductions that are calculated on an annual or periodic basis shall be allocated between such portions in proportion to the number of days in each such portion, and (b) any other Taxes, including property Taxes, that are calculated on an annual or periodic basis and not assessed with respect to a transaction or series of transactions shall be allocated to the portion of the Straddle Period ending on the Distribution Date and the portion of the Straddle Period beginning after the Distribution Date in proportion to the number of days in each such portion.

Section 2.6 Allocation of Employment Taxes. Liability for Employment Taxes shall be determined pursuant to the Employee Matters Agreement.

Section 2.7 Tax Refunds.

- (a) Remainco shall be entitled to all Refunds related to Taxes the liability for which is allocated to Remainco pursuant to this Agreement. Spinco shall be entitled to all Refunds related to Taxes the liability for which it is allocated to Spinco pursuant to this Agreement.
- (b) Parent or Spinco shall pay to Remainco any Refund received by Parent or Spinco or any member of the Spinco Group or Parent Group that is allocable to Remainco pursuant to this Section 2.7 no later than thirty (30) Business Days after the receipt of such Refund. Remainco shall pay to Spinco any Refund received by Remainco or any member of the Remainco Group that is allocable to Spinco pursuant to this Section 2.7 no later than thirty (30) Business Days after the receipt of such Refund. For purposes of this Section 2.7, any Refund that arises as a result of an offset, credit, or other similar benefit in respect of Taxes other than a receipt of cash shall be deemed to be received on the earlier of (i) the date on which a Tax Return is filed claiming such offset, credit, or other similar benefit and (ii) the date on which payment of the Tax which would have otherwise been paid absent such offset, credit, or other similar benefit is due (determined without taking into account any applicable extensions). To the extent that the amount of any Refund in respect of which a payment was made under this Section 2.7 is later reduced by a Taxing Authority or in a Tax Contest, such reduction shall be allocated to the Party to which such Refund was allocated pursuant to this Section 2.7 and an appropriate adjusting payment shall be made.

Section 2.8 Tax Benefits. Except with respect to any Tax Benefit arising as a result of the Protective 336(e) Elections, if (a) one Party is responsible for a Tax pursuant to this Agreement and (b) the other Party is entitled to a deduction, credit or other Tax benefit (a "Tax Benefit") relating to such Tax, then the Party entitled to such Tax Benefit shall pay to the Party responsible for such Tax the amount of any cash Tax savings realized by the entitled Party as a result of such Tax Benefit, net of any Taxes imposed by any Taxing Authority on, related to, or attributable to, the receipt of or accrual of such Tax Benefit, including any Taxes imposed by way of withholding or offset, no later than thirty (30) Business Days after such cash Tax savings are realized. To the extent that the amount of any Tax Benefit in respect of which a payment was made under this Section 2.8 is later reduced by a Taxing Authority or in a Tax Contest, the Party that received such payment shall refund such payment to the Party that made such payment to the extent of such reduction.

Section 2.9 Prior Agreements. Except as set forth in this Agreement and in consideration of the mutual indemnities and other obligations of this Agreement, any and all prior Tax sharing or allocation agreements or practices between any member of the Remainco Group and any member of the Spinco Group shall be terminated with respect to the Spinco Group and the Remainco Group as of the Distribution Date. No member of either the Spinco Group or the Remainco Group shall have any continuing rights or obligations under any such agreement.

ARTICLE III PREPARATION AND FILING OF TAX RETURNS

Section 3.1 Remainco's Responsibility. Remainco shall prepare and file, or shall cause to be prepared and filed, when due (taking into account any applicable extensions) all Joint Returns, and all Remainco Separate Returns, including any such amended Joint Returns or Separate Returns. In addition, Remainco shall prepare and file all Spinco Separate Returns for any Tax Period (or portion thereof) ending on or before the Distribution Date.

Section 3.2 Spinco's Responsibility. Parent or Spinco shall prepare and file, or shall cause to be prepared and filed, when due (taking into account any applicable extensions) all Spinco Separate Returns, including any such amended Spinco Separate Returns, for any Straddle Period or for tax years beginning after the Distribution Date.

Section 3.3 Right To Review Tax Returns. To the extent that a Party (the "Reviewing Party") would reasonably be expected to be adversely affected by the positions taken on any Tax Return or could reasonably be required by the terms of this Agreement to provide an indemnity or make a payment for any Taxes reported or required to be reported on any Tax Return is not the Responsible Party, the Responsible Party shall prepare the portions of such Tax Return that could affect or result in indemnification by the Reviewing Party, shall provide a draft of such portions of such Tax Return to the Reviewing Party for its review and comment at least thirty (30) days prior to the due date for such Tax Return, and shall modify such portions of such Tax Return before filing to include the Reviewing Party's reasonable comments.

Section 3.4 Cooperation. The Parties shall provide, and shall cause their Affiliates to provide, assistance and cooperation to one another in accordance with Article VII with respect to the preparation and filing of Tax Returns, including providing information required to be provided under Article VIII. Notwithstanding anything to the contrary in this Agreement, Remainco shall not be required to disclose to Parent or Spinco any consolidated, combined, unitary, or other similar Joint Return of which a member of the Remainco Group is the common parent or any information related to such a Joint Return other than information relating solely to the Spinco Group; provided, that Remainco shall provide such additional information that is reasonably required in order for Spinco to determine the Taxes attributable to the Spinco Business. If an amended Separate Return for which Parent or Spinco is responsible under this Article III is required to be filed as a result of an amendment made to a Joint Return pursuant to an audit adjustment, then the Parties shall use their respective commercially reasonable efforts to ensure that such amended Separate Return can be prepared and filed in a manner that preserves confidential information including through the use of confidentiality agreements or third party preparers.

Section 3.5 Tax Reporting Practices. Except as provided in Section 3.6, with respect to any Tax Return for any Tax Period that begins on or before the second anniversary of the Distribution Date with respect to which Parent or Spinco is the Responsible Party, such Tax Return shall be prepared in a manner (i) consistent with past practices, accounting methods, elections and conventions ("Past Practices") used with respect to the Tax Returns in question (unless there is no Reasonable Basis for the use of such Past Practices), and to the extent any items are not covered by Past Practices (or in the event that there is no Reasonable Basis for the use of such Past Practices), in accordance with reasonable Tax accounting practices selected by Spinco; and (ii) that, to the extent consistent with clause (i), minimizes the overall amount of Taxes due and payable on such Tax Return for all of the Parties by cooperating in making such elections or applications for group or other relief or allowances available in the taxing jurisdiction in which such Tax Return is filed; provided, that making such election or application for group or other relief or allowances available in the taxing jurisdiction in which such Tax Return is filed does not have a disproportionate and adverse effect on the Party filing such return. Neither Parent nor Spinco shall take any action inconsistent with the assumptions (including with respect to any Tax Item) made in determining all estimated or advance payments of Taxes on or prior to the Distribution Date. In addition, neither Parent nor Spinco shall be permitted, and shall not permit any member of the Spinco Group or Parent Group, without Remainco's prior written consent (not to be unreasonably withheld, conditioned or delayed), to make a change in any of its methods of accounting for Tax purposes until all applicable statutes of limitations for all Pre-Distribution Periods have expired.

Section 3.6 Reporting of Reorganization.

- (a) The Tax treatment of any step in or portion of the Separation, the Reorganization, the Contribution and the Distribution shall be reported on each applicable Tax Return consistently with the Tax-Free Status of the Transactions, taking into account the jurisdiction in which such Tax Returns are filed, unless there is no Reasonable Basis for such Tax treatment. In the event that a Party shall determine that there is no Reasonable Basis for such Tax treatment, such Party shall notify the other Party no later than twenty (20) Business Days prior to filing the relevant Tax Return and the Parties shall attempt in good faith to agree on the manner in which the relevant portion of the Separation, the Reorganization, the Contribution or the Distribution (as applicable) shall be reported. Notwithstanding the above, the Parties acknowledge that Remainco may engage in certain transactions with third parties after the Distribution Date that may trigger corporate level taxes for Remainco under Section 355(e) of the Code, in which case the Parties shall not be obligated to report the Contribution and Distribution as tax-free to Remainco under Section 355(e).
- (b) After the date hereof, the Parties shall negotiate and cooperate in good faith to determine whether protective elections under Section 336(e) of the Code (and any applicable state or local Tax Law) shall be made with respect to the Distribution for Spinco and each member of the Spinco Group that is a domestic corporation for Federal Income Tax purposes (the "*Protective Section 336(e) Elections*"). Such cooperation shall include the provision by Remainco of any information reasonably necessary to make such determination. If the Parties mutually determine that a Section 336(e) Election would be beneficial, then Remainco and Spinco shall enter into a written, binding agreement to make the Protective Section 336(e) Elections, and Remainco and Spinco shall timely make the Protective Section 336(e) Elections in accordance with Treasury Regulations § 1.336-2(h). For the avoidance of doubt, such agreement is intended to constitute a "written, binding agreement" to make the Protective Section 336(e) Elections within the meaning of Treasury Regulations § 1.336-2(h)(1)(i).
- (c) Remainco, Parent and Spinco shall cooperate in making any agreed-to Protective Section 336(e) Elections, including filing any statements, amending any Tax Returns or undertaking such other actions reasonably necessary to carry out the Protective Section 336(e) Elections. Remainco shall determine the "aggregate deemed asset disposition price" and the "adjusted grossed-up basis" (each as defined under applicable Treasury Regulations) and the allocation of such aggregate deemed asset disposition price and adjusted grossed-up basis among the assets of the applicable member or members of the Remainco Group or Spinco Group, each in accordance with the applicable provisions of Section 336(e) of the Code and applicable Treasury Regulations (the "Section 336(e) Allocation Statement"). Each Party agrees not to take any position (and to cause each of its Affiliates not to take any position) that is inconsistent with the Protective Section 336(e) Elections, including the Section 336(e) Allocation Statement, on any Tax Return, in connection with any Tax Contest or for any other Tax purposes (in each case, excluding any position taken for financial accounting purposes), except as may be required by a Final Determination.

(d) If any Protective Section 336(e) Elections are made, then, in the event of a failure of the Tax-Free Status of the Transactions, the actual Tax savings if, as and when realized by the Spinco Group arising from the increase in Tax basis (including, for the avoidance of doubt, any such increase in Tax basis attributable to payments made pursuant to this Section 3.6(d)) resulting from the Protective Section 336(e) Election, determined on a "with and without" basis (treating any deductions or amortization attributable to the increase in Tax basis resulting from the Protective 336(e) Election, or any other recovery of such increase in Tax basis, as the last items claimed for any taxable year, including after the utilization of any available net operating loss carryforwards), shall be shared between Remainco and Spinco in the same proportion as the Taxes imposed on the Transactions giving rise to such increased basis were borne by Remainco and Spinco (after giving effect to the indemnification obligations in this Agreement); provided, however, that payments made by Spinco pursuant to this Section 3.6(d): (i) shall be reduced by all reasonable costs incurred by any member of the Spinco Group to amend any Tax Returns or other governmental filings related to such Protective Section 336(e) Election and (ii) shall not exceed the amount of any Distribution Taxes and Tax-Related Losses attributable thereto of the Remainco Group (not taking into account this Section 3.6(d)) arising from such failure of the Tax-Free Status of the Transactions and for which Remainco is not entitled to indemnification under this Agreement.

Section 3.7 Payment of Taxes.

- (a) With respect to any Tax Return required to be filed pursuant to this Agreement, the Responsible Party shall remit or cause to be remitted to the applicable Taxing Authority in a timely manner any Taxes due in respect of any such Tax Return.
- (b) In the case of any Tax Return for which the Party that is not the Responsible Party is obligated pursuant to this Agreement to pay all or a portion of the Taxes reported as due on such Tax Return, the Responsible Party shall notify the other Party, in writing, of its obligation to pay such Taxes and, in reasonably sufficient detail, its calculation of the amount due by such other Party and the Party receiving such notice shall pay such amount to the Responsible Party upon the later of five (5) Business Days prior to the date on which such payment is due and thirty (30) Business Days after the receipt of such notice.
- (c) For the avoidance of doubt, with respect to any Taxes that are estimated Taxes, (i) the Party that is or will be the Responsible Party with respect to any Tax Return that will reflect (or otherwise give credit for) such estimated Taxes shall remit or cause to be remitted to the applicable Taxing Authority in a timely manner any estimated Taxes due, and (ii) in the case of any estimated Taxes for which the Party that is not the Responsible Party is obligated pursuant to this Agreement to pay all or a portion of the Taxes that will be reported as due on any Tax Return that will reflect (or otherwise give credit for) such estimated Taxes, the Responsible Party shall notify the other Party, in writing, of its obligation to pay such estimated Taxes and, in reasonably sufficient detail, its calculation of the amount due by such other Party and the Party receiving such notice shall pay such amount to the Responsible Party upon the later of five (5) Business Days prior to the date on which such payment is due and thirty (30) Business Days after the receipt of such notice.

Section 3.8 Amended Returns and Carrybacks.

- (a) Parent and Spinco shall not, and shall not permit any member of the Spinco Group to, file or allow to be filed any request for an Adjustment for any Pre-Distribution Period without the prior written consent of Remainco, such consent not to be unreasonably withheld, conditioned or delayed.
- (b) Except as required by applicable Law, Remainco shall not, and shall not permit any member of the Remainco Group to, file or allow to be filed any amended Tax Return or request for an Adjustment for any Pre-Distribution Period or Straddle Period if the result would be to materially increase any liability of Spinco or any member of the Spinco Group either (i) under this Agreement or (ii) for a Post-Distribution Period, in each case without the prior written consent of Spinco, such consent not to be unreasonably withheld, conditioned or delayed.
- (c) Except as prohibited by applicable Law, Parent and Spinco shall, and shall cause each member of the Spinco Group to, make any available elections to waive the right to carry back any Tax Attribute from a Post-Distribution Period to a Pre-Distribution Period.
- (d) Parent and Spinco shall not, and shall cause each member of the Spinco Group not to, without the prior written consent of Remainco, make any affirmative election to carry back any Tax Attribute from a Post-Distribution Period to a Pre-Distribution Period, such consent to be exercised in Remainco's sole discretion.
- (e) Receipt of consent by Parent, Spinco, or a member of the Spinco Group from Remainco pursuant to the provisions of this Section 3.8 shall not limit or modify Parent's or Spinco's continuing indemnification obligation pursuant to Article V.

Section 3.9 Tax Attributes. Remainco shall in good faith advise Spinco in writing of the amount, if any, of any Tax Attributes, which Remainco reasonably determines in good faith are allocable or apportionable to the Spinco Group under applicable Law. Parent, Spinco and all members of the Parent Group shall prepare all Tax Returns in accordance with such written notice. For the avoidance of doubt, Remainco may elect in its reasonable discretion, in order to comply with this Section 3.9, to create or cause to be created books and records or reports or other documents based thereon (including, without limitation, "earnings & profits studies," "basis studies" or similar determinations) that it does not typically maintain or prepare in the ordinary course of business.

ARTICLE IV TAX-FREE STATUS OF THE DISTRIBUTION

Section 4.1 Representations and Warranties.

- (a) Remainco, on behalf of itself and all other members of the Remainco Group, hereby represents and warrants that (i) it has examined any and all Tax Opinions all materials delivered or deliverable in connection with the Tax Certificates or the rendering of any Tax Opinions (collectively, the "Tax Materials"), (ii) the facts presented and representations that have been or will be made therein, to the extent descriptive of or otherwise relating to Remainco or any member of the Remainco Group (including any members of the Spinco Group prior to the Distribution Time) or the Remainco Retained Business, were or will be at the time presented or represented and from such time until and including the Distribution Time true, correct and complete in all material respects, and (iii) it has delivered copies of the Tax Materials to Parent.
- (b) Remainco, on behalf of itself and all other members of the Remainco Group, hereby confirms and agrees to comply with any and all covenants and agreements in the Tax Materials applicable to Remainco or any member of the Remainco Group or the Remainco Retained Business.
- (c) Spinco, on behalf of itself and all other members of the Spinco Group, hereby represents and warrants or covenants and agrees, as appropriate, that it has examined the Tax Materials and the facts presented and representations that have been or will be made therein, to the extent descriptive of or otherwise relating to (i) the Spinco Group or Parent Group (including the business purposes for the Distribution) after the Distribution Time and the plans, proposals, intentions and policies of the Spinco Group or Parent Group after the Distribution Time, and (ii) the actions or non-actions of the Spinco Group or Parent Group to be taken (or not taken, as the case may be) after the Distribution Time, to its knowledge are, or will be from the time presented or made through and including the Distribution Time (and thereafter as relevant) true, correct and complete in all material respects, provided that, for the avoidance of doubt, notwithstanding anything to the contrary in this Agreement, Remainco rather than Spinco or Parent shall be responsible for the accuracy of, or compliance with, any such representation, warranty, statement, or covenant with respect to the Spinco Group or the Spinco Business at the time presented or made (and, if applicable, through and including the Distribution Time).
- (d) Parent and Spinco, on behalf of themselves and all other members of their respective Groups, hereby confirm and agree to comply with any and all covenants and agreements in the Tax Materials applicable to Parent, Spinco or any member of their respective Groups or the Spinco Business pertaining to the conduct of such entities following the Distribution Time.
- (e) Each of Remainco, on behalf of itself and all other members of the Remainco Group, Spinco, on behalf of itself and all other members of the Spinco Group, and Parent, represents and warrants that it knows of no fact that may cause the failure of the Tax-Free Status of the Transactions other than those described herein.
- (f) Each of Remainco, on behalf of itself and all other members of the Remainco Group, Spinco, on behalf of itself and all other members of the Spinco Group, and Parent represents and warrants that it has no plan or intent to take any action which is inconsistent with any statements or representations made in the Tax Materials.

Section 4.2 Restrictions Relating to the Distribution.

- (a) Remainco, on behalf of itself and all other members of the Remainco Group, hereby covenants and agrees that no member of the Remainco Group will take, fail to take, or to permit to be taken: (i) any action where such action or failure to act would be inconsistent with or cause to be untrue any statement, information, covenant or representation in the Tax Materials, or (ii) any action where such action or failure to act would adversely affect, or could reasonably be expected to adversely affect, the Tax-Free Status of the Transactions. Notwithstanding the above, the Parties acknowledge that Remainco may engage in certain transactions with third parties after the Distribution Date that may trigger corporate level taxes for Remainco under Section 355(e) of the Code. Nothing in this Agreement shall be construed as preventing Remainco from engaging in these third party transactions or of being in violation of this Agreement if it engages in such transactions and properly reports and timely pays the taxes owed under Section 355(e) of the Code.
- (b) Each of Spinco and Parent, on behalf of itself and all other members of their respective Groups, hereby covenants and agrees that no member of their Group will take, fail to take, or permit to be taken any action where such action or failure to act would be inconsistent with or cause to be untrue any statement, information, covenant or representation in the Tax Materials pertaining to Spinco following the Distribution Time.
 - (c) During the Restricted Period, Parent and Spinco:
 - (i) shall continue and cause to be continued the active conduct of the Spinco Business for purposes of Section 355(b)(2) of the Code, taking into account Section 355(b)(3) of the Code, as conducted immediately prior to the Distribution,
 - (ii) shall not voluntarily dissolve or liquidate themselves or any member of the Spinco Group (including any action that is a liquidation for U.S. federal income tax purposes),
 - (iii) shall not (1) enter into any Proposed Acquisition Transaction or, to the extent Spinco or Parent has the right to prohibit any Proposed Acquisition Transaction, permit any Proposed Acquisition Transaction to occur (whether by (A) redeeming rights under a shareholder rights plan, (B) finding a tender offer to be a "permitted offer" under any such plan or otherwise causing any such plan to be inapplicable or neutralized with respect to any Proposed Acquisition Transaction, (C) approving any Proposed Acquisition Transaction, whether for purposes of Section 203 of the General Corporation Law of the State of Delaware or any similar corporate statute, any "fair price" or other provision of the charter or bylaws of Parent or Spinco, (D) amending its certificate of incorporation to declassify its board of directors or approving any such amendment, or (E) otherwise), (2) redeem or otherwise repurchase (directly or through an Affiliate) any stock, or rights to acquire stock except (A) to the extent such repurchases satisfy Section 4.05(1)(b) of Revenue Procedure 96-30 (as in effect prior to the amendment of such Revenue Procedure by Revenue Procedure 2003-48), (B) to the extent reasonably necessary to pay the total tax liability arising from the vesting of an Equity Award, or (C) through a net exercise of an Equity Award, (3) amend its certificate of incorporation (or other organizational documents), or take any other action, whether through a stockholder vote or otherwise, affecting the relative voting rights of its capital stock (including through the conversion of any capital stock into another class of capital stock), (4) merge or consolidate, or agree to merge or consolidate, Parent or Spinco with any other Person (other than pursuant to the Merger) unless, in the case of a merger or consolidation, Parent or Spinco (as applicable) is the survivor of such merger or consolidation or (5) take any other action or actions (including any action or transaction that is inconsistent with any representation made in the Tax Materials regarding Spinco following the Distribution Time) which in the aggregate (and taking into account the Merger) would, when combined with any other direct or indirect changes in ownership of Parent or Spinco capital stock pertinent for purposes of Section 355(e) of the Code, have the effect of causing or permitting one or more Persons (whether or not acting in concert) to acquire directly or indirectly stock representing a fifty percent (50%) or greater interest in Parent or Spinco or would reasonably be expected to result in a failure to preserve the Tax-Free Status of the Transactions; and

- (iv) shall not and shall not permit any member of the Spinco Group, to sell, transfer, or otherwise dispose of or agree to, sell, transfer or otherwise dispose (including in any transaction treated for U.S. federal income tax purposes as a sale, transfer or disposition) of assets (including, any shares of capital stock of a Subsidiary) that, in the aggregate, constitute more than thirty percent (30%) of the consolidated gross assets of Spinco or the Spinco Group; provided, that this clause (iv) shall not apply to (1) sales, transfers, or dispositions of assets in the ordinary course of business, (2) any cash paid to acquire assets from an unrelated Person in an arm's-length transaction, (3) any assets transferred to a Person that is disregarded as an entity separate from the transferor for U.S. federal income tax purposes or (4) any mandatory or optional repayment (or pre-payment) of any indebtedness of Spinco or any member of the Spinco Group; provided, further that the percentages of gross assets or consolidated gross assets of Spinco or the Spinco Group, as the case may be, sold, transferred, or otherwise disposed of, shall be based on the fair market value of the gross assets of Spinco and the members of the Spinco Group as of the Distribution Date. For purposes of this Section 4.2(c)(iv), a merger of Spinco or one of its Subsidiaries with and into any Person that is not a wholly owned Subsidiary of Spinco shall constitute a disposition of all of the assets of Spinco or such Subsidiary.
- (d) Notwithstanding the restrictions imposed by Sections 4.2(b) and 4.2(c), Parent, Spinco or a member of the Spinco Group may take any of the actions or transactions described therein if Spinco either (i) obtains an Unqualified Tax Opinion in form and substance reasonably satisfactory to Remainco, (ii) obtains a ruling from the IRS to the effect that such actions or transactions will not affect the Tax-Free Status of the Transactions (a "Post-Distribution Ruling") or (iii) obtains the prior written consent of Remainco waiving the requirement that Spinco obtain an Unqualified Tax Opinion or Post-Distribution Ruling, such waiver to be provided in Remainco's sole and absolute discretion. Remainco shall cooperate in good faith with any reasonable requests of Spinco in connection with securing any Post-Distribution Ruling or Unqualified Tax Opinion. Remainco's evaluation of an Unqualified Tax Opinion may consider, among other factors, the appropriateness of any underlying assumptions, representations, and covenants made in connection with such opinion. Spinco shall bear all costs and expenses of securing any such Unqualified Tax Opinion or Post-Distribution Ruling and shall reimburse Remainco for all reasonable out-of-pocket expenses that Remainco or any of its Affiliates may incur in good faith in connection with obtaining or evaluating any such Unqualified Tax Opinion or Post-Distribution Ruling. Except as otherwise provided in Section 5.1(d), neither the delivery of an Unqualified Tax Opinion, receipt of a Post-Distribution Ruling nor Remainco's waiver of Spinco's obligation to deliver an Unqualified Tax Opinion or obtain a Post-Distribution Ruling shall limit or modify Parent's or Spinco's continuing indemnification obligation Pursuant to Article V.

ARTICLE V INDEMNITY OBLIGATIONS

Section 5.1 Indemnity Obligations.

- (a) Remainco shall indemnify and hold harmless Parent and Spinco from and against, and will reimburse Spinco for, (i) all liability for Taxes allocated to Remainco pursuant to Article II, (ii) all Taxes and Tax-Related Losses attributable thereto arising out of, based upon, or relating or attributable to any breach of or inaccuracy in, or failure to perform, as applicable, any representation, covenant, or obligation of any member of the Remainco Group pursuant to this Agreement, and (iii) the amount of any Refund received by any member of the Remainco Group that is allocated to Spinco pursuant to Section 2.7(a).
- (b) Except as otherwise provided in Section 5.1(a), without regard to whether an Unqualified Tax Opinion may have been provided, any Post-Distribution Ruling obtained or whether any action is permitted or consented to hereunder and notwithstanding anything else to the contrary contained herein, in the Separation Agreement, the Merger Agreement or other Ancillary Agreement, Parent and Spinco shall indemnify and hold harmless Remainco from and against, and will reimburse Remainco for, (i) all liability for Taxes allocated to Spinco pursuant to Article II, (ii) all Taxes and Tax-Related Losses attributable thereto arising out of, based upon, or relating or attributable to any breach of or inaccuracy in, or failure to perform, as applicable, any representation, covenant, or obligation of any member of the Spinco Group pursuant to this Agreement, and (iii) the amount of any Refund received by any member of the Spinco Group that is allocated to Remainco pursuant to Section 2.7(a).

Section 5.2 Indemnification Payments.

- (a) Except as otherwise provided in this Agreement, if either Party (the "Indemnitee") is required to pay to a Taxing Authority a Tax or to another Person a payment in respect of a Tax that the other Party (the "Indemnifying Party") is liable for under this Agreement, including as the result of a Final Determination, the Indemnites shall notify the Indemnifying Party, in writing, of its obligation to pay such Tax and, in reasonably sufficient detail, its calculation of the amount due by such Indemnifying Party to the Indemnitee, including any Tax-Related Losses attributable thereto. The Indemnifying Party shall pay such amount, including any Tax-Related Losses attributable thereto, to the Indemnitee no later than the later of (i) five (5) Business Days prior to the date on which such payment is due to the applicable Taxing Authority or (ii) thirty (30) Business Days after the receipt of notice from the other Party. Any Tax indemnity payment required to be made pursuant to this Agreement shall be reduced by any corresponding Tax Benefit payment required to be made to the Indemnifying Party by the Indemnitee pursuant to Section 2.8. For the avoidance of doubt, a Tax Benefit payment is treated as corresponding to a Tax indemnity payment to the extent the Tax Benefit realized is directly attributable to the same Tax item (or adjustment of such Tax item pursuant to a Final Determination) that gave rise to the Tax indemnity payment.
- (b) If, as a result of any change or redetermination, any amount previously allocated to and borne by one Party pursuant to the provisions of Article II is thereafter allocated to the other Party, then, no later than thirty (30) Business Days after such change or redetermination, such other Party shall pay to such Party the amount previously borne by such Party which is allocated to such other Party as a result of such change or redetermination.
- (c) If an Indemnitee receives a Refund with respect to a Tax Contest for which the Indemnifying Party made an indemnity payment to the Indemnitee pursuant to Section 5.2(a), the Indemnitee shall pay the amount of such Refund to the Indemnifying Party, such payment to the Indemnifying Party not to exceed such indemnity payment, no later than thirty (30) Business Days after the receipt of such Refund.

Section 5.3 Payment Mechanics.

- (a) All payments under this Agreement shall be made by Remainco directly to Spinco and by Spinco directly to Remainco; provided, however, that if the Parties mutually agree with respect to any such indemnification payment, any member of the Remainco Group, on the one hand, may make such indemnification payment to any member of the Spinco Group, on the other hand, and vice versa. All indemnification payments shall be treated in the manner described in Section 5.4.
- (b) In the case of any payment of Taxes made by a Responsible Party or Indemnitee pursuant to this Agreement for which such Responsible Party or Indemnitee, as the case may be, has received a payment from the other Party, such Responsible Party or Indemnitee shall provide to the other Party a copy of any official government receipt received with respect to the payment of such Taxes to the applicable Taxing Authority (or, if no such official governmental receipts are available, executed bank payment forms or other reasonable evidence of payment).

Section 5.4 Treatment of Payments. The Parties agree that any payment made among the Parties pursuant to this Agreement (other than any payment of interest accruing after the Distribution Date) shall be treated, to the extent permitted by Law, for all U.S. federal income tax purposes as either (i) a non-taxable contribution by Remainco to Spinco or (ii) a distribution by Spinco to Remainco, and, with respect to any payment made among the Parties pursuant to this Agreement after the Distribution, such payment shall be treated as having been made immediately prior to the Distribution.

ARTICLE VI TAX CONTESTS

Section 6.1 Notice. Each Party shall notify the other Party in writing within ten (10) Business Days after receipt by such Party or any member of its Group of a written communication from any Taxing Authority with respect to any pending or threatened audit, claim, dispute, suit, action, proposed assessment or other proceeding (a "Tax Contest") concerning any Taxes for which the other Party may be liable pursuant to this Agreement, and thereafter shall promptly forward or make available to such Party copies of notices and communications relating to such Tax Contest.

Section 6.2 Separate Returns.

- (a) If, pursuant to Article II hereof, Spinco has sole liability for the Taxes that are the subject of a Tax Contest with respect to any Separate Return, then subject to Section 6.5 and Section 6.6, Spinco shall have the sole responsibility and right to control the prosecution of such Tax Contest, including the exclusive right to communicate with agents of the applicable Taxing Authority and to control, resolve, settle, or agree to any deficiency, claim or adjustment proposed, asserted or assessed in connection with or as a result of such Tax Contest.
- (b) With respect to any Tax Contest other than those described in Section 6.2(a), subject to Section 6.5 or Section 6.6, Remainco shall have the sole responsibility and right to control the prosecution of such Tax Contest, including the exclusive right to communicate with agents of the applicable Taxing Authority and to control, resolve, settle, or agree to any deficiency, claim or adjustment proposed, asserted or assessed in connection with or as a result of such Tax Contest.

Section 6.3 Joint Return. In the case of any Tax Contest with respect to any Joint Return, Remainco shall, subject to Section 6.5 and Section 6.6, have the sole responsibility and right to control the prosecution of such Tax Contest, including the exclusive right to communicate with agents of the applicable Taxing Authority and to control, resolve, settle or agree to any deficiency, claim or adjustment proposed, asserted, or assessed in connection with or as a result of such Tax Contest.

Section 6.4 Obligation of Continued Notice. During the pendency of any Tax Contest or threatened Tax Contest, each of the Parties shall provide prompt notice to the other Party of any written communication received by it or a member of its respective Group from a Taxing Authority regarding any Tax Contest for which it is indemnified by the other Party hereunder or for which it may be required to indemnify the other Party hereunder. Such notice shall attach copies of the pertinent portion of any written communication from a Taxing Authority and contain factual information (to the extent known) describing any asserted Tax liability in reasonable detail and shall be accompanied by copies of any notice and other documents received from any Taxing Authority in respect of any such matters. Such notice shall be provided in a reasonably timely fashion; provided, however, that in the event that timely notice is not provided, a Party shall be relieved of its obligation to indemnify the other Party only to the extent that such delay results in actual increased costs or actual prejudice to such other Party.

Section 6.5 Settlement Rights. Unless waived by the Parties in writing, in connection with any potential adjustment in a Tax Contest as a result of which adjustment the Non-Controlling Party may reasonably be expected to become liable to make any indemnification payment to the Controlling Party under this Agreement: (i) the Controlling Party shall keep the Non-Controlling Party informed in a timely manner of all actions taken or proposed to be taken by the Controlling Party with respect to such potential adjustment in such Tax Contest; (ii) the Controlling Party shall timely provide the Non-Controlling Party with copies of any correspondence or filings submitted to any Taxing Authority or judicial authority in connection with such potential adjustment in such Tax Contest; (iii) the Controlling Party shall defend such Tax Contest diligently and in good faith; and (iv) the Controlling Party shall not settle or agree to any deficiency, claim or adjustment proposed, asserted or assessed without the prior written consent of the Non-Controlling Party (not to be unreasonably withheld, conditioned or delayed). The failure of the Controlling Party to take any action specified in the preceding sentence with respect to the Non-Controlling Party shall not relieve the Non-Controlling Party of any liability and/or obligation which it may have to the Controlling Party from any other liability or obligation which it may have to the Controlling Party.

Section 6.6 Tax Contest Participation. Unless waived by the Parties in writing, the Controlling Party shall provide the Non-Controlling Party with written notice reasonably in advance of, and the Non-Controlling Party shall have the right to attend and participate in, any formally scheduled meetings with Taxing Authorities or hearings or proceedings before any judicial authorities in connection with any potential adjustment in a Tax Contest pursuant to which the Non-Controlling Party may reasonably be expected to become liable to make any indemnification payment to the Controlling Party under this Agreement (including any Tax Contest related to the Tax-Free Status of the Transactions) or may reasonably be expected to give rise to Tax liabilities of the Non-Controlling Party for any Post-Distribution Period. The failure of the Controlling Party to provide any notice specified in this Section 6.7 to the Non-Controlling Party shall not relieve the Non-Controlling Party of any liability and/or obligation which it may have to the Controlling Party under this Agreement except to the extent that the Non-Controlling Party was actually harmed by such failure, and in no event shall such failure relieve the Non-Controlling Party.

ARTICLE VII COOPERATION

Section 7.1 General.

- (a) Each Party shall fully cooperate, and shall cause all members of such Party's Group to fully cooperate, with all reasonable requests in writing from the other Party, or from an agent, representative or advisor to such Party, in connection with the preparation and filing of any Tax Return, claims for Refunds, the conduct of any Tax Contest, and calculations of amounts required to be paid pursuant to this Agreement, in each case, related or attributable to or arising in connection with Taxes of either Party or any member of either Party's Group covered by this Agreement and the establishment of any reserve required in connection with any financial reporting (a "Tax Matter"). Such cooperation shall include the provision of any information reasonably necessary or helpful in connection with a Tax Matter and shall include, without limitation, at each Party's own cost:
 - (i) the provision of any Tax Returns of either Party or any member of either Party's Group, books, records (including information regarding ownership and Tax basis of property), documentation and other information relating to such Tax Returns, including accompanying schedules, related work papers, and documents relating to rulings or other determinations by Taxing Authorities; and
 - (ii) the execution of any document (including any power of attorney) in connection with any Tax Contest of either Party or any member of either Party's Group, or the filing of a Tax Return or a Refund claim of either Party or any member of either Party's Group.

Each Party shall make its employees and facilities available, without charge, on a mutually convenient basis to facilitate such cooperation.

Section 7.2 Consistent Treatment. Unless and until there has been a Final Determination to the contrary, each Party agrees not to take any position on any Tax Return, in connection with any Tax Contest or otherwise that is inconsistent with (a) the treatment of payments between the Remainco Group and the Spinco Group as set forth in Section 5.4, (b) the Tax Materials or (c) the Tax-Free Status of the Transactions.

ARTICLE VIII RETENTION OF RECORDS; ACCESS

Section 8.1 Retention of Records. For so long as the contents thereof may become material in the administration of any matter under applicable Tax Law, but in any event until the later of (i) sixty (60) days after the expiration of any applicable statutes of limitation (including any waivers or extensions thereof) and (ii) seven (7) years after the Distribution Date, the Parties shall retain records, documents, accounting data and other information (including computer data) necessary for the preparation and filing of all Tax Returns (collectively, "Tax Records") in respect of Taxes of any member of either the Remainco Group or the Spinco Group for any Pre-Distribution Period or Post-Distribution Period or for any Tax Contests relating to such Tax Returns. At any time after the Distribution Date when the Remainco Group proposes to destroy any Tax Records, the Remainco Group shall first notify the Parent Group in writing, and the Parent Group shall be entitled to receive such records or documents proposed to be destroyed. At any time after the Distribution Date when the Spinco Group or Parent Group proposes to destroy any Tax Records, Spinco or Parent, as appropriate, shall first notify Remainco in writing and the Remainco Group shall be entitled to receive such records or documents proposed to be destroyed. The Parties will notify each other in writing of any waivers or extensions of the applicable statute of limitations that may affect the period for which the foregoing records or other documents must be retained.

Section 8.2 Access to Tax Records. The Parties and their respective Affiliates shall make available to each other for inspection and copying during normal business hours upon reasonable notice all Tax Records (including, for the avoidance of doubt, any pertinent underlying data accessed or stored on any computer program or information technology system) in their possession and shall permit the other Party and its Affiliates, authorized agents and representatives and any representative of a Taxing Authority or other Tax auditor direct access, during normal business hours upon reasonable notice to any computer program or information technology system used to access or store any Tax Records, in each case to the extent reasonably required by the other Party in connection with the preparation of Tax Returns or financial accounting statements, audits, litigation, or the resolution of items pursuant to this Agreement. The Party seeking access to the records of the other Party shall bear all costs and expenses associated with such access, including any professional fees.

ARTICLE IX DISPUTE RESOLUTION

Section 9.1 Dispute Resolution. In the event of any dispute between the Parties as to any financial matter covered by this Agreement, the Parties shall appoint a nationally recognized independent public accounting firm (the "Accounting Firm") to resolve such dispute. In this regard, the Accounting Firm shall make determinations with respect to the disputed items based solely on representations made by Remainco, Spinco, Parent, and their respective representatives, and not by independent review, and shall function only as an expert and not as an arbitrator and shall be required to make a determination in favor of one Party only. The Parties shall require the Accounting Firm to resolve all disputes no later than ninety (90) days after the submission of such dispute to the Accounting Firm, but in no event later than the due date for the payment of Taxes or the filing of the applicable Tax Return, if applicable, and agree that all decisions by the Accounting Firm with respect thereto shall be final and conclusive and binding on the Parties. The Accounting Firm shall resolve all disputes in a manner consistent with this Agreement and, to the extent not inconsistent with this Agreement, in a manner consistent with the Past Practices of Remainco and its Subsidiaries, except as otherwise required by applicable Law. The Parties shall require the Accounting Firm to render all determinations in writing and to set forth, in reasonable detail, the basis for such determination. The fees and expenses of the Accounting Firm shall be borne equally by Remainco, on the one hand, and Parent and Spinco, on the other hand.

ARTICLE X MISCELLANEOUS PROVISIONS

Section 10.1 Entire Agreement; Construction. This Agreement shall constitute the entire agreement between the Parties with respect to the subject matter hereof and shall supersede all previous negotiations, commitments, course of dealings and writings with respect to such subject matter. Except as expressly set forth in this Agreement, the Separation Agreement or any Ancillary Agreement: (i) all matters relating to Taxes and Tax Returns of the Parties and their respective Subsidiaries shall be governed exclusively by this Agreement and (ii) for the avoidance of doubt, in the event of any conflict between this Agreement, on the one hand, and the Separation Agreement or any Ancillary Agreement, on the other hand, with respect to such matters, the terms and conditions of this Agreement shall govern. Notwithstanding the foregoing, in the event of any conflict between this Agreement and the Employee Matters Agreement with respect to the Company Equity Awards, the Spinco Equity Awards (as such terms are defined in the Employee Matters Agreement), payroll Taxes, or Code Section 409A, the Employee Matters Agreement shall govern.

Section 10.2 Interest on Late Payments. With respect to any payment between the Parties pursuant to this Agreement not made by the due date set forth in this Agreement for such payment, the outstanding amount will accrue interest at a rate per annum equal to the rate in effect for underpayments under Section 6621 of the Code from such due date to and including the payment date.

Section 10.3 Successors and Assigns. The provisions of this Agreement and the obligations and rights hereunder shall be binding upon, inure to the benefit of and be enforceable by (and against) the Parties and their respective successors and permitted assigns.

Section 10.4 Subsidiaries. Each of the Parties shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth herein to be performed by any Subsidiary of such Party or by any entity that becomes a Subsidiary of such Party at and after the Distribution Time, to the extent such Subsidiary remains a Subsidiary of the applicable Party.

Section 10.5 Assignability. This Agreement shall not be assignable, in whole or in part, directly or indirectly, by any party hereto without the prior written consent of the other Party, and any attempt to assign any rights or obligations arising under this Agreement without such consent shall be void.

Section 10.6 No Fiduciary Relationship. The duties and obligations of the Parties, and their respective successors and permitted assigns, contained herein are the extent of the duties and obligations contemplated by this Agreement; nothing in this Agreement is intended to create a fiduciary relationship between the Parties hereto, or any of their successors and permitted assigns, or create any relationship or obligations other than those explicitly described.

Section 10.7 Further Assurances. Subject to the provisions hereof, the Parties hereto shall make, execute, acknowledge and deliver such other instruments and documents, and take all such other actions, as may be reasonably required in order to effectuate the purposes of this Agreement and to consummate the transactions contemplated hereby.

Section 10.8 Survival. Notwithstanding any other provision of this Agreement to the contrary, all representations, covenants and obligations contained in this Agreement shall survive until the expiration of the applicable statute of limitations with respect to any such matter (including extensions thereof).

Section 10.9 Notices. All notices, requests, claims, demands and other communications under this Agreement and, to the extent applicable shall be in English, shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by registered or certified mail (return receipt requested), or by e-mail (provided confirmation of transmission is electronically generated and kept on file by the sending party), to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this Section 10.9):

If to Remainco, to:

Inpixon 2479 E. Bayshore Road, Suite 195 Palo Alto, California 94303 Attn: Nadir Ali, Chief Executive Officer Email: nadir.ali@inpixon.com

If to Spinco, to:

CXApp Holding Corp.
Four Palo Alto Square, Suite 200
3000 El Camino Real
Palo Alto, California 94306
Attn: Khurram Sheikh, Chief Executive Officer
Email: khurram@kins-tech.com

If to Parent, to:

KINS Technology Group Inc. Four Palo Alto Square, Suite 200 3000 El Camino Real Palo Alto, California 94306 Attn: Khurram Sheikh, Chief Executive Officer Email: khurram@kins-tech.com

Section 10.10 Counterparts. This Agreement may be executed in more than one counterpart, all of which shall be considered one and the same agreement, and shall become effective when one or more such counterparts have been signed by each of the Parties and delivered to each of the Parties.

Section 10.11 Consents. Any consent required or permitted to be given by any Party to the other Party under this Agreement shall be in writing and signed by the Party giving such consent and shall be effective only against such Party (and its Group).

Section 10.12 Expenses. Except as otherwise specified in this Agreement, or as otherwise agreed in writing between Remainco, Parent, and Spinco, Remainco, Parent, and Spinco shall each be responsible for its own fees, costs and expenses paid or incurred in connection with this Agreement.

Section 10.13 Termination and Amendment. This Agreement may not be terminated, modified or amended except by an agreement in writing signed by Remainco, Parent, and Spinco.

Section 10.14 Titles and Headings. Titles and headings to articles herein are inserted for the convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.

Section 10.15 Severability. In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein shall not in any way be affected or impaired thereby. The Parties shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.

Section 10.16 Interpretation. The Parties have participated jointly in the negotiation and drafting of this Agreement. This Agreement shall be construed without regard to any presumption or rule requiring construction or interpretation against the Party drafting or causing any instrument to be drafted.

Section 10.17 No Duplication; No Double Recovery. Nothing in this Agreement is intended to confer to or impose upon any Party a duplicative right, entitlement, obligation or recovery with respect to any matter arising out of the same facts and circumstances.

Section 10.18 No Waiver. No failure to exercise and no delay in exercising, on the part of any Party, any right, remedy, power or privilege hereunder shall operate as a waiver hereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

Section 10.19 Governing Law. This Agreement and any dispute arising out of, in connection with or relating to this Agreement shall be governed by and construed in accordance with the Laws of the State of Delaware, without giving effect to the conflicts of laws principles thereof.

Section 10.20 Distribution Time. This Agreement shall become effective only upon the Distribution Time on the Distribution Date.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties hereto have duly executed this Agreement as of the day and year first above written.

KINS TECHNOLOGY GROUP INC.

By: /s/ Khurram Sheikh

Name: Khurram P. Sheikh Title: Chief Executive Officer

INPIXON

By: /s/ Nadir Ali

Name: Nadir Ali

Title: Chief Executive Officer

CXAPP HOLDING CORP.

By: /s/ Nadir Ali

Name: Nadir Ali

Title: Chief Executive Officer

TRANSITION SERVICES AGREEMENT

This TRANSITION SERVICES AGREEMENT (this "Agreement"), effective as of the Distribution Time of the Separation and Distribution Agreement (as defined below) (the "Effective Date"), by and between Inpixon, a Nevada corporation ("Inpixon"), and CXApp Holding Corp., a Delaware corporation ("CXApp"). Each of Inpixon and CXApp may be referred to herein individually as a "Party" and collectively as the "Parties".

WHEREAS, Inpixon and CXApp are parties to a certain Separation and Distribution Agreement dated as of September 25, 2022, among the Parties, Design Reactor, Inc., a California corporation, and KINS Technology Group Inc., a Delaware corporation (the "Separation and Distribution Agreement"), pursuant to which Inpixon has agreed to provide to CXApp, or CXApp has agreed to provide to Inpixon, the services described herein during the term of this Agreement, on the terms and subject to the conditions contained herein.

NOW, THEREFORE, in consideration of the premises and the mutual promises and conditions hereinafter set forth and set forth in the Separation and Distribution Agreement, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, do hereby agree as set forth herein.

ARTICLE I DEFINITIONS

- 1.1 <u>Certain Defined Terms</u>. Unless otherwise specifically provided herein, capitalized terms used, but not otherwise defined, herein shall have the meanings ascribed thereto in the Separation and Distribution Agreement. As used herein, the following terms have the following meanings.
 - (a) "Affiliate(s)" means, with respect to a particular entity or Person, any Person that controls, is controlled by, or is under common control with that Party. For the purpose of this definition, "control" will mean, direct or indirect ownership of more than 50% of the shares of stock entitled to vote for the election of directors, in the case of a corporation, or more than 50% of the equity interest in the case of any other type of legal entity, status as a general partner in any partnership, or any other arrangement whereby the entity or Person controls or has the right to control the board of directors or equivalent governing body of a corporation or other entity, or the ability to cause the direction of the management or policies of a corporation or other entity. For purposes of this Agreement, Inpixon and CXApp shall not be considered Affiliates of each other.
 - (b) "CXApp Recipient" means, with respect to a particular CXApp Transition Service, either CXApp or the applicable member of the CXApp Group (as defined in the Separation and Distribution Agreement) receiving such CXApp Transition Service.
 - (c) "Governmental Authority" means (a) any court, agency, department, authority or other instrumentality of any national, state, county, city or other political subdivision; (b) any public international organization; or (c) any department, agency or instrumentality thereof, including any company, business, enterprise or other entity owned or controlled, in whole or in part, by any government.
 - (d) "Inpixon Recipient" means, with respect to a particular Inpixon Transition Service, either Inpixon or the applicable member of the Inpixon Group (as defined in the Separation and Distribution Agreement) receiving such Inpixon Transition Service.

- (e) "Intellectual Property" means any and all intellectual property and other proprietary rights throughout the world, including any and all state, United States, international or foreign or other territorial or regional rights in, arising out of or associated with any of the following: (a) all patents and applications therefor, including all related provisionals, continuations, continuations-in-part, divisionals, reissues, renewals and extensions ("Patents"), (b) all inventions (whether patentable or not), invention disclosures, improvements, trade secrets, proprietary information, know how (including formulations, specifications, formulae, manufacturing and other processes, operating procedures, methods, techniques and all research and development information), technology, technical data and customer lists, and all documentation relating to any of the foregoing, (c) all copyrights, copyrightable works, copyright registrations and applications therefor, including all rights of authorship, use, publication, reproduction, distribution, performance and transformation ("Copyrights"), (d) all industrial designs and any registrations and applications therefor, (e) all domain names, uniform resource locators and other names and locators associated with the internet ("Domain Names"), and all social media accounts and handles and appregistrations, (f) all trade names, logos, common law trademarks and service marks, trademark and service mark registrations and applications therefor and all goodwill associated therewith ("Trademarks"), (g) all rights in databases and data collections, (h) all moral and economic rights of authors and inventors, however denominated, (i) rights in computer software (including source code, object code, firmware, algorithms, operating systems and specifications) and related technology, (j) all rights in documentation, files, cgi and other scripts and programming code, (k) all rights of publicity or privacy, including with respect to name, likeness or persona, and (l) all rights to sue or reco
 - (f) "Law" means any law (including common law), statute, code, ordinance, rule, regulation, order or charge of any Governmental Authority.
 - (g) "Person" means any individual, partnership, limited liability company, firm, corporation, association, trust, unincorporated organization or other entity.
- (h) "Personal Information" means any data or information that identifies, relates to, describes, is reasonably capable of being associated with, or could reasonably be linked, directly or indirectly, with a particular natural person or household and any information derived from the foregoing.
 - (i) "Provider" means Inpixon with respect to the CXApp Transition Services or CXApp with respect to the Inpixon Transition Services, as applicable.
- (j) "Recipient" means the CXApp Recipient with respect to the CXApp Transition Services or the Inpixon Recipient with respect to the Inpixon Transition Services, as applicable.
- (k) "Representatives" means, as to any Person, such Person's Affiliates and its and their successors, owners, controlling Persons, directors, officers, employees, agents, representatives, subcontractors, or other third party acting for or on its behalf, including, as to Provider, any Vendor providing any Transition Services as permitted in this Agreement.

ARTICLE II TRANSITION SERVICES PROVIDED

2.1 <u>Transition Services</u>.

- (a) Upon the terms and subject to the conditions set forth in this Agreement, Inpixon shall provide, or cause one or more of its Representatives to provide, to the applicable CXApp Recipient each of the services set forth on Schedule A attached hereto (hereinafter referred to individually as a "CXApp Transition Service", and collectively as the "CXApp Transition Services"), at the corresponding costs set forth on Schedule A, and the CXApp Recipient agrees to receive the CXApp Transition Services and pay the costs therefor during the time period specified for each such CXApp Transition Service in such Schedule or for such other time period as permitted pursuant to this Agreement (hereinafter referred to collectively as the "CXApp Service Periods" for all of the CXApp Transition Services, and individually a "CXApp Service Period" for each CXApp Transition Services from time to time upon mutual agreement by executing a signed amendment to Schedule A.
- (b) Upon the terms and subject to the conditions set forth in this Agreement, CXApp shall provide, or cause one or more of its Representatives to provide, to the applicable Inpixon Recipient each of the services set forth on Schedule B attached hereto (hereinafter referred to individually as a "Inpixon Transition Service", and collectively as the "Inpixon Transition Services", and, together with the CXApp Transition Services, the "Transition Services"), at the corresponding costs set forth on Schedule B, and the Inpixon Recipient agrees to receive the Inpixon Transition Services and pay the costs therefor during the time period specified for each such Inpixon Transition Service in such Schedule or for such other time period as permitted pursuant to this Agreement (hereinafter referred to collectively as the "Inpixon Service Periods" for all of the Inpixon Transition Services, and individually a "Inpixon Service Period" for each Inpixon Transition Service, and, together with the CXApp Service Periods, the "Service Periods"). The Parties may amend the scale and scope of the Inpixon Transition Services from time to time upon mutual agreement by executing a signed amendment to Schedule B.
- (c) If, during the Term, Recipient identifies in good faith any service that was provided by Provider or one of its Affiliates (excluding the Recipient's Group) to the Enterprise Apps Business (as defined in the Separation and Distribution Agreement) in the case of a CXApp Recipient or the Inpixon Retained Business (as defined in the Separation and Distribution Agreement) in the case of an Inpixon Recipient, as applicable (the "Applicable Business"), prior to the Effective Date that is not listed on Schedule A or Schedule B, as applicable, and is necessary to (i) effectuate the Separation or (ii) operate the Applicable Business (an "Omitted Service"), then Recipient shall notify Provider thereof and the Parties shall cooperate in good faith in determining whether there is a mutually acceptable arm's length basis on which one Party will provide such Omitted Service to the other Party in exchange for a fee which shall be set forth on an amended Schedule A or Schedule B as may be applicable.
- 2.2 Personnel; Affiliates; Vendors. In providing the Transition Services, Provider may, as it deems necessary or appropriate, (i) use the qualified personnel of Provider or its Affiliates, and (ii) employ the services of qualified third parties ("Vendors") to the extent that, and subject to the condition that, such Vendor's services (A) were utilized by or for the benefit of the Applicable Business prior to the Effective Date, (B) are routinely utilized to provide similar services to other businesses of Provider or (C) are reasonably necessary for the efficient performance of such Transition Services. Notwithstanding the foregoing, Provider shall not, without first complying with vendor diligence and other risk management processes and procedures at least as stringent as Provider would undertake in the ordinary course of business for onboarding vendors or other service providers to perform the same or similar types of services for Provider or its Affiliates (the "Vendor Diligence and Risk Management Standard"), employ the services of a Vendor in providing the Transition Services if such Vendor will (i) have, process or otherwise have access to Recipient's Confidential Information or Recipient's information systems; (ii) provide a material component of any Transition Service; (iii) provide a service, feature or functionality that is customer-facing or public-facing; or (iv) use any Trademark of Recipient. Furthermore, each Party shall, and shall cause its Representatives to, comply, in all material respects, with all Laws which may be applicable to the Transition Services. Each Party shall be responsible for its Representatives, including for such Representatives adhering to any reasonable written health, safety, and security regulations and other published policies of the other Party provided in advance while on the other Party's premises or when given access to any equipment, computer, databases, systems, software, network or other files (collectively, "Systems") owned or controlled by the other Party (the "System Owner"). If a Party or one or more of its Representatives needs access to the premises or Systems of the other Party or one or more of its Representatives to provide or receive the Transition Services (as applicable), then (x) the accessing Party shall advise the other Party in writing in advance of such access of the name of each of the accessing Party's Representatives who shall require such access, (v) the accessing Party and its Representatives shall not attempt to obtain access to, use or interfere with any of the premises or Systems of the other Party or such other Party's Representatives, except to the extent permitted by the other Party or required to do so to provide or receive the Transition Services (as applicable), and (z) the accessing Party and its Representatives shall not intentionally damage, disrupt or impair the normal operation of any of the premises or Systems of the other Party or such other Party's Representatives. Additionally, the accessing Party shall not (and shall ensure that its Representatives shall not): (i) use the System Owner's Systems to develop software, process data or perform any work or services other than for the purpose of exercising its rights or performing its obligations under this Agreement or (ii) obtain, or attempt to obtain, access to any hardware, program or data stored in the System Owner's Systems except to the extent reasonably necessary to exercise the accessing Party's rights or perform its obligations under this Agreement. All user identification numbers and passwords for the Systems disclosed to the accessing Party, and any information obtained from the use of the System Owner's Systems, shall be deemed Confidential Information of the System Owner.

2.3 <u>Coordinators</u>. Each of Provider and Recipient shall nominate a representative to act as its primary contact person to coordinate the provision of all Transition Services (collectively, the "*Primary Coordinators*"). Each Primary Coordinator may designate one or more service coordinators for each specific Transition Service (the "*Service Coordinators*"). Each Party may treat an act of a Primary Coordinator or Service Coordinator of another Party as being authorized by such other Party without inquiring behind such act or ascertaining whether such Primary Coordinator or Service Coordinator had authority to so act, *provided, however*, that no such Primary Coordinator or Service Coordinator has authority to amend this Agreement. Provider and Recipient shall advise each other promptly (in any case no more than five (5) business days) in writing of any change in the Primary Coordinators and any Service Coordinator for a particular Transition Service, setting forth the name of the Primary Coordinator or Service Coordinator to be replaced and the name of the replacement, and certifying that the replacement Primary Coordinator or Service Coordinator is authorized to act for such Party in all matters relating to this Agreement, in the case of a Primary Coordinator or, in the case of a Service Coordinator, with respect to the Transition Service for which such Service Coordinators has been designated. Provider and Recipient each agrees that all communications relating to the provision of the Transition Service shall be directed to the Service Coordinators for such Transition Service with copies to the Primary Coordinators. Inpixon's initial Primary Coordinator shall be Nadir Ali. CXApp's initial Primary Coordinator shall be Khurram Sheikh.

2.4 Level of Transition Services.

- (a) Recipient acknowledges and agrees that Provider is not in the business of providing services to third parties and is entering into this Agreement only in connection with the Separation and Distribution Agreement. Provider shall, and shall cause each of its Representatives to, provide the Transition Services with substantially the same degree of skill, quality and standard of care as that utilized by Provider (or its Affiliates) to perform similar activities in the six (6) month period (or twelve (12) month period solely with respect to activities that are customarily performed on an annual basis) prior to the Effective Date, and, in any event, no less than with commercially reasonable care and diligence (collectively, the "Services Standard"). Under no circumstances shall Provider or any of its Representatives be held accountable to a greater standard of care, efforts or skill than the Services Standard in the performance of the Transition Services. Recipient acknowledges and agrees that the Transition Services do not include the exercise of business judgment or general management.
- (b) If the Transition Services to be provided to Recipient materially increase in scale or in scope as compared to the level of the services reasonably anticipated as of the Effective Date to be provided based on the projected growth or natural evolution of the Applicable Business, Provider may, at its election, choose to not provide such increased scale or scope of Transition Services, and if Provider elects to perform such increased scale or scope of Transition Services, all costs incurred in connection therewith shall be mutually agreed upon by Provider and Recipient prior to the time such additional Transition Services are performed, shall be set forth in an amended Schedule A or Schedule B, as applicable, and shall be borne by Recipient.

- (c) In addition to being subject to the terms and conditions of this Agreement for the provision of the Transition Services, Provider and Recipient each agree that the Transition Services provided by any Vendor shall be subject to the terms and conditions of any agreements between Provider and such Vendor, which agreements shall be on substantially the same conditions as Provider would enter into with such Vendor for its own account, and no such agreements shall be binding on Recipient after the Term hereof without Recipient's express written consent. Provider shall consult with Recipient concerning the terms and conditions of any such agreements to be entered into, or proposed to be entered into, or amended, with any Vendors after the Effective Date.
- (d) Without relieving Provider of its obligation to perform the Transition Services in accordance with the Services Standard, Provider shall not be (i) obligated to perform the Transition Services to the extent that such performance would be unlawful or that would require Provider to violate applicable Law; (ii) obligated to perform the Transition Services to the extent that such performance, in Provider's reasonable determination, could create deficiencies in Provider's controls over financial information or adversely affect the maintenance of Provider's financial books and records or the preparation of its financial statements; (iii) obligated to hire any additional employees to perform the Transition Services or maintain the employment of any specific employee; (iv) obligated to hire replacements for employees that resign, retire or are terminated; (v) obligated to enter into retention agreements with employees or otherwise provide any incentive beyond payment of regular salary and benefits; (vi) prevented from transferring after the Effective Date any employees who were supporting the business operations as of the Effective Date to support other business operations for Provider or its Affiliates or to assume other roles with Provider or its Affiliates to the extent such employees are not required to provide Transition Services; (vii) prevented from determining, in its sole discretion, the individual employees who will provide Transition Services; or (viii) obligated to purchase, lease or license any additional equipment or software.
- 2.5 <u>Location of Services Provided; Travel Expenses.</u> Provider shall provide the Transition Services to Recipient from locations of Provider's choice in its sole discretion except to the extent the nature of the Transition Services necessitates performance at a specific location, as mutually agreed upon by the Parties. Subject to <u>Section 3.1</u>, should the provision of the Transition Services require any directors, officers, employees, agents, representatives, or subcontractors of Provider or its Affiliates to travel beyond fifty (50) miles from his or her employment location, Recipient shall reimburse Provider for all reasonable travel-related out-of-pocket costs, consistent with Provider's travel policy as provided to Recipient in advance in writing.

2.6 Limitation of Liability.

The Parties hereto acknowledge and agree that the Transition Services are provided by Provider: (a) at the request of Recipient in order to accommodate it following the closing under the Separation and Distribution Agreement; (b) at the costs set forth on Schedule A or Schedule B hereto, as applicable, and with no expectation of profit being made by Provider thereon; and (c) with the expectation that Provider is not assuming any financial or operational risks, including those usually assumed by a service provider, except for those risks explicitly set forth herein. Accordingly, each Party agrees that, absent gross negligence or willful misconduct, and except for breaches of Article V (Confidentiality) and except for a Party's obligations under Section 2.7 (Indemnification), the other Party, its Affiliates and their directors, officers, employees, representatives, consultants and agents shall not be liable for any indirect, special, incidental or consequential damages, including lost profits or savings, whether or not such damages are foreseeable, or for any third party claims relating to the Transition Services or to each Party's performance under this Agreement. Notwithstanding anything to the contrary contained herein, in the event Provider commits an error with respect to or incorrectly performs or fails to perform any Transition Service, at Recipient's request, Provider shall use commercially reasonable efforts and in good faith attempt to correct such error, re-perform or perform such Transition Service at no additional cost to Recipient; provided that, absent gross negligence or willful misconduct, and assuming that Provider uses commercially reasonable data backup processes, Provider shall have no obligation to recreate any lost or destroyed data to the extent the same cannot be cured by the re-performance of the Transition Service in question.

2.7 Indemnification.

- (a) Recipient shall indemnify, defend and hold harmless Provider and its Affiliates and its and their respective officers, directors, employees, representatives, subcontractors and agents from and against any and all damages, liabilities, losses, taxes, fines, penalties, costs and expenses (including, without limitation, reasonable fees of counsel) incurred by any of them in connection with any Third Party Claim (as defined below) (each, a "Loss" and, collectively, the "Losses") relating to, arising out of or resulting from or based on (i) Recipient's material breach of this Agreement or (ii) any gross negligence or willful misconduct of Recipient, except in each case of (i) or (ii) to the extent such Losses are subject to indemnification pursuant to Section 2.7(b).
- (b) Provider hereby agrees to indemnify, defend and hold harmless Recipient and its Affiliates and its and their respective officers, directors, employees, representatives, subcontractors and agents from and against any and all Losses relating to, arising out of or resulting from (i) Provider's breach of this Agreement, (ii) any gross negligence or willful misconduct in the performance of its obligations under this Agreement or (iii) actual or alleged infringement, misappropriation or violation of Intellectual Property arising out of or in connection with the receipt or use of the Transition Services provided by or on behalf of Provider (excluding any actual or alleged infringement, misappropriation or violation of Intellectual Property to the extent caused by (x) any instruction, information, designs, specifications, or other materials provided by Recipient to Provider, (y) use of the Services in combination with any materials or equipment not supplied or specified by Provider or (z) any modifications or changes made to the Services by or on behalf of any Person other than Provider), except in each case of (i), (ii) or (iii) to the extent such Losses are subject to indemnification pursuant to Section 2.7(a).

2.8 Indemnification Procedures.

(a) If any claim or demand is made by a third party (including any action or proceeding commenced or threatened to be commenced) with respect to which a Party seeking indemnification (the "Indemnified Party") intends to seek indemnity under Section 2.7 (a "Third Party Claim"), the Indemnified Party shall promptly give written notice thereof to the other Party (the "Indemnifying Party") indicating, with reasonable specificity, the nature of such Third Party Claim, the basis therefor, and a copy of any documentation received from such third party. A failure by the Indemnified Party to give notice and to tender the defense of any action or proceeding in a timely manner pursuant to this Section 2.8(a) shall not limit the obligation of the Indemnifying Party under Section 2.7, except to the extent such Indemnifying Party is actually and materially prejudiced thereby.

(b) Upon receipt of a notice for indemnity from the Indemnified Party pursuant to Section 2.8(a) with respect to any Third Party Claim, the Indemnifying Party shall have the right to assume the defense of, at its own expense and by its own counsel, any such Third Party Claim. If the Indemnifying Party shall, in accordance with the immediately preceding sentence, undertake to compromise or defend any such Third Party Claim, it shall notify the Indemnified Party of its intention to do so, and the Indemnified Party shall agree to cooperate with the Indemnifying Party and its counsel in the compromise of, or defense against, any such Third Party Claim; provided that the Indemnifying Party shall not settle or compromise any such Third Party Claim without the written consent of the Indemnified Party (not to be unreasonably withheld, conditioned or delayed) unless such settlement or compromise fully and irrevocably releases the Indemnified Party in connection with such Third Party Claim and provides relief consisting solely of money damages borne by the Indemnifying Party. Notwithstanding an election of the Indemnifying Party to assume the defense of such Third Party Claim, the Indemnified Party shall have the right to employ separate legal counsel, its own cost and expense, and to participate in the defense thereof.

2.9 Modification of Transition Services Procedures.

- (a) Without limiting Section 2.4, Provider may make changes from time to time in its standards and procedures for performing the Transition Services, provided that any such change shall not interfere in any material respect with the continued provision or cost of the Transition Services. Notwithstanding the foregoing sentence, unless required by Law, Provider shall not implement any substantial or material changes to such standards and procedures in a manner affecting the operation of the Applicable Business unless Recipient agrees in writing to such changes and Provider gives Recipient ten (10) business days to adapt its operations to accommodate such changes to the extent commercially reasonable.
- (b) During the term of this Agreement, if Recipient intends to make any changes that may affect the provision of any of the Transition Services, Recipient shall provide Provider with a plan identifying any changes as soon as reasonably practicable, but in any case no less than ten (10) business days before implementing such changes; provided, however, that Provider shall not be required to alter the method in which it provides any of the Transition Services or increase the level of any such Transition Services in any material manner except as expressly provided herein; provided, further, however, that the failure of Recipient to provide such notice shall not alter or diminish Provider's obligations to provide the Transition Services on the terms set forth herein except where the failure to provide notice has materially increased Provider's cost or burden to provide such Transition Service.
- 2.10 <u>Cooperation</u>. The Parties will use commercially reasonable efforts to reasonably cooperate and cause each of their respective Representatives to reasonably cooperate in a professional and workmanlike manner with each other to the extent necessary to assist the other Party in performance of its obligations under this Agreement, including with respect to the provision and receipt of the Transition Services. Such cooperation shall include exchanging information relevant to and reasonably necessary for the provision or receipt of the Transition Services hereunder and the performance of such other duties and tasks as may be reasonably required for the provision or receipt of the Transition Services. Without limiting the foregoing:
 - (a) Recipient shall permit Provider and its Representatives reasonable access during regular business hours (or otherwise upon reasonable prior notice) to any data, records and personnel involved in receiving or overseeing the Transition Services as reasonably required by Provider to facilitate Provider's performance of this Agreement. Any such data and records shall be subject to Article V. Before the Parties exchange any Personal Information in connection with the Transition Services, the Parties will enter into a data processing agreement in accordance with applicable Laws.

- (b) Provider shall obtain at its sole cost and expense any consents, licenses, waivers or approvals necessary to permit Provider to perform its obligations hereunder; provided, however, that under no circumstances shall Provider be relieved of its obligation to provide the relevant part of any Transition Services to the extent that Provider is unable to obtain necessary third party consents, licenses, waivers or approvals relating to such part of the Transition Services.
- (c) Recipient shall obtain all necessary consents, licenses, waivers and approvals necessary for it to receive the Transition Services and perform its obligations under this Agreement.

ARTICLE III COMPENSATION

- 3.1 Consideration. As consideration for the Transition Services, Recipient shall pay to Provider the amount specified for each such Transition Service as set forth in Schedule A or Schedule B, as applicable, including any "pass-through costs" expressly identified as such in Schedule A or Schedule B, as applicable. The fees set forth on Schedule A and Schedule B will be equitably reduced if any Transition Service is suspended, terminated or removed from the scope of this Agreement and will be equitably prorated for partial months. In addition, Recipient shall reimburse Provider (upon receipt of applicable receipts and other reasonable supporting documentation if requested by Provider) for all reasonable documented out of pocket costs of Provider in connection with performance of the Transition Services by Provider, including: (a) shipping and transportation costs (including the cost of any insurance related thereto), duties and other taxes (excluding taxes on Provider's income); (b) travel-related costs, (c) out of pocket costs or expenses incurred with third parties by Provider, its Affiliates or subcontractors, including for the extraction, conversion and transfer of data and (d) any other out of pocket costs and expenses incurred with third parties described herein as reimbursable by Provider (the "Reimbursable Expenses"); provided that if any particular Recipients and Expenses exceeds Two Thousand Five Hundred Dollars (\$2,500), Provider must obtain Recipient's consent prior to any obligation of Recipient to reimburse Provider for such Reimbursable Expenses; provided, further that until Recipient consents to such Reimbursable Expenses exceeding Two Thousand Five Hundred Dollars (\$2,500), Provider shall not be required to provide the relevant part of the Transition Services for which such Reimbursable Expenses exceeding Two Thousand Five Hundred Dollars (\$2,500) is necessary.
- 3.2 <u>Invoices</u>. Provider shall, on a monthly basis within fifteen days of the last day of each calendar month, submit a single itemized invoice to Recipient for all Transition Services provided to Recipient during such month. All invoices shall be sent to the attention of the Primary Coordinators at the address set forth in <u>Section 7.5</u> hereof or to such other address as Recipient shall have specified by notice in writing to Provider.

3.3 Payment of Invoices.

- (a) Recipient shall pay any undisputed invoice for Transition Services promptly but in no event later than thirty (30) days after the date of receipt of such invoice and such payment shall be made by wire transfer of immediately available funds to such bank account as shall have been notified in writing to Recipient by Provider. Payment of all invoices in respect of the Transition Services shall be made by check or electronic funds transmission in U.S. Dollars, without any offset or deduction of any nature whatsoever (except that offset or deduction may be made in regard to other invoiced amounts due under this Agreement or to the extent of a dispute in good faith concerning amounts due under this Agreement). All payments shall be made to the account designated by Provider to Recipient.
- (b) If any payment is not paid when due (except to the extent disputed in good faith) and Recipient does not make such payment within sixty (60) days of receiving a past-due notice from Provider, Provider shall have the right, without any liability to Recipient, or anyone claiming by or through Recipient, to, upon written notice to Recipient, immediately cease providing any or all of the Transition Services provided by Provider to Recipient or to terminate this Agreement in its entirety, which right may be exercised by Provider in its sole and absolute discretion. Notwithstanding the above, Provider shall not cease providing any Transition Service or terminate this Agreement if such lack of payment is due to a good faith dispute, the details of which Recipient has indicated to Provider in writing.

3.4 <u>Taxes</u>. The amount specified for each Transition Service as set forth in <u>Schedule A</u> and <u>Schedule B</u> does not include any applicable sales, use, transfer, value-added, goods or services Taxes or similar Taxes imposed or assessed on the provision of the Transition Services (other than gross-receipts based Taxes, and not including any Taxes based upon or calculated by reference to net income, receipts or capital) ("Sales and Services Taxes"), and such Sales and Services Taxes will be separately stated on the relevant invoice. Provider shall be entitled to charge and collect from the Recipient an additional amount equal to such Sales and Service Taxes and shall timely remit such Taxes to the appropriate tax authorities. Provider shall be responsible for any Losses (including any deficiency, interest and penalties) imposed as a result of a failure to timely remit such Taxes if and only if Recipient timely remits such additional amount to Provider; otherwise Recipient shall be responsible for such Losses and shall hold Provider harmless in respect of them. Provider shall cooperate with the Recipient and take any requested action in order to minimize any Sales and Services Taxes imposed on the sale of the Transition Services, including timely providing resale or other applicable Tax exemption certificates or other documentation necessary to support Tax exemption.

ARTICLE IV OWNERSHIP OF INTELLECTUAL PROPERTY

4.1 Ownership; Delivery. Each Party retains the ownership and title to any and all of its data and Intellectual Property as of the Effective Date. Except as expressly set forth herein, neither Party will obtain, by virtue of this Agreement or the Transition Services, by implication or otherwise, any rights of ownership or use of any property or Intellectual Property owned by the other. All Intellectual Property conceived, created or made by Provider or any of its Representatives (whether alone or jointly with Recipient) in the course of Provider's performance of the Transition Services and other activities under this Agreement to the extent (a) related to the Applicable Business or (b) based on, derived from, or improvements of any of Recipient's (i) background Intellectual Property or (ii) Confidential Information or (c) generated by Recipient's use of a Transition Service in the ordinary course of operating the Applicable Business (copyrights in reports, documents or data generated through Recipient's use of a Transition Service) (altogether, (a) and (b) and (c), the "Assigned IP") shall be solely owned by Recipient, and Provider hereby assigns to Recipient all of Provider's right, title, and interest in and to such Assigned IP. All other Intellectual Property conceived, created or made by Provider or any of its Representatives in the course of Provider's or such Representative's performance of any Transition Services or other activities under this Agreement shall be solely owned by Provider. All Intellectual Property created or developed by Recipient or any of its Representatives in connection herewith shall be owned by Recipient.

4.2 Limited Licenses.

- (a) Recipient (on behalf of itself and its controlled Affiliates) hereby grants to Provider a limited, non-exclusive, royalty-free, non-transferable license, with the right to grant sublicenses to its Affiliates and its and their subcontractors during the Service Periods, under the Intellectual Property owned or controlled by Recipient, solely to the extent necessary for Provider and its Affiliates and its and their subcontractors to perform the Transition Services hereunder for the benefit of Recipient during the applicable Services Period.
- (b) Provider (on behalf of itself and its controlled Affiliates) hereby grants to Recipient and its Affiliates a limited, non-exclusive, royalty-free, non-transferable license, with the right to grant sublicenses to its and their Affiliates and subcontractors, under the Intellectual Property owned or controlled by Provider, solely to the extent necessary for Recipient and its Affiliates and its and their subcontractors to (i) receive the Transition Services during the applicable Service Period or (ii) use or exploit any deliverables provided by Provider to Recipient as part of the Transition Services in the operation of the Applicable Business.

ARTICLE V CONFIDENTIALITY

5.1 Confidential Information.

(a) Each Party recognizes that in the performance of this Agreement, or as a result of the Parties' ongoing relationship, Confidential Information (as defined in the Separation and Distribution Agreement) belonging to the other Party regarding the Transition Services may be disclosed or become known to the Party or its Affiliates. Unless otherwise expressed in writing to the other Party, confidential information and confidential materials concerning a Party's business and products (including information and materials contained in technical data, information concerning the Applicable Business, financial information and data, strategies and marketing and customer information), including that expressed orally, that is exchanged between the Parties in connection with the performance of this Agreement shall be considered to be Confidential Information.

(b) Notwithstanding any termination of this Agreement, Provider and Recipient shall hold and shall cause their respective Representatives to hold, in strict confidence (and not to disclose or release or, except as otherwise permitted by this Agreement or the Separation and Distribution Agreement, use, including for any ongoing or future commercial purpose, without the prior written consent of the Party to whom the Confidential Information relates (which may be withheld in such Party's sole and absolute discretion, except where disclosure is required by applicable Law), any and all Confidential Information concerning or belonging to the other Party or its Affiliates; provided that each Party may disclose, or may permit disclosure of, Confidential Information: (i) to its respective auditors, attorneys, financial advisors, bankers and other appropriate consultants and advisors who have a need to know such Information (as defined in the Separation and Distribution Agreement) for auditing and other noncommercial purposes and are informed of the obligation to hold such Information confidential and in respect of whose failure to comply with such obligations, the applicable Party will be responsible, (ii) if any Party or its Affiliates or any of its respective Subsidiaries is required or compelled to disclose any such Confidential Information by judicial or administrative process or by other requirements of Law or stock exchange rule or is advised by outside counsel in connection with a proceeding brought by a Governmental Authority that it is advisable to do so, (iii) as required in connection with any legal or other proceeding by one Party against the other Party or in respect of claims by one Party against the other Party brought in a proceeding, (iv) as necessary in order to permit a Party to prepare and disclose its financial statements in connection with any regulatory filings or tax returns, (v) as necessary for a Party to enforce its rights or perform its obligations under this Agreement or the Separation and Distribution Agreement, (vi) to Governmental Authorities in accordance with applicable procurement regulations and contract requirements or (vii) to other Persons in connection with their evaluation of, and negotiating and consummating, a potential strategic transaction, to the extent reasonably necessary in connection therewith, provided an appropriate and customary confidentiality agreement has been entered into with the Person receiving such Confidential Information at least as protective of such Confidential Information as this Agreement. Notwithstanding the foregoing, in the event that any demand or request for disclosure of Confidential Information is made by a third party pursuant to clause (ii), (iii), (v) or (vi) above, each Party, as applicable, shall promptly notify (to the extent permissible by Law) the Party to whom the Confidential Information relates of the existence of such request, demand or disclosure requirement and shall provide such affected Party a reasonable opportunity to seek an appropriate protective order or other remedy, which such Party will cooperate in obtaining to the extent reasonably practicable. In the event that such appropriate protective order or other remedy is not obtained, the Party which faces the disclosure requirement shall furnish only that portion of the Confidential Information that is required to be disclosed and shall take commercially reasonable steps to ensure that confidential treatment is accorded such Confidential Information.

- (c) Each Party acknowledges that it and its Affiliates may have in its or their possession confidential or proprietary Information of third parties that was received under confidentiality or non-disclosure agreements with such third party while such Party or its Affiliates were part of the Inpixon Group. Each Party shall comply, shall cause its Affiliates to comply, and shall cause its and their respective officers, employees, agents, consultants and advisors (or potential buyers) to comply, with all terms and conditions of any such third-party agreements entered into prior to the Effective Date, with respect to any confidential and proprietary Information of third parties to which it or any other member of its Group has had access.
- (d) Notwithstanding anything to the contrary set forth herein, (i) the Parties shall be deemed to have satisfied their obligations hereunder with respect to Confidential Information if they exercise at least the same degree of care that applies to Inpixon's confidential and proprietary information pursuant to policies in effect as of the Effective Date and (ii) confidentiality obligations provided for in any Contract between each Party or its Affiliates or Subsidiaries and their respective employees shall remain in full force and effect. Notwithstanding anything to the contrary set forth herein, Confidential Information of a Party in the possession of and used by the other Party as of the Effective Date may continue to be used by such Party in possession of the Confidential Information in and only in the operation of the Applicable Business; provided that such Confidential Information may only be used by such Party and its officers, employees, agents, consultants and advisors in the specific manner and for the specific purposes for which it is used as of the Effective Date, and may only be shared with additional officers, employees, agents, consultants and advisors of such Party on a need-to-know basis exclusively with regard to such specified use; provided further that such Confidential Information may be used only so long as the Confidential Information is maintained in confidence and not disclosed in violation of this Section 5.1.
- (e) The Parties agree that irreparable damage may occur in the event that the provisions of this <u>Section 5.1</u> were not performed in accordance with their specific terms. Accordingly, it is hereby agreed that the Parties shall be entitled to seek an injunction or injunctions to enforce specifically the terms and provisions hereof in any court having jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity.
- (f) Upon expiration of the Service Periods or termination of this Agreement for any reason whatsoever, except for such retention and use as expressly provided for in the Separation and Distribution Agreement, each Party shall not disclose and shall make no further use of the other Party's Confidential Information and upon written request shall immediately destroy or, with respect to Confidential Information in written or other tangible form (including all copies thereof), return to the other Party, all such Confidential Information; *provided* that (i) each Party shall be entitled to retain one record copy in its legal department solely to determine the extent of its continuing obligations or as otherwise required to comply with applicable Law, and (ii) neither Party nor its Representatives shall be required to expunge Confidential Information from computer archiving conducted as part of established record retention policies (*provided* that the foregoing shall not be deemed to permit the accessing, retrieval or use of any Confidential Information).

ARTICLE VI TERM

6.1 <u>Term</u>. This Agreement shall commence on the Effective Date and shall continue in full force and effect until the earliest of (a) the date on which this Agreement is terminated in accordance with this <u>Article VI</u> or (b) the expiration of the last Service Period, such that Provider is no longer obligated to provide any Transition Services pursuant to this Agreement (the "*Term*"). If no expiration date is provided for any Transition Service, then such Transition Service will terminate twelve (12) months after the Effective Date, *provided* that Recipient shall have the right to an extension of each or any Transition Service for up to six (6) months by providing written notice to Provider in advance of the original termination date for such Transition Service if, prior to such request for extension, Recipient has used commercially reasonable efforts to establish analogous capabilities of its own. The Parties will discuss in good faith any subsequent requests to further extend the Transition Services.

6.2 Termination of Services.

- (a) Recipient may, at any time prior to the end of the Service Period for any Transition Service(s) and upon thirty (30) days' prior written notice to Provider, terminate any Transition Service(s) or this Agreement in its entirety, whereupon, from and after the date of termination specified in such written notice, Provider's obligation to provide such Transition Service(s) to Recipient shall cease and Recipient shall have no obligation to pay Provider for such Transition Service(s); provided that if termination of any Transition Service would materially inhibit Provider's ability to provide or prevents Provider from providing any other Transition Services as indicated in <u>Schedule A</u> or <u>Schedule B</u>, as applicable ("Bundled Services"), such other Bundled Services shall also shall be deemed terminated, subject to Recipient's prior written consent of such termination; and provided further that partial reduction of any specific Transition Service may only be made with the prior written consent of Provider, which consent shall not be unreasonably withheld, delayed or conditioned.
- (b) Except as set forth in Section 3.3(b), in the event that either Party breaches any of its material obligations under this Agreement (the "Breaching Party"), the other Party may terminate this Agreement in its entirety upon thirty (30) days' prior written notice (such thirty (30) day period, the "Notice Period") to the Breaching Party, specifying the breach and its claim of right to terminate; provided, that the termination of this Agreement shall not become effective at the end of the Notice Period if (i) the Breaching Party cures such breach during the Notice Period or (ii) such breach cannot be cured during the Notice Period and the Breaching Party commences and diligently pursues actions to cure such breach within the Notice Period, in which case the Breaching Party shall have an additional thirty (30) day period to cure such breach before such termination shall become effective.
- (c) Either Party may terminate this Agreement in its entirety immediately upon written notice to the other Party if the other Party (i) files in any court or with any other Governmental Authority, pursuant to any Law of any state or country, a petition in bankruptcy or insolvency or for reorganization or for an arrangement or for the appointment of a receiver or trustee of such other Party or of its assets; (ii) is served with an involuntary petition against it, filed in any insolvency proceeding, and such petition is not dismissed within sixty (60) days after the filing thereof; (iii) consents to the appointment or taking possession by a receiver, liquidator, assignee, custodian, trustee, sequestrator (or similar official) of such other Party or for any substantial part of its property or makes any assignment for the benefit of creditors; (iv) admits in writing its inability to pay its debts generally as they become due; or (v) has issued or levied against its property any judgment, writ, warrant of attachment or execution or similar process that represents a substantial portion of its property.
 - (d) Any Transition Service, or this Agreement in its entirety, may be terminated upon the mutual written agreement of Provider and Recipient at any time.

- 6.3 <u>Termination of Obligations</u>. Recipient specifically agrees and acknowledges that all obligations of Provider to provide each Transition Service shall immediately cease upon the expiration of the Service Period (as may be extended as set forth in this Agreement) for such Transition Service, and Provider's obligations to provide all of the Transition Services shall immediately cease upon termination of this Agreement. Recipient shall bear sole responsibility for instituting permanent services, or obtaining replacement services, in respect of any Transition Service terminated in accordance with the provisions hereof, and, except to the extent provided in the Schedules, Provider shall bear no liability for Recipient's failure to implement or obtain such service or for any difficulties in transitioning from the Transition Service to such permanent or replacement service.
- 6.4 <u>Accrued Rights</u>. Termination or expiration of this Agreement for any reason shall be without prejudice to any rights that shall have accrued to the benefit of a Party prior to such termination or expiration. Such termination or expiration shall not relieve a Party from obligations that are expressly indicated to survive the termination or expiration of this Agreement.
- 6.5 <u>Surviving Obligations</u>. Without limiting the foregoing, <u>Article VI</u> and <u>Article VII</u> and <u>Sections 2.7, 2.8, 2.9, 3.2</u> (solely with respect to accrued, unpaid fees as of such termination or expiration), <u>4.1, 4.2(b)(ii), 6.3</u> and <u>6.5</u> shall survive the termination or expiration of this Agreement for any reason.

ARTICLE VII MISCELLANEOUS

- 7.1 Non-Solicitation. During the Term of this Agreement and for a period of one (1) month after the Term, neither Party shall, directly or indirectly, in any manner solicit or induce for employment, or hire or engage the services of, any employee of the other Party without the other Party's prior written consent. A general advertisement or notice of a job listing or opening or other similar general publication of a job search or availability of employment positions, including on the internet, shall not be construed as a solicitation or inducement for the purposes of this provision.
- 7.2 Force Majeure. Provider shall not be liable for any failure to perform or any delays in performance (other than the payment of money owed and the providing of indemnity and defense), and Provider shall not be deemed to be in breach or default of its obligations set forth in this Agreement, if, to the extent and for so long as, such failure or delay is due to any causes that are beyond its reasonable control and not to its fault or negligence, including, such causes as acts of God, epidemic, pandemic, natural disasters, fire, flood, severe storm, earthquake, civil disturbance, strike, lockout, riot, order of any court or administrative body, embargo, acts of government, war (whether or not declared), acts of terrorism, or other similar causes. For clarity, in the event of any such delay, the time for performance shall be extended for a period equal to the time lost by reason of the delay.

- 7.3 <u>Complete Agreement; Construction</u>. This Agreement, including the Schedules hereto, shall constitute the entire agreement between the Parties with respect to the subject matter hereof and shall supersede all previous negotiations, commitments and writings with respect to such subject matter. In the event of any inconsistency between this Agreement and any Schedule hereto, this Agreement shall prevail. No rule of construction that disfavors the drafting party will apply to this Agreement. As used in this Agreement, "including" and words of similar import mean "including but not limited to." The use of "or" will not be deemed to be exclusive.
- 7.4 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more such counterparts have been signed by each of the Parties and delivered to the other Party.
- 7.5 Notices. All notices and other communications hereunder shall be in writing and hand delivered or mailed by registered or certified mail (return receipt requested) or sent by any means of electronic message transmission with delivery confirmed (by read receipt, voice or otherwise) to the parties at the following addresses (or at such other addresses for a party as shall be specified by like notice) and will be deemed given on the date on which such notice is received:

To Inpixon:

Inpixon 2479 E. Bayshore Road, Suite 195 Palo Alto, California 94303 Attn: Nadir Ali, Chief Executive Officer Email: nadir.ali@inpixon.com

To CXApp:

CXApp Holding Corp.
Four Palo Alto Square, Suite 200
3000 El Camino Real
Palo Alto, California 94306
Attn: Khurram Sheikh, Chief Executive Officer
Email: khurram@kins-tech.com

- 7.6 <u>Waivers</u>. The failure of any Party to require strict performance by any other Party of any provision in this Agreement will not waive or diminish that Party's right to demand strict performance thereafter of that or any other provision hereof.
 - 7.7 Amendments. This Agreement may not be modified or amended except by an agreement in writing signed by each of the Parties hereto.
- 7.8 <u>Assignment</u>. This Agreement shall not be assignable, in whole or in part, directly or indirectly; *provided, however*, that (a) either Party may assign this Agreement without the other's consent to any of its controlled Affiliates and (b) either Party may assign this Agreement in its entirety to any successor to its business, whether by merger, reorganization or otherwise; *provided, further*, that any such assignment shall not relieve the assignor of its obligations under this Agreement. Any attempt to assign any rights or obligations arising under this Agreement in contravention with this paragraph shall be null and void *ab initio*.
- 7.9 <u>Successors and Assigns</u>. The provisions of this Agreement shall be binding upon, inure to the benefit of and be enforceable by the Parties and their respective successors and permitted assigns.

- 7.10 Third Party Beneficiaries. This Agreement is solely for the benefit of the Parties hereto and should not be deemed to confer upon third parties any remedy, claim, liability, reimbursement, claim of action or other right in excess of those existing without reference to this Agreement.
- 7.11 <u>Title and Headings</u>. Titles and headings to sections herein are inserted for the convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 7.12 Schedules. The Schedules to this Agreement shall be construed with and as an integral part of this Agreement to the same extent as if the same had been set forth verbatim herein.
- 7.13 Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware (without regard to its conflicts of Law doctrines).

7.14 Jurisdiction; Waiver of Jury Trial.

- (a) Any proceeding or action based upon, arising out of or related to this Agreement or the transactions contemplated hereby must be brought in the Court of Chancery of the State of Delaware (or, to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware), or, if it has or can acquire jurisdiction, in the United States District Court for the District of Delaware, and each of the parties irrevocably and unconditionally (i) consents and submits to the exclusive jurisdiction of each such court in any such proceeding or action, (ii) waives any objection it may now or hereafter have to personal jurisdiction, venue or to convenience of forum, (iii) agrees that all claims in respect of the proceeding or action shall be heard and determined only in any such court, and (iv) agrees not to bring any proceeding or action arising out of or relating to this Agreement or the transactions contemplated hereby in any other court. Nothing herein contained shall be deemed to affect the right of any party to serve process in any manner permitted by Law or to commence Legal Proceedings or otherwise proceed against any other party in any other jurisdiction, in each case, to enforce judgments obtained in any action, suit or proceeding brought pursuant to this Section 7.14.
- (b) EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY, UNCONDITIONALLY AND VOLUNTARILY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY ACTION, SUIT OR PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY.
- 7.15 Enforcement. The parties hereto agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy could occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent any breach, or threatened breach, of this Agreement and to specific enforcement of the terms and provisions of this Agreement, without proof of damages, in addition to any other remedy to which any party is entitled at law or in equity. In the event that any action shall be brought in equity to enforce the provisions of this Agreement, no party shall oppose the granting of specific performance and other equitable relief or allege, and each party hereby waives the defense, that there is an adequate remedy at law, and each party agrees to waive any requirement for the securing or posting of any bond in connection therewith. The parties hereto acknowledge and agree that the right of specific enforcement is an integral part of the transactions contemplated hereby and without that right, none of the parties hereto would have entered into this Agreement.

- 7.16 Severability. In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein and therein shall not in any way be affected or impaired thereby. The Parties shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.
- 7.17 <u>Relationship of Parties</u>. Nothing in this Agreement shall be deemed or construed by the Parties or any third party as creating a partnership or the relationship of principal and agent or joint venturer between the Parties, it being understood and agreed that no provision contained herein, and no act of the Parties, shall be deemed to create any relationship between the Parties other than the relationship of Provider and Recipient of the Transition Services nor be deemed to vest any rights, interests or claims in any third parties.
- 7.18 <u>Insurance</u>. During the Term, Provider shall carry commercially appropriate and customary levels of insurance with a reputable insurance provider covering business interruptions and general liability insurance (including errors & omissions and contractual liability) to protect its own business and property interests.
- 7.19 <u>Audit</u>. During the term of this Agreement and for one (1) year thereafter (or such longer period as may be required by applicable Law), Provider and Recipient shall each use commercially reasonable efforts to maintain complete and accurate records related to any Transition Service provided, fees invoiced and payments made hereunder (the "Service Records"). Recipient may request a certified audit of Provider's Service Records from the date of commencement of the Transition Services to be performed by an independent certified public accountant which (a) is reasonably acceptable to Provider and (b) may not be compensated on a contingency basis or otherwise have any financial interest in the outcome of such audit. Any such audit shall be at the expense of Recipient. Recipient may not request such an audit more than one (1) time within any twelve (12) month period with respect to any particular Transition Service. The accountant shall be required to execute a confidentiality and non-disclosure agreement if requested by Provider and shall hold all information confidential. The accountant may reveal to Recipient only the amounts of any underpayment or under reimbursement, or overbilling, as applicable. The accountant shall provide to Provider a final report of its work, including both overbilling and underpayment information. The audit shall take place during normal business hours and upon reasonable notice and such accountant shall use commercially reasonable efforts to minimize interference with the normal business activities of Provider. If any audit reveals an overpayment by Recipient provider shall reimburse Recipient for the costs of conducting such audit; provided, however, in no event shall such costs exceed \$25,000. If any audit reveals an underpayment by Recipient, Recipient shall promptly pay Provider such underpayment amount.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the Effective Date.

INPIXON

By: /s/ Nadir Ali

Name: Nadir Ali

Title: Chief Executive Officer

CXAPP HOLDING CORP.

By: /s/ Nadir Ali

Name: Nadir Ali

Title: Chief Executive Officer

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

Introduction

On September 25, 2022, Inpixon (the "Company") entered into an Agreement and Plan of Merger (the "Merger Agreement") by and among the Company, KINS Technology Group Inc., a Delaware corporation ("KINS"), CXApp Holding Corp., a Delaware corporation and newly formed wholly-owned subsidiary of Inpixon ("CXApp" and, together with Inpixon, collectively, the "Companies"), and KINS Merger Sub Inc., a Delaware corporation and a wholly-owned subsidiary of KINS ("Merger Sub"), pursuant to which KINS will acquire Inpixon's enterprise apps business (including its workplace experience technologies, indoor mapping, events platform, augmented reality and related business solutions) (the "Enterprise Apps Business") in exchange for the issuance of shares of KINS capital stock (the "Business Combination") to be issued to Inpixon stockholders.

Immediately prior to the Merger and pursuant to the Separation and Distribution Agreement, dated as of September 25, 2022, among KINS, Inpixon, CXApp and Design Reactor, and other ancillary conveyance documents, Inpixon will among other things and on the terms and subject to the conditions of the Separation and Distribution Agreement, effect the reorganization by the transfer of the Enterprise Apps Business, including certain related subsidiaries of Inpixon, including Design Reactor, to CXAapp and, in connection therewith, will effect the distribution by distributing to Inpixon securityholders 100% of the CXApp common stock.

Immediately following the distribution, in accordance with and subject to the terms and conditions of the Merger Agreement, Merger Sub will merge with and into CXApp, which CXApp continuing as the surviving company in the Merger and as a wholly-owned subsidiary of KINS (the "Merger")

The Merger Agreement, along with the Separation Agreement and the other transaction documents entered into in connection therewith, provides for, among other things, the consummation of the following transactions: (i) Inpixon will transfer the Enterprise Apps Business (the "Separation") to its wholly-owned subsidiary, CXApp, and contribute \$10 million in capital thereto (net of cash held by CXApp as of the effective time) (the "Cash Contribution"), (ii) following the Separation, Inpixon will distribute 100% of the shares of CXApp Common Stock to Inpixon stockholders and other security holders by way of the Distribution and (iii) following the completion of the foregoing transactions and subject to the satisfaction or waiver of certain other conditions set forth in the Merger Agreement, the parties shall consummate the Merger. The Separation, Distribution and Merger are intended to qualify as "tax-free" transactions.

On March 14, 2023, the Company completed the tax-free distribution of 100% of the outstanding capital stock of CXApp Holding Corp., a wholly owned subsidiary, on a pro rata basis to holders of the Company's outstanding capital stock and certain other securities of record as of March 6, 2023. The distribution was immediately followed by the closing of the business combination between CXApp and KINS.

The following unaudited pro forma condensed consolidated financial information is presented in accordance with the rules specified by Article 11 of Regulation S-X promulgated by the U.S. Securities and Exchange Commission (the "SEC") and has been prepared subject to the assumptions and adjustments as described in the notes thereto. Specifically, the unaudited pro forma condensed consolidated financial information set forth below reflects the effects of the Business Combination on (i) the Company's condensed consolidated balance sheet as of September 30, 2022, as if the Business Combination had occurred on that date, and (ii) Company's condensed consolidated statement of operations for the nine months ended September 30, 2022, and the year ended December 31, 2021, as if the Business Combination had occurred on January 1, 2021. Management believes that the assumptions used, and adjustments made are reasonable under the circumstances and given the information available.

The following unaudited pro forma condensed consolidated financial statements have been derived from historical financial statements prepared in accordance with U.S. generally accepted accounting principles ("US GAAP"). The pro forma adjustments reflect the impacts of events directly attributable to the Business Combination, that are factually supportable, and for purposes of the unaudited pro forma condensed consolidated statements of operations, expected to have a continuing impact on the Company. The following unaudited proforma condensed consolidated financial information is for illustrative and informational purposes only and is not necessarily indicative of the financial condition or results of operations of the Company that would have occurred if the Business Combination had occurred on the dates indicated, nor is it indicative of the future financial condition or results of operations of the Company.

The unaudited pro forma condensed consolidated financial information should be read in conjunction with the following:

- The accompanying notes to the unaudited pro forma condensed consolidated financial statements;
- The Company's unaudited condensed consolidated financial statements as of and for the nine months ended September 30, 2022, included in its Quarterly Report on Form 10-Q for the quarter ended September 30, 2022; and
- The Company's audited consolidated financial statements as of and for the year ended December 31, 2021, included in its Annual Report on Form 10-K for the year ended December 31, 2021.

The historical condensed consolidated statement of operations for the year ended December 31, 2021, has been adjusted by Company management to reflect certain reclassifications to conform with current financial statement presentation.

INPIXON AND SUBSIDIARIES UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET As of September 30, 2022

(In thousands, except number of shares and par value data)

	Inpixion and Subsidiaries Historical (a)		Pro Forma Adjustments		Note	Inpixion and Subsidiaries Pro Forma	
Assets							
Current Assets							
Cash and cash equivalents	\$	63,153	\$	(33,550)	(c), (d), (g), (h), (i)	\$	29,603
Accounts receivable, net of allowances		2,879		(1,496)	(c)		1,383
Other receivables		137		(45)	(c)		92
Inventory		2,702		(10)	(c)		2,692
Note receivable Prepaid expenses and other current assets		150		(2.022)	(a)		150
Prepard expenses and other current assets		3,258		(2,033)	(c)	_	1,225
Total Current Assets		72,279		(37,134)			35,145
Property and equipment, net		1,307		(213)	(c)		1,094
Operating lease right-of-use asset, net		1,323		(730)	(c)		593
Software development costs, net		1,684		(535)	(c)		1,149
Investments in equity securities		1,124		-			1,124
Long-term investments		2,500		(2,500)	(e)		-
Intangible assets, net		28,174		(20,215)	(c)		7,959
Other assets		204	_	(57)	(c)		147
Total Assets	\$	108,595	\$	(61,384)		\$	47,211
Liabilities and Stockholders' Equity							
Current Liabilities							
Accounts payable	\$	2,559	\$	(786)	(c)	\$	1,773
Accrued liabilities	_	4,370		(2,050)	(c)		2,320
Operating lease obligation, current		514		(256)	(c)		258
Deferred revenue		3,730		(2,576)	(c)		1,154
Short-term debt		6,179		-			6,179
Acquisition liability		3,376		-			3,376
Total Current Liabilities		20,728		(5,668)			15,060
Long Term Liabilities							
Operating lease obligation, noncurrent		852		(502)	(c)		350
Other liabilities, noncurrent		28		(30)	(c)		(2)
Total Liabilities		21,608		(6,200)			15,408
Commitments and Contingencies							
Mezzanine Equity							
Series 8 Convertible Preferred Stock- 53,197.7234 shares authorized, issued and		<i>5</i> 2 100		(52 100)	(L)		
outstanding as of September 30, 2022, respectively.		53,198		(53,198)	(h)		-
Stockholders' Equity							
Preferred Stock -\$0.001 par value; 5,000,000 shares authorized							
$Series\ 4\ Convertible\ Preferred\ Stock\ -\ 10,\!415\ shares\ authorized;\ 1\ issued\ and\ outstanding$							
as of September 30, 2022.		-		-			-
Series 5 Convertible Preferred Stock - 12,000 shares authorized; 126 issued and outstanding as of September 30, 2022.		_		_			_
Common Stock - \$0.001 par value; 26,666,667 shares authorized; 2,250,597 issued		-		-			-
and 2,250,596 outstanding as of September 30, 2022, respectively.		2		10	(f), (g), (i)		12
Additional paid-in capital		331,487		29,638	(g), (i)		361,125
Treasury stock, at cost, 1 share		(695)		-	(8/) (-)		(695)
Accumulated other comprehensive income		1,496		-			1,496
Accumulated deficit		(299,123)		(31,634)	(c), (d), (e)		(330,757)
Stockholders' Equity Attributable to Inpixon		33,167		(1,986)			31,181
Non-controlling Interest		622		<u> </u>			622
Total Stockholders' Equity		33,789		(1,986)			31,803
Total Liabilities, Mezzanine Equity and Stockholders' Equity	\$	108,595	\$	(61,384)		\$	47,211

INPIXON AND SUBSIDIARIES UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS For the nine months ended September 30, 2022

(In thousands, except number of shares and par value data)

		on and diaries orical a)	Pro Forma Adjustments	Note	Sul	Inpixion and Subsidiaries Pro Forma	
Revenues	\$	14,133	\$ (6,47	(j)	\$	7,660	
Cost of Revenues		4,037	(1,62	(j)		2,409	
Gross Profit		10,096	(4,84	15)		5,251	
Operating Expenses							
Research and development		13,642	(6,92	(j)		6,713	
Sales and marketing		6,757	(3,79	(j), (k)		2,960	
General and administrative		19,788	(8,39	(j), (k)		11,395	
Acquisition-related costs		270		(j)		254	
Impairment of goodwill		7,570	(5,54			2,030	
Amortization of intangibles		4,056	(2,91	<u>9</u>) (j)		1,137	
Total Operating Expenses		52,083	(27,59	<u>04</u>)		24,489	
Loss from Operations		(41,987)	22,74	19		(19,238)	
Other Income (Expense)							
Interest expense, net		(62)	((3) (j)		(65)	
Other expense, net		(637)		1 (j)		(636)	
Unrealized loss on equity securities		(7,110)		-		(7,110)	
Total Other Expense		(7,809)		(2)		(7,811)	
Net Loss, before tax		(49,796)	22,74	17		(27,049)	
Income tax provision		(84)	6	52 (j)		(22)	
Net Loss		(49,880)	22,80	9		(27,071)	
Net Loss Attributable to Non-controlling Interest		(1,206)		<u>-</u>		(1,206)	
Net Loss Attributable to Stockholders of Inpixon		(48,674)	22,80	9		(25,865)	
Accretion of Series 7 preferred stock		(4,555)		-		(4,555)	
Accretion of Series 8 Preferred Stock		(13,089)		-		(13,089)	
Deemed dividend for the modification related to Series 8 Preferred Stock		(2,627)		-		(2,627)	
Deemed contribution for the modification related to Warrants issued in connection with Series 8 Preferred Stock		1,469		-		1,469	
Amortization premium-modification related to Series 8 Preferred Stock		2,626		-		2,626	
Net Loss Attributable to Common Stockholders	\$	(64,850)	\$ 22,80	9	\$	(42,041)	
Basic and diluted loss per share	\$	(31.08)		(f)	\$	(2.94)	
Weighted Average Shares Outstanding, basic and diluted	2	2,086,633	12,219,30	9 (f), (g), (i)		14,305,942	

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

INPIXON AND SUBSIDIARIES UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS For the year ended December 31, 2021

(In thousands, except number of shares and par value data)

	Inpixion and Subsidiaries Historical (b)	Pro Forma Adjustments	Note	Inpixion and Subsidiaries Pro Forma	
Revenues	\$ 15,995	\$ (6,368)	(j)	\$ 9,627	
Cost of Revenues	4,374	(1,646)	(j)	2,728	
Gross Profit	11,621	(4,722)		6,899	
Operating Expenses					
Research and development	14,121	(6,704)	(j)	7,417	
Sales and marketing	8,261	(4,763)	(j), (k)	3,498	
General and administrative	41,478	(21,056)	(j), (k)	20,422	
Acquisition-related costs	1,248	(628)	(j)	620	
Impairment of goodwill	14,789	(11,896)	(j)	2,893	
Amortization of intangibles	4,467	(3,047)	(j)	1,420	
Total Operating Expenses	84,364	(48,094)		36,270	
Loss from Operations	(72,743)	43,372		(29,371)	
Other Income (Expense)					
Interest income, net	1,183	(1)	(j)	1,182	
Loss on exchange of debt for equity	(30)	-		(30)	
Benefit for valuation allowance on related party loan - held for sale	7,345	-		7,345	
Other expense, net	(47)	-		(47)	
Gain on related party loan - held for sale	49,817	-		49,817	
Unrealized loss on equity securities	(57,067)	-		(57,067)	
Total Other Income	1,201	(1)		1,200	
		42.2-:		40.00	
Net Loss, before tax	(71,542)	43,371		(28,171)	
Income tax benefit (provision)	1,412	(2,527)	(j)	(1,115)	
Net Loss	(70,130)	40,844		(29,286)	
Net Loss Attributable to Non-controlling Interest	(975)	<u>-</u>		(975)	
Net Loss Attributable to Stockholders of Inpixon	(69,155)	40,844		(28,311)	
Accretion of Series 7 preferred stock	(8,161)	-		(8,161)	
Net Loss Attributable to Common Stockholders	\$ (77,316)	\$ 40,844		\$ (36,472)	
Basic and diluted loss per share	\$ (51.18)		(f)	\$ (2.66)	
Weighted Average Shares Outstanding, basic and diluted	1,510,678	12,219,309	(f), (g), (i)	13,729,987	

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

INPIXON AND SUBSIDIARIES NOTES TO UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

(1) Basis of presentation

The unaudited pro forma condensed financial statements are based on the historical consolidated financial statements of the seller as adjusted to give effect to the Separation. The unaudited pro forma condensed consolidated statements of operations for the nine months ended September 30, 2022, and the year ended December 31, 2021, give effect to the Separation as if it were completed on January 1, 2021. The unaudited pro forma condensed consolidated balance sheet as of September 30, 2022, gives effect to the Separation as if it were completed on September 30, 2022. The transaction accounting adjustments for the Separation consist of those necessary to account for the Separation. Certain reclassifications have been made to the prior period financial statements to conform to the current period financial statement presentation. Certain amounts related to foreign currency transaction expense from the prior period were reclassified from Other Income / (Expense) line item to General and Administration line item on the Unaudited Condensed Consolidated Statement of Operations. These reclassifications had no net effect on loss from operations, net loss, or cash flows as previously reported.

(2) Unaudited Pro Forma Adjustments

The following is a summary of the unaudited pro forma adjustments reflected in the unaudited pro forma condensed consolidated financial statements based on preliminary estimates, which may change as additional information is obtained.

- a. Reflects amounts as originally reported by the Company in its Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2022.
- b. Reflects amounts as originally reported by the Company in its Annual Report on Form 10-K for the year ended December 31, 2021.
- c. Reflects the elimination of the Enterprise Apps Business assets, liability and historical equity balances including within the Company's consolidated financial statements. The Company notes \$69 million of consideration in connection with the Business Combination was paid directly to the shareholders of the Company as a result of the spin-off, and no adjustment was included in the unaudited pro forma consolidated balance sheet as a result.
- d. Reflects adjustments for remaining cash contribution of \$4.0 million to reach \$10 million cash contribution in accordance within the Separation and Distribution agreement.
- e. Reflects adjustments to the Company's investment in Class A and Class B Units of Cardinal Ventures Holdings LLC, which has certain interests in the sponsor of KINS. The Company distributed its ownership interests to certain employees and members of management on February 28, 2023 as pre-requisite to the Business Combination.
- f. The Inpixon and Subsidiaries historical September 30, 2022 and December 31, 2021, information reflects the adjustment for the reverse stock split of the Company's authorized and issued and outstanding shares of common stock at a ratio of one (1) share of common stock for every seventy five (75) shares of common stock. The Company executed the reverse stock split on October 7, 2022.
- g. Reflects adjustment for offering for the sale of an aggregate of 253,112 shares of common stock at an offering price of \$5.85 per share, Purchase Warrants to purchase 3,846,153 shares of common stock at an exercise price of \$5.85 per share and Pre-Funded Warrants to purchase 2,310,990 shares of common stock, at a purchase price equal to the price at which each share of common stock is sold in this offering, minus \$0.001, and an exercise price of \$0.001 per share. The net proceeds from the offering was approximately \$14.2 million. The Company executed the securities purchase agreement on October 18, 2022.
- h. Reflects adjustment for the redemption of Series 8 Convertible Preferred Stock occurring between October 1, 2022 to December 31, 2022. Redemption notices covered a total of 53,198 shares of Series 8 Convertible Preferred Stock for aggregate cash required to be paid of \$53.2 million.
- i. Reflects adjustment for offering for the sale of an aggregate of 9,655,207 shares of common stock at share prices between \$1.15 and \$1.86 per share that have occurred from January 1, 2023 to March 15, 2023. The Company recorded gross proceeds of approximately \$15.4 million.
- j. Reflects the elimination of the historical revenues and expenses directly to related to the Enterprise Apps Business that will not recur in the Company's statement of operations beyond the date of the Business Combination.
- k. Reflects management's estimates of approximately \$1.1 million and \$0.8 million of historical costs mainly for executive salaries and benefits in general and administrative expenses and sales and marketing expenses that were allocated to Enterprise Apps Business. The historical costs were added back to the statement of operations for the year ended December 31, 2021 and for the nine months ended September 30, 2022, respectively, as the costs would be incurred by the Company.